Executive summary

The Government has an objective of lowering regulatory costs on business and improving market outcomes for consumers, by encouraging self-regulation, where this is the most effective option for addressing an identified problem. The Government also has the objective that industry should take increased ownership and responsibility for developing efficient and effective self-regulation.

Before deciding on the most appropriate regulatory response, any specific industry problems and objectives need to be clearly defined. Once the decision has been made that intervention is necessary, recognising that regulation in any form can potentially set up barriers to entry to the industry, or stifle innovation or competition amongst industry participants, then the focus can properly shift to choosing the most appropriate model of regulation to achieve the desired outcome.

In a broad sense, regulation can be considered as a spectrum ranging from self-regulation where there is little or no government involvement, through quasi-regulation which refers to a range of rules, instruments or standards that government expects businesses to comply with, to explicit government regulation.

Consistent with its Terms of Reference, the Taskforce has examined the option of self-regulation.

Australia is at the forefront of international policy initiatives to promote regulatory reform and effective self-regulation. The Taskforce has had regard to the international experience with industry self-regulation in this report. An outline of international policy and practice can also be found in Appendix D of the Taskforce's *Draft Report*.

Self-regulatory schemes tend to promote good practice and target specific problems within industries, impose lower compliance costs on business, and offer quick, low cost dispute resolution procedures. Effective self-regulation can also avoid the often overly prescriptive nature of regulation and allow industry the flexibility to provide greater choice for consumers and to be more responsive to changing consumer expectations.

The Taskforce recognised that self-regulation may not be appropriate in all circumstances. Other forms of regulation may provide more cost effective outcomes in certain cases. As well, community cynicism regarding industry regulating itself may lead to a distrust of self-regulatory schemes unless schemes operate effectively and consumers have confidence in them.

Individual firms that are not part of a self-regulatory scheme may also gain commercial advantages from avoiding the costs and sanctions involved in

participating in self-regulation. However, these problems are not unique to self-regulatory schemes and in practice can exist within regulated industries.

Self-regulation includes a host of options ranging from a simple code of ethics, to codes that are drafted with legislative precision together with sophisticated customer dispute resolution mechanisms.

The Taskforce has examined self-regulation in terms of significantly improved market outcomes for consumers with direct reference to lowering costs to industry participants, thus providing shared benefits to both businesses and consumers. This approach focuses on the efficiency and effectiveness of self-regulatory structures. The Taskforce has assessed that there needs to be a balance between minimising costs for business and the benefits to both business and consumers by looking at the market circumstances where self-regulation arose.

Good practice in self-regulation involves applying an appropriate scheme to a specific market failure or social policy objective. Ascertaining which scheme should be applied will depend on the nature and risk of the market failure and the consequences of no action. In other words, there is no one model for self-regulation. Nonetheless, it is possible to identify common characteristics of successful schemes and the Taskforce has done so in Chapter 6 of its report.

The Taskforce has applied these principles when inquiring into and reporting on its Terms of Reference. The Taskforce has reached the following conclusions which are cross-referenced to the body of the report to assist readers to locate the discussion of, and rationale for, each finding.

Chapter 3: Types of self-regulation in consumer markets in Australia

There are different reasons for establishing self-regulatory schemes. Industries may self-regulate to improve the image of suppliers or to promote consumer confidence in new products or technologies. Industries may also self-regulate to avoid regulation, satisfy legislative requirements or minimise costly litigation.

Industry has an array of self-regulatory options available to address specific problems and objectives, including codes of conduct, industry service charters, guidelines and standards, as well as industry-based accreditation and complaint handling schemes.

The Taskforce examined self-regulation across a broad range of industries, including broadcasting and media, telecommunications, and financial services. The Taskforce also examined many self-regulatory schemes dealing with marketing practices generally — including advertising, direct marketing and the use of scanning equipment in supermarkets.

The Taskforce reached the following conclusions:

- 1. There is a broad and diverse range of self-regulation at the national level affecting consumers (page 24).
- 2. There is no single model for industry self-regulation as it depends on what is trying to be achieved (page 24).

Chapter 4: Gaps and overlaps in the coverage of self-regulation

Self-regulatory schemes operate in dynamic markets, which are influenced by globalisation, increasing vertical integration, and the growth of 'hybrid' products that span traditional markets or industries.

As a consequence, gaps and overlaps can emerge in the coverage of various products, services, sectors and industries. Similarly, existing self-regulatory schemes may find themselves covering the same ground where the distinction between products or services has become blurred.

It is undesirable that there be market problems that are not addressed by industry self-regulation, as consumers may find it costly and time consuming to obtain redress through the Courts. It is equally undesirable that there be inefficient duplication of self-regulatory schemes. The Taskforce is confident that self-regulation is sufficiently flexible to respond quickly to new market issues.

In short, the Taskforce concluded that:

- 3. Gaps and overlaps continually emerge and re-emerge in dynamic markets (page 33).
- 4. A 'gap' in the market does not necessarily mean that self-regulation is the appropriate solution (page 34).
- 5. Self-regulation is a flexible response to market failure and may fill a 'gap' quickly and efficiently (page 34).
- 6. Some small businesses can have difficulties in participating in self-regulatory schemes as can consumers (page 34).

Chapter 5: Industry environment and market circumstances where self-regulation is likely to be most effective

Self-regulation is not the answer to every market failure and all social policy objectives. The Taskforce was asked to provide some guidance for industry and policymakers as to where self-regulation is likely to prove most effective.

There is a general recognition that industry self-regulation is often more flexible and less costly for both business and consumers than direct government involvement.

However, it is necessary to ensure that self-regulation is the appropriate form of intervention given the particular industry environment and market circumstances.

The circumstances where self-regulation is likely to be most effective will depend on the nature and extent of market failure, the market structure, industry and consumer interests.

Nature and extent of market failure

7. Self-regulation is likely to be most effective where there are clearly defined problems but no high risk of serious or widespread harm to consumers (page 44).

Market structure

- 8. An industry environment with an active industry association and/or industry cohesiveness is most likely to administer effective self-regulation as industry participants are more likely to commit financial resources, consult with stakeholders and monitor the effectiveness of self-regulation (page 48).
- 9. Self-regulation is less effective where there is a broad spread of smaller businesses that do not communicate with each other (page 49).
- 10. Self-regulation is more likely to be effective in a competitive market as industry participants are more likely to be committed to it, either to differentiate their products or in fear of losing market share (page 48).
- 11. A more mature industry may be able to administer more effective self-regulation, as industry participants are more likely to have sufficient resources and be more committed while any 'shakeout' of rogue traders may already have occurred (page 50).

Industry and consumer interests

- 12. Self-regulation is likely to be most effective where firms recognise that their future viability depends not only on their relationship with their current customers and shareholders, but also on the wider community (page 50).
- 13. The more incentives there are for industry participants to initiate and comply with self-regulation, then the more chance a scheme can remedy specific industry problems (page 53).
- 14. The extent to which industry participants are prepared to sign up to a self-regulatory scheme will affect the ability of that scheme to provide effective self-regulation. Where a scheme has a high level of consumer recognition, to the point where consumers will favour scheme participants when making purchasing decisions, then the scheme is most likely to be effective. This will create incentives for non-members to join the scheme (page 55).

- 15. The interests of all levels of industry should be considered in the development and maintenance of a self-regulatory scheme, and particularly the level of involvement of smaller businesses where appropriate (page 56).
- 16. Where there are cost advantages and/or increased flexibility in self-regulatory initiatives to address specific industry problems compared with government regulation or the court system, then there is a greater chance of improving market outcomes for both business and consumers, and minimising compliance costs for businesses (page 56).

Chapter 6: Good practice and cost effective self-regulation methods

There is no single 'best practice' model for self-regulation because a successful model needs to be designed to address particular problems identified in the context of particular market circumstances. Accordingly, the Taskforce considered it inappropriate to develop a 'checklist' of features of good self-regulation. Nonetheless, it is possible to identify critical elements that, individually or collectively, have underpinned effective schemes.

Good practice in self-regulation can be understood as significantly improving market outcomes for consumers at the lowest cost to businesses, and the following factors were seen as contributing to this.

Consultation

17. Consultation between industry, consumers and government can help ensure that specific problems and social policy objectives can be identified and addressed (page 63).

Coverage and publicity

- 18. Increased industry coverage of schemes ensures that the benefits from standards of practice in schemes flow to consumers. Wide coverage also ensures that consumers can identify self-regulatory schemes (page 65).
- 19. Clarity in the schemes' documentation can help industry understand their obligations and assist dispute schemes interpret legal rights. Clarity can also help consumers understand their rights (page 65).
- 20. Consumer awareness of schemes ensures that consumers know where to lodge complaints. Schemes are encouraged to make use of new technologies such as the Internet, make complaints cost free to the consumer, write sample letters of complaint, take oral complaints, provide personal contact and transfer complainants between schemes (page 66).
- 21. Industry awareness campaigns and education about schemes is needed to make sure industry participants understand their obligations and, where appropriate, understand the consequences of failing to abide by these obligations (page 68).

Administration

- 22. A good administrative body can identify issues, collect data, monitor the scheme, enhance credibility and ensure compliance costs are at an effective minimum level (page 69).
- 23. Data collection by an industry scheme is a valuable tool in identifying systemic issues and allows industry to address these problems, which in turn, can improve market outcomes for both businesses and consumers (page 70).
- 24. As consumers cannot guard against specific industry problems that they do not know exist, transparency in schemes is an important mechanism to ensure credibility and accountability (page 71).

Dispute procedures and sanctions

- 25. Industry adherence to self-regulatory schemes is essential to ensure that the benefits flowing from the standards of practice set by schemes are passed onto the consumer (page 71).
- 26. Where the standard of conduct has been breached, self-regulatory schemes should incorporate complaint handling and dispute resolution mechanisms to provide appropriate redress to consumers. The appropriate redress mechanism will depend on the nature of the specific problem and the consequences of non-compliance (page 73).
- 27. A range of sanctions can be used by industry in order to achieve compliance depending on the nature of the specific problem and consequences of non-compliance. The severity of the sanction should depend on the seriousness of the breach (page 75).
- 28. Industry needs to manage the risk of any anti-competitive practices in schemes, particularly where sanctions are involved (page 77).

Monitoring and reviewing

29. Monitoring of self-regulation is essential to ensure that it is still relevant to the industry addressing specific problems and improving market outcomes. In this context, reviews and annual reporting are important tools for monitoring schemes and can also assist in the transparency and accountability of schemes. Preferably, reviews should be periodic, independent and the results made publicly available (page 78).

Cost-effectiveness

30. Self-regulation comes at a cost, in administration, promotion and compliance. However, self-regulation can be cheaper (in terms of compliance costs) and more flexible than government regulation and the court system. Ultimately, the consumer bears the cost of regulation in most cases as it is part of a firm's cost structure (page 82).

31. Any funding arrangement for self-regulation should be transparent and designed so as not to put businesses at a competitive disadvantage through excessive compliance costs (page 85).

Chapter 7: Approaches to promoting and coordinating industry self-regulation

There are a variety of options for designing and promoting self-regulatory schemes and what works for one industry may not work for another. It follows that the 'mix' of industry, government and consumer involvement that works well for one self-regulatory scheme may be inappropriate for another.

Industry approaches to promoting self-regulation

- 32. Experience has shown that industry will initiate a self-regulatory scheme in response to a clear commercial imperative to win consumer confidence and boost sales (page 89).
- 33. Industry may promote self-regulation as an alternative to government regulation where there is perceived to be a serious market failure or important social policy objective (page 90).

Role of government in promoting and coordinating self-regulation

- 34. Government involvement in self-regulation is justified when there is a public policy objective that would otherwise call for a regulatory response (page 91).
- 35. Government can assist in analysing systemic problems in an industry and in facilitating the design of a self-regulatory response to address those systemic problems (page 96).
- 36. Government can assist in integrating schemes into the regulatory framework (page 96).
- 37. Government is uniquely placed to promote international cooperation and harmonisation of self-regulatory initiatives (page 96).
- 38. The degree of government involvement will depend on the significance of the market failure or social policy objective being addressed and the consequences of self-regulation proving ineffective (page 99).

Role of consumer advocates in promoting self-regulation

- 39. Consumer input is important in the development and in maintaining the relevance of self-regulation. Consumer advocates can promote consumer confidence in self-regulatory schemes (page 104).
- 40. Consumer participation will be limited by human and financial resource constraints if there is no external financial assistance forthcoming (page 106).

Other conclusions

41. Code administration authorities established by industry should take responsibility for the monitoring and review of self-regulation, in consultation with government and consumer groups.

Chapter 8: Options that facilitate the improvement and harmonisation of dispute resolution schemes

Effective dispute resolution is a crucial element of industry self-regulation offering redress to consumers and it can also identify systemic problems in the industry. Dispute resolution schemes are an excellent monitoring tool increasing performance and industry standards.

- 42. In the future dispute resolution schemes may operate across different sectors with similar products/services, driven by changes in technology and market circumstances. Harmonisation of schemes would be less costly and less confusing to consumers and the use of umbrella-type arrangements with a single co-ordinated access point would likewise be of assistance to consumers (page 115).
- 43. Promotion of dispute resolution schemes to consumers raises their awareness of the availability of quick and inexpensive redress (page 116).

Other conclusions

The Taskforce encourages the government, in addition to existing guidelines and benchmarks, to provide industries with further practical guidelines based on the principles flagged in this report to help inform/assist the development and review of self-regulatory schemes.

The Taskforce also encourages the government to consider up-dating its guidelines for policy makers on how to assess the range of options for addressing a particular market failure or social policy objective. The purpose of such a revision would be to incorporate the Taskforce findings on the industry environment and market circumstances that are most likely to lead to effective self-regulation.

Introduction

On 12 August 1999, the Minister for Financial Services & Regulation, the Hon Joe Hockey MP, who is also the Minister with responsibility for Consumer Affairs, established an independent Taskforce to advise him on a range of issues regarding industry self-regulation in Australia.

The Taskforce members to undertake this inquiry were drawn mainly from the private sector, with the emphasis placed on representation from key stakeholders, with knowledge of, and involvement in, self-regulation.

Inquiry process

The Taskforce met for the first time on 30 September 1999, to discuss the format and direction of the inquiry. One of the most important objectives of the Taskforce was to ensure all stakeholders had ample opportunity to contribute to the inquiry. The Taskforce considered that consultations with stakeholders across industry, business, consumers groups and government were imperative to explore a variety of experiences in the field and generate an informed and comprehensive report.

The Taskforce agreed that the first step to achieving effective consultation was authorising the release of the *Issues Paper*. The purpose of this *Issues Paper* was to provide information on the scope of the Terms of Reference and the inquiry methodology to assist stakeholders in preparing submissions to the inquiry. The call for submissions was advertised nationally on Friday, 15 October 1999 (the *Australian Financial Review*) and in *The Australian* and major metropolitan dailies in each State and Territory on the weekend of 16 October 1999.

To supplement this national advertising campaign, the Chair of the Taskforce wrote to a wide range of industry, consumer groups and government agencies (over 80 organisations) likely to have an interest in the inquiry, inviting then to make a submission and attend consultative meetings with the Taskforce. In addition, other organisations, including the Council of Small Business Organisations of Australia and the Australian Communications Industry Forum, widely distributed the *Issues Paper* to interested parties to ensure thorough saturation across Australia.

⁻

¹ The Issues Paper can be found at http://www.treasury.gov.au/self-regtaskforce.

Regional Australia was also targeted with one-page fliers, prepared by the Taskforce, for distribution to country areas with assistance from the Business and Professional Women of Australia. Letters were also sent to all regional development councils around Australia inviting them to participate in the inquiry through encouraging submissions on issues such as access to self-regulatory schemes for regional consumers and industry. A number of regional development councils took up the Taskforce's offer to receive a copy of the *Draft Report* with and participated in the second round of consultations.

In addition to this media and promotional campaign, the Taskforce set up its own webpage on the Treasury site (http://www.treasury.gov.au/self-regtaskforce) and electronic letterbox for lodgement of submissions and queries (selfregtaskforce@treasury.gov.au). Links were also established to the Taskforce webpage from other sites that stakeholders were likely to visit. The *Issues Paper* and *Draft Report* were published on the website, together with reference documents such as the *Codes Policy Framework* and *Grey-letter Law*.

The Taskforce was pleased at the numerous responses elicited from stakeholders throughout the inquiry. The Taskforce received 59 submissions (listed at Appendix A) which are used extensively within the body of the report.

In addition to calling for submissions, the first round of consultations was undertaken during November and December 1999 and March 2000. Public consultations were held with industry, business, consumer groups and government agencies located in Canberra, Melbourne and Sydney. The consultations were conducted in a variety of formats with a number of individual and roundtable discussions held. A list of organisations consulted during the inquiry is listed at Appendix B.

The second meeting of the Taskforce was held on 14 December 1999. The Taskforce reviewed the inquiry process to date and was enthused by the response received by stakeholders. The Taskforce examined the trends of information received. The Taskforce was also concerned at the short timeframe attributed to the inquiry considering the large amount of work that was still required to be undertaken. The Taskforce was anxious to avoid shortening the public consultation period due to a lack of time. Hence, the Taskforce applied to the Minister for a three month extension on the reporting date. The Minister granted this extension. The Taskforce also commissioned a number of special research tasks to further investigate the Terms of Reference.

The importance placed on a comprehensive and thorough analysis of market circumstances where industry self-regulation was likely to be most and least effective, became the catalyst for the Taskforce to engage a consultant. The Taskforce required the consultant to structure its research around case studies where industry self-regulation had been implemented. As part of its research strategy, the consultant met with stakeholders in each of the identified industries to gain a comprehensive insight into the market. The aim of this research was not to identify self-regulatory

success and failure in particular industries but to more broadly identify the characteristics of the environment and market that have influenced the effectiveness of self-regulation. This report can be found on the Taskforce's webpage and has also been incorporated into chapter 5 of the report.

The Taskforce was also keen to research the changing regulatory environment within the international arena, with special attention to be given to developments in industry self-regulation. To ensure Australia could potentially learn from this research, the Taskforce supported the need to focus on governments that had already implemented some form of industry self-regulation policy. The Secretariat to the Taskforce undertook this research with the view to investigate current issues and trends in the international environment.

In addition, the Taskforce Secretariat subscribed to an international on-line forum on voluntary codes, hosted by the Canadian Office of Consumer Affairs. The members of this forum come from a wide variety of government and non-government organisations throughout the world and provide invaluable information, networks and links to self-regulatory practices and voluntary codes in operation. The Terms of Reference of the Taskforce inquiry were posted on this forum and consequently generated a large amount of interest. The numerous replies provided valuable information on self-regulatory practices, from a variety of international governments, organisations and individuals. Further detail on the international work can be found at Appendix D of the *Draft Report*.

The third meeting of the Taskforce was held on 8 May 2000. As requested in the Terms of Reference for the inquiry, the Taskforce authorised the release of a *Draft Report* for comment by interested parties. The *Draft Report* drew heavily on the submissions received together with the findings from consultations and research undertaken by the consultant and Secretariat.

The purpose of this *Draft Report* was to gather feedback on the Taskforce conclusions and find out whether there were further issues it should raise in this report. The Taskforce sought both written and oral comments. In seeking comments, the second round of consultations was undertaken during June and July 2000. Public consultations were held with industry, business, consumer groups and government agencies located in Canberra, Melbourne, Sydney, Brisbane, Adelaide, Perth and in regional centres (see below for more detail). Although the focus of the inquiry was on self-regulation in consumer markets where the Commonwealth Government has constitutional responsibility or where there is a national scheme in place, the Taskforce was also keen to learn from States' experiences with self-regulation and consulted with a number of State bodies including some State Fair Trading agencies.

On 23 August 2000, the Taskforce met for the last time to finalise the report. The Taskforce was very pleased with the positive feedback on its *Draft Report* which resulted in some material being added and refined. However, the *Draft Report* forms the nucleus for this report.

Consultation in regional areas

The Taskforce was very keen to ensure that organisations in more regional areas had an adequate opportunity to participate in the inquiry. As a result, the Taskforce travelled to Mildura, Rockhampton, Tamworth and Bunbury to enquire about the self-regulatory issues facing rural Australians. The Taskforce also offered to travel to other regional areas if there was sufficient interest.

The Taskforce heard from these consultations that rural Australians can have trouble accessing services in comparison to metropolitan residents. Accessing self-regulatory schemes was not so problematic in itself given the technology available, but problems lay with knowing how to lodge complaints with schemes, having the time to write letters of complaint, the lack of personal contact and the time associated with having complaints investigated.

During some its regional consultations, the Taskforce also heard that rural primary industries like to have some level of government involvement as a safety net. In particular, concern was expressed over rogue traders being able to keep operating under a self-regulatory scheme and potentially undermining it.² The Taskforce acknowledges these concerns but also recognises that similar problems exist with rogue players in regulated and deregulated industries.

Structure of report

The main body of the report addresses the six broad issues that the Taskforce was asked to inquire and report on.

In particular, chapter 2 sets out the framework for the Taskforce inquiry and discusses the steps involved in assessing whether self-regulation is the most appropriate tool.

Chapter 3 discusses the reasons for the initiation of self-regulatory schemes and the spectrum of schemes in Australia ranging from guidelines to more sophisticated codes of practice. The directory of self-regulatory schemes also lists a host of self-regulatory schemes operating at the national level in consumer markets.³

Chapter 4 then discusses gaps and overlaps in the coverage of self-regulation. There has been considerable growth in the number of self-regulatory schemes across many industries. In addition, these self-regulatory schemes operate in dynamic markets, which are influenced by globalisation, increasing vertical integration, and the growth of 'hybrid' products that span traditional markets or industries. As a consequence,

² For example, Taskforce consultations held in Tamworth on 28 June 2000 and Bunbury on 20 July 2000.

³ The directory of self-regulatory schemes can be found at the Taskforce's website.

gaps and overlaps can emerge in the coverage of various products, services, sectors and industries.

Chapter 5 maps out the industry environment and market circumstances where self-regulation is likely to be most effective. It is necessary to ensure that self-regulation is the appropriate form of intervention given particular industry environment and market circumstances, otherwise inappropriate intervention could create new problems. As discussed above, the consultant's report also analyses the circumstances where industry self-regulation is likely to be most and least effective.

Chapter 6 discusses good practice and cost-effective self-regulatory methods. Good practice in self-regulation can be understood as significantly improving market outcomes for both business and consumers at the lowest cost to businesses. A particular self-regulatory scheme may not be appropriate in circumstances where other forms of regulation are able to provide better outcomes at a lower cost.

Chapter 7 then discusses approaches to promote and co-ordinate self-regulation. The roles of industry, government and consumer groups are dynamic, adapting to the changing face of the Australian economy and, in particular, responding to competitive pressures, regulatory reform, new technologies and the increasing globalisation of consumer markets. This chapter first examines the role that industry has played in promoting and coordinating self-regulatory schemes. It then discusses the role of government as a stakeholder, developer, promoter, monitor and enforcer of schemes, as well as the crucial role that consumer groups have played and will continue to play in the development of industry self-regulation. Finally, the chapter analyses a number of options to co-ordinate and promote self-regulation including discussion of whether a centralised government agency, an oversight committee, or model codes would be appropriate.

Finally, chapter 8 discusses options to facilitate the improvement and harmonisation of dispute resolution schemes. Effective dispute resolution is a crucial element of industry self-regulation offering redress to consumers and it can also identify systemic problems in the industry. However, dispute resolution schemes come at a cost. In particular, they can be costly for small industry groups.

Chapter 2

Framework for the Taskforce Inquiry

At the outset of the inquiry, the Taskforce considered the approach it would take to examining the matters under reference. The scope of the Terms of Reference and the inquiry methodology the Taskforce intended to adopt were set out in the *Issues Paper* released in October 1999.

The Taskforce was established to advise the government on promoting *effective* industry self-regulation. The Taskforce inquiry is occurring in the context of the government's overarching policy on 'making markets work' for the shared benefit of business and consumers. In this context, the Taskforce has defined *effective* industry self-regulation as industry initiatives that significantly improve market outcomes for consumers while reducing compliance costs for business.

Onerous compliance burdens are recognised as detrimental to business. What is less obvious but equally important is that consumers may also 'pay' for sophisticated and costly industry self-regulatory schemes — notably in the form of higher prices. Moreover, industry self-regulation can also have the purpose or effect of inhibiting competition — with serious implications for consumers.⁴

It follows that consumers and businesses have a mutual interest in finding simple and inexpensive mechanisms for resolving market problems.

Principles

- The appropriate form of self-regulation will depend on what is trying to be achieved — that is the way in which it is necessary to significantly improve market outcomes for consumers. This can vary within and between industries.
- The form of self-regulation adopted by industry should be the one which effectively solves the identified problem and minimises costs for industry.

Industry self-regulatory schemes that inhibit competition are at risk of breaching the restrictive trade practices provisions of the *Trade Practices Act 1974*. The Australian Competition and Consumer Commission (ACCC) has the power to authorise such schemes on public benefit grounds, giving the schemes immunity from court action. Details of the ACCC authorisation process are available from the ACCC website at: http://www.accc.gov.au/adjudication/fs-adjudicate.htm.

Reviewing particular schemes

Throughout the inquiry, the Taskforce was encouraged by some stakeholders to appraise particular models of industry self-regulation, to identify problems with particular schemes and to recommend specific improvements, perhaps including underpinning in law. Some stakeholders also anticipated that the Taskforce would identify particular markets where self-regulation should be initiated or where dispute schemes should be set up.

However, while the Taskforce supports regular reviews of self-regulatory schemes, the Taskforce itself did not have the resources or timeframe to be able to conduct meaningful and fair reviews of particular schemes - nor to conduct rigorous assessments of particular market circumstances that might benefit from the development of a self-regulatory regime.

The Taskforce is aware that there have been independent reviews or audits of certain existing code and dispute resolution schemes, each of which has involved a resource commitment comparable to the Taskforce inquiry. The Taskforce was cautious not to draw any conclusions about particular schemes on the basis of a fairly superficial analysis of each scheme.

Instead, the Taskforce has attempted to develop some principles to shape the development and review of schemes in the future. The Taskforce itself was not established to initiate or review individual schemes.

Analytical tools for identifying lowest cost effective options

Internationally, and also in Australia, in line with the drive for efficiency gains, there has been an increasing focus on regulatory reform.

Many of the analytical tools that have been developed to ensure effective regulation by governments can be adapted easily to ensure effective self-regulation by industry, and some of these are discussed below.

The Council of Australian Governments has issued *Principles and Guidelines for National Standard Setting and Regulatory Action.*⁵ While these guidelines were framed for proposed regulatory action by governments, the general principles apply equally to industry self-regulation. Under the COAG Guidelines, the impact of proposed

http://www.pmc.gov.au/briefing/doc/coagpg.pdf.

Council of Australian Governments (COAG) endorsed by COAG in April 1995 and amended in November 1997, *Principles and Guidelines for National Standard Setting and Regulatory Action by Ministerial Councils and Standard-Setting Bodies.* This document is available from the Department of the Prime Minister and Cabinet, including on the Departmental website at:

regulation must be assessed to ascertain that regulation is necessary, and if so, what is the most efficient regulatory approach to use. This assessment should consider:

- > the objective;
- a consideration of alternative approaches;
- the impact on affected groups of proposed approaches;
- a cost/benefit analysis;
- consistency or any proposed approach with international standards; and
- mechanisms for reviewing the proposed regulation.

The option of no action should be considered if it will produce the best outcome for consumers and industry. This option is worth pursuing if a self-regulatory scheme cannot prove that it will improve the situation.

Similarly, the guidelines prepared by the Office of Regulation Review to assist Commonwealth agencies to prepare Regulation Impact Statements (RIS), offered the Taskforce an attractive analytical framework for assessing the effectiveness of self-regulation. The RIS Guidelines also require identification of the problem being addressed, specification of the desired objective(s), identification of options and an assessment of the costs and benefits of each option.

Catalysts for industry self-regulation

Industry self-regulation is increasingly being seen as an alternative means of promoting fair trading, ethical conduct and streamlining compliance with agreed product and service standards in an industry. While industry self-regulation can advance consumer confidence in products and individual companies, it also can promote good business practices.

The Government is encouraging self-regulation because this mechanism is often more flexible and less costly for both business and consumers than direct government regulation. In the government response to the report of the Small Business Deregulation Taskforce, it was made clear that:

The Government is keen for industry to take ownership and responsibility for developing effective and efficient self-regulatory mechanisms where this is appropriate.⁷

Office of Regulation Review 1998, *A Guide to Regulation*, Second Edition. Available on the Internet at http://www.pc.gov.au/orr/reguide2/index.html.

⁷ Statement by the Prime Minister, the Hon John Howard MP 24 March 1997, *More Time for Business*. Available on the Internet at: http://www.dewrsb.gov.au/group_osb/smallbus/moretime.htm.

Properly conceived and drafted, industry self-regulation can be a positive tool for industry and a safeguard for consumers.

It is generally accepted that well functioning markets produce better results for the community. Competition results in greater choice and lower prices for consumers and efficient resource allocation towards more successful suppliers.

Generally with clear information flows, markets provide incentives for business and consumers to resolve many of the problems without intervention.

However, markets can 'fail' to deliver the optimal efficient allocation of resources in the economy for reasons, including the following.

- ➤ The market is characterised by imperfect competition;
- ➤ There is insufficient information available to consumers to allow them to make informed choices;⁸ and/or
- There are high transaction costs for consumers.9

In such circumstances, there is often an incentive for industries to self-regulate.

Markets may also fail to fulfil significant social policy objectives, with the result that the relevant industry may face a choice between government intervention or industry-based initiatives to ensure the market delivers results consistent with those desired by the community.

When industry is confronted with the demonstrated failure of the market mechanism to deliver a problem in the marketplace, the nature and magnitude of that problem must be accurately assessed. Failure to understand the problem may lead to an inappropriate solution being used. Such an outcome may have unintended consequences for many sectors of the community.

This form of market failure is typically referred to as 'information asymmetry' since there is an imbalance in the information available to suppliers and consumers. This does not necessarily imply that suppliers have 'withheld' information that consumers need to make decisions; it may result from the complexity of the transactions involved and the expertise required to understand all aspects of such transactions. Nonetheless, information asymmetries are often remedied by improved information disclosure to consumers. The various forms of market failure are

addressed in the *Codes of Conduct Policy Framework* released by the then Minister for Customs and Consumer Affairs in March 1998; this document is available on the Internet at http://www.treasury.gov.au (choose Consumer Affairs/Publications/Industry Self-Regulation Publications).

High transaction costs refer to the costs of participating in a market and include the costs of searching for relevant information and the costs of obtaining redress if a supplier fails to honour its side of the bargain.

Identifying and quantifying the problem

The types of factors that need to be considered include:

- the causes of the problem;
- who is affected:
- the consequences of the problem for the people affected and the wider community;
- who benefits from the situation and to what extent:
- the scale of the problem; and
- whether it is local, state, national or international.

The presence of market failure or absence of socially desirable outcomes may not be sufficient to justify industry setting up a self-regulatory regime. Inappropriate intervention could create new problems that are greater than the problems it was designed to fix. In particular, a self-regulatory scheme may have an incidental anti-competitive effect, the impact of which is more damaging to consumers than the original market problem.

For this reason, any intervention needs to be weighed up to ascertain whether the extent of the problem is sufficient to justify intervention.

The Taskforce believes that the specific problem and objectives need to be clearly defined before any decision is made about how the desired outcome is to be achieved. Once the decision has been made that intervention is necessary then the focus can properly shift to choosing the most appropriate model of regulation to achieve the desired outcome.

Choosing the appropriate solution from the spectrum of self-regulatory options

In a broad sense, regulation can be considered as a spectrum ranging from self-regulation where there is little or no government involvement, to quasi-regulation which refers to a range of rules, instruments or standards that government expects businesses to comply with, through to explicit government regulation. Further, there is an overarching legal framework governing fair trading, contract, negligence, privacy etc underpinning self-regulation, and the Consumer Law Centre of Victoria made the point that existing self-regulatory schemes can

14 June and 18 July 2000).

Australian Business Limited submitted that self-regulation can enhance compliance with the law by adding detail and industry-specific guidance within a regulatory framework. (Submissions of

¹⁹

sometimes simply establish mechanisms for industry participants to comply with the law.¹¹

There is also a spectrum of self-regulatory options to address market failure and social policy objectives and the art of developing effective self-regulation is to 'customise' solutions to provide optimal outcomes.

The spectrum of self-regulatory options available to industry is a continuum.

- Towards the least costly, least interventionist end of the spectrum are industry agreements to improve the disclosure of information to consumers. Such initiatives may involve voluntary disclosure standards or guidelines, but would be a deliberate attempt to address identified market failure due to consumers making poor choices on the basis of insufficient information. The mere publication of a brochure by an industry association would not constitute 'self-regulation'.
- Somewhere on the continuum are self-regulatory options like customer service charters (that provide information on respective rights and obligations) and voluntary industry codes that provide guidance for members but do not monitor or enforce compliance.
 - Such initiatives may be effective in addressing market failure provided there are commercial incentives for industry participants to comply (or at least an absence of commercial imperatives for industry participants to rely on the market failure).
- At the most interventionist end of the spectrum are industry self-regulatory schemes that basically mirror regulation in that they incorporate industry codes drafted like legislative provisions, mechanisms to ensure compliance by all industry participants, and redress mechanisms to resolve customer disputes.

It is a basic principle of industry efficiency and public welfare that the degree of intervention should be the minimum necessary to achieve the identified objectives. The manner of intervention should be that which imposes the least cost of compliance consistent with achieving the identified objectives.

The advantages of industry self-regulation to address market failure

As discussed in the *Grey-letter Law* report (1997), self-regulatory approaches can effectively remedy market problems, but can be as inefficient as any form of

Submission of 24 July 2000. The Consumer Law Centre of Victoria points out that the Code of Practice for the Fruit Juice Industry is based on trade practices law.

regulation if they do not address the underlying problem.¹² Self-regulation is a means to an end, it is not an end in itself.

Amongst others, the Australian Information Industry Association recognised this, supporting self-regulation that enables industry to respond to rapid technological change, but warning that 'a code that is poorly designed and improperly implemented can actually harm both its proponents and the public'. ¹³ The Consumer Law Centre Victoria also thought it was important to acknowledge that self-regulation can be detrimental to consumer interests if it lowers standards or gives an appearance of legitimacy to questionable market players or practices. ¹⁴

Self-regulation is a viable option if it can improve market outcomes with direct reference to lowering costs to industry participants and providing benefits to both businesses and consumers.

Self-regulatory schemes tend to promote good practice and target specific problems within industries, impose lower compliance costs on business, and offer quick, low cost dispute resolution procedures. Effective self-regulation can also avoid the often overly prescriptive nature of regulation and allow industry the flexibility to provide greater choice for consumers and to be more responsive to changing consumer expectations.

Specific problems can be addressed on an industry wide basis, and so enhance the competitive process. However, it is also necessary to minimise the anti-competitive potential of industry self-regulatory schemes by ensuring that such schemes do not set up barriers to entry to the industry, nor stifle innovation or competition amongst industry participants. Self-regulation may not be appropriate in circumstances where other forms of regulation are able to provide more cost-effective outcomes.

As well as the costs involved in the implementation, administration, monitoring and enforcement of self-regulation, there may be other disadvantages. For example, community cynicism regarding industry regulating itself may lead to a distrust of self-regulation on some issues and the industry as a whole may be blamed for the practices of one or two disreputable firms. Individual firms that are not part of a self-regulatory scheme may also gain commercial advantages from having immunity from sanctions.

As a general guide to whether self-regulation is appropriate, the Taskforce endorses the Commonwealth Office of Regulation Review's Regulatory Impact Statement checklist. The checklist states that self-regulation should be considered where:

there is no strong public interest concern, in particular, no major public health and safety concern;

14 Submission of 24 July 2000.

-

¹² Grey-letter Law: Report of the Commonwealth Interdepartmental Committee on Quasi-regulation, 1997. Available on the Internet at http://www.pc.gov.au/orr/.

¹³ Submission of 20 July 2000.

- the problem is a low risk event, of low impact/significance, in other words the consequences of self-regulation failing to resolve a specific problem are small; and
- the problem can be fixed by the market itself, in other words there is an incentive for individuals and groups to develop and comply with self-regulatory arrangements (e.g. for industry survival, or to gain a market advantage).

However, changes in the industry environment and market developments can effect the conditions underpinning self-regulation. It is important to monitor self-regulation to ensure that it is addressing what it was designed to achieve and to assess whether it is still the most appropriate form of intervention.

It is also evident that there is no one model for self-regulation. The Taskforce considers that good practice in self-regulation involves applying an appropriate scheme to a specific problem or objective. Ascertaining which scheme should be applied will depend on the nature and risk of the problem and the consequences of no action.

Chapter 5 will examine in more detail the industry environments and market circumstances where self-regulation may be appropriate and where it is not.

Chapter 3

Types of self-regulation in consumer markets in Australia

The Taskforce is to inquire into and report on the types of self-regulation in use in consumer markets in Australia.

Within the scope of self-regulation there is a host of options to deal with specific problems and objectives ranging from a simple code of ethics, to schemes incorporating codes that are drafted with legislative precision together with sophisticated customer dispute resolution mechanisms. The various forms of self-regulation at the national level also cover a broad range of industries, including:

- advertising;
- broadcasting and the media;
- direct marketing;
- financial services sector;
- general industry schemes;
- pharmaceuticals and proprietary medicines;
- professional associations;
- retail sector schemes; and
- > telecommunications.

This chapter gives a snapshot of the reasons for, and types of, self-regulation. For further detail on the types of self-regulation, see the directory of self-regulatory schemes at the national level in consumer markets.¹⁵

¹⁵ The directory of self-regulatory schemes can be found at the Taskforce's website: http://www.treasury.gov.au/self-regtaskforce.

Conclusions

- 1. There is a broad and diverse range of self-regulation at the national level affecting consumers.
- 2. There is no single model for industry self-regulation as it depends on what is trying to be achieved.

Reasons for self-regulation

The Taskforce has found that the factors behind self-regulatory schemes range from marketing exercises to legislative requirements. The following section discusses these factors.

Raising industry standards

A common reason for self-regulation, often in conjunction with other reasons, is the desire to raise industry standards. Self-regulation is a means to exceed minimum legal requirements and can also enhance understanding and compliance with regulations. In a competitive environment there is a strong incentive for businesses to continually improve standards and exceed the benchmark service levels in order to gain market share. Various forms of self-regulation can set a benchmark for minimum service levels, and allow businesses flexibility in how these services are to be met and exceeded. For example, the Association of Superannuation Funds of Australia has produced best practice papers on appointment of policy committees, member booklets and annual report checklists.¹⁶

Raising industry standards often refers to the ability to deal with rogue players or poor reputation. The role of reputation can be very important to a business, particularly when the business is operating in a competitive environment. For example, the financial services industry is very competitive and values customer loyalty. One reason as to why the Financial Industry Complaints Scheme was established was the emphasis the industry placed on measures for customer retention and customer satisfaction.¹⁷

Marketing tool

Using self-regulation as a marketing tool is another reason why self-regulation has been developed by industry. Membership of a recognised form of self-regulation (e.g. code of conduct) can often constitute an important selling point for businesses to

¹⁶ Submission number 22, p. 6.

¹⁷ Submission number 15, p. 2.

attract new customers, and may increase the bargaining power of the business when entering new arrangements with other parties. Also, businesses can advertise the fact that they are in a self-regulatory scheme as a means of product differentiation. For example, the Australian Direct Marketing Association Code enables consumers to differentiate between marketplace players.¹⁸

Similarly, in the grains industry, since most of Australia's grain production is exported, one of the means of achieving an edge for Australian grain in the competitive world market is to stress its high quality. Under the new privatised arrangements, the Australian Wheat Board Ltd sets the standards which growers must meet.¹⁹

Enhancing the level of information

Increasing the level of information on products and services is a further reason for self-regulation. By enhancing information flows, businesses can boost consumer confidence in products. The *Code of Conduct for the Provision of Information on Food Products* was introduced to complement regulations and increase consumer knowledge. It provides a degree of consistency in food companies' approaches to labelling and the use of terms to describe food products, thereby providing a greater degree of certainty and confidence to consumers about the nature of the products they are purchasing.²⁰

Self-regulation is also a means of building consumer confidence when introducing new technology to the industry. For example, the *Australian Code of Practice for Computerised Checkout Systems in Supermarkets.*²¹

Threat of government regulation

The actual or perceived 'threat' of government regulation, or a 'push' by government because of poor industry practices was found to be a further reason for industry to self-regulate. For example, the *Code of Banking Practice* was primarily developed by a committee of officials and implemented by Australian banks.²²

Legislative requirements

Self-regulation may have also been imposed via legislative requirements. For example, in the telecommunications industry, Part 6 of the *Telecommunications Act 1997* establishes a scheme of industry codes and industry standards.²³ Also, the

25

¹⁸ Submission number 36, p. 11.

¹⁹ Submission number 28, p. 5.

²⁰ Submission number 30, p. 15.

²¹ Submission number 11, p. 2.

²² Grey-letter Law: Report of the Commonwealth Interdepartmental Committee on Quasi-regulation, 1997, p. XIII.

²³ Submission number 17, p. 1.

Act provides for the Telecommunications Industry Ombudsman Scheme which is an independent alternative dispute resolution scheme. Membership of this scheme is a legislative requirement of all carriers and eligible carriage service providers.

Similarly, under section 123 of the *Broadcasting Services Act 1992*, commercial broadcasting licensees are now required to develop codes of practice in consultation with the Australian Broadcasting Authority.²⁴

Combination of factors

The Taskforce recognises that often there will be a number of reasons to self-regulate. For instance, schemes may have been formed in part due to the threat of government regulation. However, there were probably good reasons why government was threatening to regulate in the first place such as the need to raise industry standards or to increase information flows. For example, the Australian Competition and Consumer Commission (ACCC) assisted the film industry to develop a code to avoid problems in the industry.

Self-regulatory options available to industry

Industry has an array of self-regulatory options available to address specific problems and objectives. These options can range from a simple information campaign to a complex dispute resolution scheme. In addition, there is a spectrum within each type of self-regulatory option.

Information campaign

At the lower end of the self-regulatory spectrum are information campaigns. As part of increasing the flow of information, the Taskforce has found that education of both industry members and the consumer is important. For example, the *Australian Cold Chain Guidelines* seek to strengthen the cold chain by recommending practices for each link, from the manufacturer to consumer to ensure the safety and quality of frozen and chilled foods.²⁵

Also, as discussed elsewhere, self-regulation plays an important role in complementing regulation. For example, the Responsible Serving of Alcohol Program complements the liquor licensing laws. Business and Professional Women of Australia commented that it is in businesses' own interest to support the program as servers of alcohol can be charged for possible manslaughter when serving alcohol to drunken people.²⁶ Similarly, Coatings Care is a voluntary program which companies that deal with paints and coatings may follow to meet regulatory

-

²⁴ Submission number 35, p. 5.

²⁵ Submission number 30, p. 16.

²⁶ Taskforce consultation with Business and Professional Women of Australia, Melbourne, 22 November 1999.

requirements in a manner complementary to their operations. The Department of Industry, Science and Resources submitted that the program assists companies to comply with the diverse requirements for protecting worker health, safety and the environment.²⁷

Service charters

Related to information campaigns, a service charter is a simple and short plain-language document which sets out the quality of service standards customers can expect to receive from that organisation. As discussed in the ACCC's submission, the move for service charters started in the United Kingdom in the early 1990s, but they are becoming increasingly popular in other countries. In Australia, the Commonwealth Government and some State and local governments are implementing charters. Private companies are also starting to follow suit. For example, the AAMI insurance company has recently produced a service charter which sets out some clearly defined rights for its customers. 29

Internal complaints handling departments and procedures

Another form of self-regulation is the use of internal complaints handling departments and procedures. Companies are increasingly looking at ways to gain a competitive edge over their rivals in the marketplace. In particular, bigger companies have an advantage in this area by virtue of their size. Many companies are now internalising consumer protection by establishing corporate consumer affairs departments. For example, the NRMA has a Customer Relations unit that provides services for customers such as interpreters and conference call facilities.³⁰

A further development has been the use of internal complaints handling systems. For example, the Insurance Council of Australia pointed to the *General Insurance Code of Practice* establishing internal dispute arrangements within each insurer.³¹ Generally, before a complaint goes to an external disputes scheme, businesses will try and solve it internally as a first port of call. Standards Australia, which is a non-government body, has also developed a Standard on Complaints Handling (AS4269).³²

In addition, companies have established their own internal/company-based compliance systems. Standards Australia has recently released an Australian standard on Compliance Programs (AS3806).³³ This standard proposes requirements

²⁷ Submission number 31, pp. 11-12.

²⁸ Submission number 42, p. 31.

Taskforce consultation with the Society of Consumer Affairs Professionals in Business, Canberra, 7 December 1999.

³⁰ Submission number 7, p. 4.

³¹ Submission number 18, p. 2.

³² Submission number 42, p. 32.

³³ Submission number 42, p. 31.

and recommendations for the development, implementation and maintenance of a compliance system that can assist an organisation in its compliance with the law.

Accreditation, licensing and membership certification

Accreditation, licensing and membership certifications are a means to create consumer confidence in the level of professionalism and technical competence of members of industry and professional associations. It also serves to set standards within an industry or profession. For example, the accounting profession requires prospective members to complete professional accreditation programs prior to being admitted as either a certified practising accountant or chartered accountant.³⁴

Quality assurance systems

Quality assurance systems (QAS) are another form self-regulation that aim to enhance the quality of a good or service. For example, the increasing adoption of QAS reflects the widespread recognition of the emerging role quality management is playing in world agrifood markets. In the meat and livestock industry, Flockcare is an on-farm QAS introduced by the Sheepmeat Council of Australia to provide a systematic way to ensure producers supply a safe, consistent product while reducing waste and on-farm costs. ³⁵

Standards

Another common type of self-regulation is the use of standards. Many standards are developed to provide a demonstration that certain technical requirements are being met. For example, there are standards in engineering that need to be met before a professional engineer is registered. These standards of competency are measured against a benchmark at the time of first registration and are required to be progressively enhanced through continuing professional development. ³⁶

Australian standards are consensus-based voluntary documents with which compliance is non-mandatory unless incorporated into law or called up in contractual documents. For example, Standards Australia has published four toy safety standards (AS1647 series) which apply to the construction of toys so that the risk of ingestion for children less than three years of age is reduced.³⁷ Presently, only one of these standards for small parts has been declared mandatory.

35 Submission number 28, p. 2.

³⁴ Submission number 33, p. 2.

³⁶ Submission number 26, p. 17.

³⁷ Submission number 23, p. 2.

Codes and dispute resolution schemes

By far the most common form of self-regulation is codes of conduct or codes of ethics that are usually built around membership of a professional or industry association. Codes can range from setting out general statements of principle about how an industry or business will operate, to listing specific business practices which are guaranteed. They can either contain minimum standards or standards, which are, aimed at best practice.

Institutional and functional codes

Codes can also be institutional or functional in their nature. Most codes are institutionally based — that is, codes are industry based. As discussed in chapter 5 — industry environment and market circumstances where self-regulation is likely to be most effective, a major contributing factor to an effective code of conduct is the strength of industry support. However, some products/services can exist across different sectors and industries. Hence a functional self-regulatory scheme covers products/services that span more than one industry. An example of a functional code is the Australian Direct Marketing Association (ADMA) Code. ADMA represents over 400 organisations involved in information-based marketing including financial institutions, publishers, catalogue and mail order traders, Internet-based marketers and service providers, airlines and travel services, telecommunications service providers, and a host of other users and suppliers of direct marketing services.³⁸

Dispute resolution

Codes can also differ in other respects, including whether or not they provide a method of dispute resolution depending on the nature of the specific problems trying to be addressed. For example, the *National Code for the Safe Production of Enzymatic Detergents* allows the manufacture of enzymatic detergents in Australia. Following the production requirements of the code, manufacturers monitor their workplaces using the analytical procedures it specifies to ensure that employees are not exposed to enzyme dust.³⁹ Whereas, there are a number codes that have an accompanying alternate dispute resolution (ADR) scheme. Some of the biggest ADR schemes are the Australian Banking Industry Ombudsman, the Telecommunications Industry Ombudsman, and the General Insurance Enquiries and Complaints Scheme.

Sanctions

Codes may also differ in their level of sanctions for non-compliance. The *Jewellery and Timepieces Industry Code* lists the types of remedial action that can apply when the code is breached such as withdrawal of, or corrective advertising, writing to

³⁸ Submission number 36, p. 2.

³⁹ Submission number 27, p. 9.

consumers, offering consumers a refund, offering alternative merchandise or offering a raincheck.⁴⁰

Compliance

In addition, compliance by industry with a code of conduct can be mandatory or voluntary. For example, section 113 of the *Insurance Act 1973* makes it mandatory for general insurers of certain types of policies to be members of the *General Insurance Code of Practice.*⁴¹ In contrast, the advertising self-regulatory scheme is voluntary in its nature and participants do not have to abide by the Advertising Board's determinations.⁴²

In effect, there is a spectrum of different codes within the codes framework. This re-iterates the Taskforce's finding that there is no one model for self-regulation.

40 Taskforce consultation with the Jewellers Association of Australia, Canberra, 7 December 1999.

⁴¹ Submission number 18, p. 6.

⁴² Submission number 12, p. 2.

Chapter 4

Gaps and overlaps in the coverage of self-regulation

The Taskforce is to inquire into and report on gaps and overlaps in the coverage of self-regulation.

There has been considerable growth in the number of self-regulatory schemes across many industries. In addition, these self-regulatory schemes operate in dynamic markets, which are influenced by globalisation, increasing vertical integration, and the growth of 'hybrid' products that span traditional markets or industries.

As a consequence, gaps and overlaps can emerge in the coverage of various products, services, sectors and industries. For example, a specific problem may emerge from the use of new technology in an industry and may not be covered by any form of regulation. Self-regulation is one means to overcome any specific problems associated with the new technology. Similarly, some self-regulatory schemes may have a degree of overlap where the distinction between products or services has become blurred.

This chapter looks at the broad gaps and overlaps in self-regulation in Australia.

Conclusions

- 3. Gaps and overlaps continually emerge and re-emerge in dynamic markets;
- 4. A 'gap' in the market does not necessarily mean that self-regulation is the appropriate solution;
- 5. Self-regulation is a flexible response to market failure and may fill a 'gap' quickly and efficiently; and
- 6. Some small businesses can have difficulties in joining self-regulatory schemes as can consumers.

Converging sectors and product lines

The Taskforce considers that with increased technology and increasing merger of product lines and new products, there will always be gaps and overlaps emerging. Therefore, it is important to monitor self-regulation to ensure that it is addressing what it was designed to achieve and to assess whether it is still the most appropriate form of intervention.

The financial services industry is an example of an industry that is undergoing rapid and continuous change with new technology and new products. There is also an increasing merger of product lines. For example, some banks are now selling insurance policies under separate entities.

Similarly, in the telecommunications sector gaps and overlaps may increase as technology and discrete industries converge. Presently, separate self-regulatory arrangements currently apply to the telecommunications, Internet and broadcasting industries. The Telecommunications Industry Ombudsman stated that at the moment there are only a few and episodic instances of self-regulatory overlap, although the nature of these overlaps, driven by convergence of both technology and of previously discrete industries, suggest that they will increase. Similarly, the Service Providers Industry Association commented that there are daily examples where such separation is proving problematic, for example in digital television, datacasting, and Internet content. Cable & Wireless Optus also commented that as the technologies of these industries converge, there will be an increasing need for regulatory schemes to respond in a manner which enables industries to deliver market efficiencies.

The Taskforce discusses the role of institutional and functional self-regulatory schemes whilst weighing up the importance of industry ownership in chapter 8 — Options that facilitate the improvement and harmonisation of dispute resolution schemes.

Globalisation

The markets in which businesses operate are also becoming increasingly global, with consumers trading online with merchants all over the world. In particular, Internet use has increased global competition in consumer markets. As a result, self-regulation initiatives need to take into account international as well as domestic industry participants.

⁴³ Submission number 21, p. 7.

⁴⁴ Submission number 25, p. 2.

⁴⁵ Submission number 6, p. 5.

The Australian Consumers' Association stated that consumers are increasingly interacting across jurisdictions and finding examples of good practice. If domestic industry is going to compete in this market it will need to match these practices. ⁴⁶

Similarly, the NRMA considered that with the increasingly globalised nature of many markets and the growth of e-commerce, there is a need for international harmonisation of codes. ⁴⁷ Clayton Utz stated that dialogue between industry bodies in Australia and in overseas markets should be actively encouraged and pursued to develop harmonious self-regulatory schemes which encourage bilateral trade and discourage protectionism where possible. Clayton Utz submitted that subscription to overseas self-regulation schemes by Australian organisations and subscription to Australian schemes by foreign organisations should also be encouraged. The development of international or multi-jurisdictional self-regulation schemes could also occur. ⁴⁸

The Australian Toy Association commented that any imposts or obligations on companies under self-regulatory arrangements should not disadvantage Australian companies via international competition.⁴⁹ Similarly, the National Furnishing Industry Association of Australia commented that its code is becoming a burden in terms of competing against cheap imports.⁵⁰ Harmonisation will help prevent Australian businesses from losing customers to overseas countries that may offer cheaper products but provide less consumer protection through industry self-regulation.

In developing and modifying self-regulatory schemes, the Taskforce considers that the impact of globalisation needs to be considered. The challenge is to implement schemes that provide choice and security for the consumer while enhancing competitiveness of Australian business.

Gaps in the market

Is self-regulation appropriate?

As discussed in chapter 2, the type of self-regulation (or any regulation) should depend on what is trying to be achieved. The following chapter (chapter 5) discusses the general market circumstances and industry environments where self-regulation may be appropriate.

_

⁴⁶ Submission number 16, p. 2.

⁴⁷ Submission number 7, p. 6.

⁴⁸ Submission number 43, p. 4.

⁴⁹ Submission number 23, p. 3.

⁵⁰ Taskforce consultation with the National Furnishing Industry Association of Australia, Melbourne, 23 November 1999.

Hence, a 'gap' in a market where there is no form of regulation may not necessarily mean that self-regulation is the answer. In fact, no regulation, or explicit government regulation could be the minimum effective solution.

The Taskforce stresses that the specific problems and/or objectives need to be clearly specified before any type of self-regulation is considered, then the benefits and costs of ways to deal with the problem can be analysed together with consultation with effected parties.

If self-regulation is the appropriate tool to deal with specific problems and/or objectives, then the Taskforce considers that self-regulatory schemes can be very flexible and responsive and to market circumstances and a changing industry environment. For example, the Telephone Information Services Standards Council was set up by industry in response to consumer complaints about primarily '0055/1900' telephone numbers. The Council has also been modified to include new competitors such as Optus and service providers. ⁵¹

Small business

Some small businesses can find it difficult, or may be unwilling, to be part of a self-regulatory scheme, perhaps because they do not perceive themselves as belonging to a particular industry segment. The Micro Business Network stated that those businesses that are not currently involved in an industry association would find it hard to regulate and many small businesses are anti-regulation from a general viewpoint. Similarly, the Office of Small Business suggested that it is generally accepted that small business is less able than big business to cope with the costs of participating in a scheme such as a code of conduct, particularly when such a scheme is funded by industry levies. The Office of Small Business asserted that small businesses in many cases have no option but to pass on these costs to the consumer in the form of higher prices for goods and services. It argued that this can place small business at a competitive disadvantage to their larger counterparts.

The Office of Small Business commented that although small businesses often have legal recourse in disputes, their access to justice can be constrained by the cost of going to court, delays before their case is heard, the disparity in the quality of representation and their need to preserve business relationships. It suggested that in many cases, neither party achieves a satisfactory result from a Court judgement. ⁵⁴

During its consultations, observers also noted that smaller industry players can find it difficult to influence the development and administration of self-regulatory

_

⁵¹ Taskforce consultation with the Telephone Information Services Standards Council, Sydney, 1 March 1999.

⁵² Submission number 2, p. 1.

⁵³ Submission number 38, p. 3.

⁵⁴ Ibid.

schemes.⁵⁵ Australian Business Limited also commented that some small businesses prefer the certainty of regulation.⁵⁶

Self-regulatory approaches can offer small business a low-cost, quick and flexible system for resolving disputes. Self-regulation can also assist small business to understand and comply with the law.

The Taskforce discusses ways in which self-regulation can be more readily accessible to small businesses in chapter 6 — good practice and cost-effective practice in self-regulation. This discussion includes items such as the funding of schemes not placing businesses at a competitive disadvantage, a transparency of fees, and possible small business representation.

Awareness of codes

Another issue that could be loosely termed as a 'gap' is that consumers may not be aware of various dispute-handling schemes. The Consumer Law Centre of Victoria, Consumer Credit Legal Service of Victoria and the Financial and Consumer Rights Council of Victoria suggested that only a small proportion of Australian consumers are empowered to be able to advocate on their own behalf within the marketplace and challenge unsatisfactory behaviour by industries.⁵⁷ Further, the Law Council of Australia points out that statistics often do not take into account the reasons why and how the disputes are resolved but simply state that they are 'resolved'. The Council argues that many of the disputes are resolved through consumer frustration and ultimate abandonment of complaints.⁵⁸

In addition, the *Consumer Redress Study* (1999) identified the most and least common demographic characteristics of a typical user of a redress mechanism: ⁵⁹

Most Common

Age	Sex	Education Level	Employment Status	First Language	Place of Residence
45-54	Male	At least completed high school	Full-time	English	Metropolitan
35-44	Male	Some tertiary education	Full-time	English	Metropolitan

⁵⁵ For example, Taskforce consultation in Bunbury, 20 July 2000.

⁵⁶ Australian Business Limited, 18 July 2000.

⁵⁷ Submission number 29, p. 12.

⁵⁸ Submission number 19, p. 5.

Consumer Affairs Division, Commonwealth Department of Treasury 1999, *Consumer Redress Study*, released by the Minister for Financial Services and Regulation, the Hon Joe Hockey MP. This document is available on the Internet at http://www.treasury.gov.au (choose Consumer Affairs/Publications/Industry Self-Regulation Publications).

Least Common

Age	Sex	Education Level	Employment Status	First Language	Place of Residence
18-24	Female	Did not complete high school	Part-time	Non-English	Remote area

While the sample data were relatively small, and therefore generalisations from the information should be made with caution, the survey outcomes reflect the experience of the Consumer Law Centre of Victoria, Consumer Credit Legal Service of Victoria and the Financial and Consumer Rights Council of Victoria. Further, the results are echoed in data collected by the Energy Industry Ombudsman of Victoria in relation to the demographics of its complainants.⁶⁰

The Taskforce considers that this stresses the importance of effective consumer access to self-regulation (see further discussion on this in chapter 6 — good practice and cost-effective self-regulation methods and approaches).

Regulatory gaps in the market

A number of organisations brought to the Taskforce's attention the existence of gaps in particular markets or industries.

The joint submission from the Consumer Law Centre of Victoria, Consumer Credit Legal Service of Victoria and the Financial and Consumer Rights Council of Victoria submitted that there are a number of gaps in markets. For example, they submitted that there are gaps in the airline industry, financial services industry, food industry and in telecommunications. They commented that there is a need for urgent and appropriate responses to maintain public faith in co-regulatory processes. ⁶¹

The Australian Securities and Investments Commission (ASIC) also commented that there are gaps in the coverage of formal alternative dispute resolution schemes in the finance services industry including:

- credit (finance companies, building societies and some credit unions);62
- accountants (although accountants that provide financial advice are required to be members of an alternative dispute resolution scheme);
- real estate investments other than real estate managed investments;

61 Submission number 29, pp. 3-9.

⁶⁰ Submission number 29, p. 12.

⁶² The Taskforce notes that under the Uniform Consumer Credit Code, consumer credit is the responsibility of the States.

- some 'transaction only' activities (e.g. on-line share broking services); and
- some cross-border financial services.

However, ASIC commented that it does not believe that all these areas need immediate coverage, but rather that consideration should be given as to whether the absence of coverage by dispute resolution schemes is a reflection of a lack of market problems or because of the difficulties in establishing schemes in such areas. 63

NSW Legal Aid also commented that there are gaps in consumer credit insurance policies.⁶⁴

Further, the Consumer Law Centre of Victoria observed that interest rates have been very high in the pawnbroking industry, and there has been no push by the industry to set up a code of practice. The Centre commented that any self-regulatory scheme should look at the types of consumers using the service. 65

The Taskforce re-iterates that a gap in the market where there is no form of regulation may not necessarily mean that self-regulation is the most appropriate solution, or indeed that any form of regulation is required. The type of intervention (if any) will depend on the nature of the specific problem and should be the effective minimum solution.

Overlap in the market

Overlapping schemes

The Taskforce believes that in the great majority of cases the nature of any consumer complaint is likely to make the appropriate code self-evident. For example, if there are concerns over an advertisement, then the consumer can complain to the Advertising Board. The Taskforce is conscious of the need for a seamless transition between and among codes for the benefit of consumers and to avoid duplication of costs. It is vital to retain the 'one stop shop' approach to complaints handling. From the consumers' perspective, a multiple complaints handling environment can be inefficient, burdensome and frequently frustrating.

However, as discussed above, the growth of different products and changing technology means that there can be multiple schemes in an industry (e.g. financial services industry). For example, ASIC commented that the Financial Industry Complaints Scheme (FICS) and Financial Services Complaint Resolution Scheme had a substantial degree of overlap in the area of complaints about licensees who provide investment advice to retail investors and about responsible entities of managed

Submission number 24, p. 6.

Submission number 37, p. 22. 63

Taskforce consultation with the Victorian Consumer Law Centre, Melbourne, 23 November 1999.

investment schemes. These schemes were merged on 1 January 2000, and will now operate under the FICS banner. ASIC commented that this merger will deliver both cost savings to industry and more consistent complaints handling for consumers. ⁶⁶

In contrast, the Australian Direct Marketing Association suggested that overlapping codes is a strength rather than a weakness of self-regulatory systems. ⁶⁷ Indeed it is arguable that some degree of overlap is necessary in order for these systems to be effective since any significant number of complaints 'falling between the cracks' would tend to bring the whole system into question.

Australian Business Limited disagreed with the notion that self-regulation 'overlaps' in coverage across products, services, sectors or industries, suggesting that the problem stemmed from the detailed and specific nature of self-regulatory schemes. Australian Business suggested that the answer lay in the development of flexible self-regulatory schemes of general application, promoting compliance with regulatory obligations. ⁶⁸

As a general rule, the Taskforce considers that any significant overlap between schemes should be avoided. Further, multiple schemes in the same sector can be confusing for consumers. As pointed out by the NSW Legal Aid it has the potential for members to seek out the scheme that they perceive will be most sympathetic to them.⁶⁹

The Taskforce also encourages industry groups that administer complaints handling systems to have their own networks to ensure complaints are channelled appropriately. These networks need to be constantly nurtured. For example, in the financial services industry, the General Insurance and Enquiries and Complaints scheme stated that consumers are referred between schemes where necessary. This scheme and the Financial Industry Complaints Scheme (FICS), have a direct telephone line between the services so that consumers may be transferred directly when necessary. In addition, these schemes also regularly participate in roundtable meetings with other dispute resolution scheme managers and with ASIC. ⁷⁰

Multiple membership of codes

A related issue to scheme overlap is that companies may be members of more than one code and disputes scheme, and for larger companies this can extend to three or four. For example a major finance company providing banking and insurance services directly to consumers may belong to the Banking Code, the General Insurance Code of Practice and the Code of the Australian Direct Marketing

⁶⁶ Submission number 37, p. 21.

⁶⁷ Submission number 36, p. 10.

⁶⁸ Submission of 18 July 2000.

⁶⁹ Submission number 24, p. 7.

⁷⁰ Submission number 8, p. 3.

Association.⁷¹ Similarly, in the telecommunications industry, some industry players participate in multiple forums, such as the Australian Direct Marketing Association Code, or the Internet Industry Association Code.⁷²

In addition to multiple schemes, there is also often a mix of self-regulatory and regulatory frameworks companies have to comply with, as well as national and State schemes.

In the financial services industry, to the extent that schemes continue to operate along sectoral or industry lines (e.g. general insurance, life insurance and managed investments, banking) the ASIC requirement to join an approved scheme means that there will be circumstances in which participants are members of more than one scheme. For example, a bank that is a member of the Australian Banking Industry Ombudsman will be required to join an approved scheme (currently Financial Industry Complaints Scheme) in respect of its retail advisory activities.

It can be confusing and costly for companies to comply with multiple schemes. However, it is important to recognise industry differences and tailor self-regulation to the circumstances of each industry. Also, there are other factors driving multiple schemes such as the dynamic nature of markets and companies delving into different products and services.

The Taskforce considers that better co-ordination of self-regulatory schemes may allay these concerns. Some approaches to coordinate self-regulation are discussed in chapter 7 — approaches to promoting and coordinating industry self-regulation, including the appropriate role of government.

-

⁷¹ Submission number 18, p. 5.

⁷² Taskforce consultation with Government agencies, Canberra, 6 December 1999.

Chapter 5

Industry environment and market circumstances where self-regulation is likely to be most effective

The Taskforce is to inquire into and report on those industry environments and market circumstances where different types of self-regulation are likely to be most effective.

There is a general recognition that industry self-regulation is often more flexible and less costly for both business and consumers than direct government involvement.⁷³ However, it is necessary to ensure that self-regulation is the appropriate form of intervention given particular industry environment and market circumstances, otherwise inappropriate intervention could create new problems.

There has already been some work done on the market preconditions for effective self-regulation. However, the Taskforce was keen to not only learn from other work that has analysed where self-regulation may be most effective, but also wanted rigorous analysis conducted to test hypotheses. The importance placed on a comprehensive and thorough analysis of market circumstances where industry self-regulation was likely to be most and least effective, became the catalyst for the Taskforce to engage a consultant. The Taskforce required the consultant to structure research around case studies where industry self-regulation had been implemented. The aim of this research was not to identify self-regulatory success and failure in particular industries but to more broadly identify the characteristics of the environment and market that have influenced the effectiveness of self-regulation.

This chapter draws out the Taskforce findings based on a review of existing information, the views of stakeholders collected through the consultation process, and advice from the consultant.

⁷³ See chapter 6.

⁻

⁷⁴ Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective. Its report can be located at the Taskforce webpage at http://www.treasury.gov.au/self-regtaskforce.

Conclusions

Nature and extent of market failure

7. Self-regulation is likely to be most effective where there are clearly defined problems but no high risk of serious or widespread harm to consumers.

Market structure

- 8. An industry environment with an active industry association and/or industry cohesiveness is most likely to administer effective self-regulation as industry participants are more likely to commit financial resources, consult with stakeholders and monitor the effectiveness of self-regulation.
- 9. Self-regulation is less effective where there is a broad spread of smaller businesses that do not communicate with each other.
- 10. Self-regulation is more likely to be effective in a competitive market as industry participants are more likely to be committed to it, either to differentiate their products, or in fear of losing market share.
- 11. A more mature industry may be able to administer more effective self-regulation, as industry participants are more likely to have sufficient resources and be more committed while any 'shakeout' of rogue traders will already have occurred.

Industry and consumer interests

- 12. Self-regulation is likely to be most effective where firms recognise that their future viability depends not only on their relationship with their current customers and shareholders, but also on the wider community.
- 13. The more incentives there are for industry participants to initiate and comply with self-regulation, then the more chance a scheme can remedy specific industry problems.
- 14. The extent to which industry participants are prepared to sign up to a self-regulatory scheme will affect the ability of that scheme to provide effective self-regulation. Where a scheme has a high level of consumer recognition, to the point where consumers will favour scheme participants when making purchasing decisions, then the scheme is most likely to be effective. This will create incentives for non-members to join the scheme.

Conclusions

- 15. The interests of all levels of industry should be considered in the development and maintenance of a self-regulatory scheme, and particularly the level of involvement of smaller businesses where appropriate.
- 16. Where there are cost advantages and/or increased flexibility in self-regulatory initiatives to address specific industry problems compared with government regulation or the court system, then there is a greater chance of improving market outcomes for both business and consumers, and minimising compliance costs for businesses.

Office of Regulation Review checklist

There has already been some work done in identifying industry environments and market circumstances that are more likely to lead to effective self-regulation. In particular, a general guide to whether self-regulation is appropriate is the Commonwealth Office of Regulation Review's Regulatory Impact Statement checklist. The checklist states that self-regulation should be considered where:

- there is no strong public interest concern, in particular, no major public health and safety concern;
- the problem is a low risk event, of low impact/significance, in other words the consequences of self-regulation failing to resolve a specific problem are small; and
- the problem can be fixed by the market itself, in other words there is an incentive for individuals and groups to develop and comply with self-regulatory arrangements (e.g. for industry survival, or to gain a market advantage).

In addition, for self-regulatory industry schemes, the checklist determines success factors to include:

- presence of a viable industry association;
- adequate coverage of the industry by the industry association;
- cohesive industry with like minded/motivated participants committed to achieving the goals;
- voluntary participation effective sanctions and incentives can be applied, with low scope for the benefits being shared with non-participants; and

cost advantages from tailor-made solutions and less formal mechanisms such as access to quick complaints handling and redress mechanisms.⁷⁵

During the inquiry, these observations were often reinforced and expanded upon. The following sections discuss these observations and other industry environments and market circumstances where self-regulation is more likely to be effective.

Nature and extent of market failure

Clearly defined problems and low risk of serious or widespread harm to consumers

The Taskforce considers that self-regulation is most effective where there are clearly defined problems and low risk of serious or widespread harm to consumers. In other words, the consequences of self-regulation failing to resolve a specific problem would not seriously harm consumers. Where there are strong public interest concerns, such as major health and safety issues, and the specific problems are of high risk and/or high frequency then other forms of regulation may be more appropriate.

The Australian Food and Grocery Council commented that self-regulation is most suitable where there is no strong public interest, health or safety concern and the potential market failure would result in an event of low risk.⁷⁶

Cable & Wireless Optus suggested that it is important to realise that the self-regulatory process cannot be used to resolve all competitive and public policy issues that arise within the industry. It submitted where there are significant competition issues or where the commercial interests of carriers with significant power are affected, self-regulation is relatively ineffective in driving policy change expeditiously. Therefore, Cable & Wireless Optus argued that there clearly remains a significant role for statutory bodies in regulating industries particularly those dominated by vertically integrated monopolies with significant market power. PowerTel also commented that there are elements of regulation that must be kept outside of self-regulation such as monopoly control.

In the agriculture, fisheries and forestry industries, the Department of Agriculture, Forestry and Fisheries Australia recognised that there will be circumstances in which self-regulation may not be the most appropriate form of regulation. It commented that the expectations of Australian consumers and customers overseas will see some form of statutory regulation in relation to food safety and in the area of import and

Office of Regulation Review 1998, *A Guide to Regulation*, 2nd edition. This publication is available from the following website: www.pc.gov.au/orr/.

⁷⁶ Submission number 30, p. 3.

⁷⁷ Submission number 6, pp. 2-3.

⁷⁸ Submission number 9, p. 1.

export inspection and the management of agricultural and veterinary chemicals for some time to come. ⁷⁹

The Institution of Engineers commented that explicit regulation rather than self-regulation should regulate the engineering services that result in risk to the health, safety and welfare of the community. For the public, the risk of inadequate engineering depends on their exposure to engineering services. It submitted that every person's lifestyle is dependent on engineering via transport, communications, manufacturing and utilities. 80

Standards Australia commented that a risk based approach should be taken into account when deciding whether regulations should be put into place with the level of regulation being assessed against the potential harm resulting from non-compliance.⁸¹

Similarly, the consultant's report submitted that self-regulation will be more effective where the product supplied is not essential to the welfare of individuals. The consultant's report submitted that the effectiveness of self-regulation as a means of achieving such social welfare objectives depends on the extent to which firms have both the incentive and ability to achieve them. In general, the incentive for firms to self-regulate will be greater, the greater the extent to which those firms stand to benefit from those self-regulatory activities. 82

The Taskforce was interested to learn that self-regulation tends to be more effective in those markets where consumers and other individuals in the community who are adversely affected by market failure share a common interest in eliminating that market failure.⁸³

The Taskforce considers that good practice in self-regulation involves applying an appropriate scheme to a specific problem or objective. Ascertaining which scheme should be applied will depend on the nature and risk of the problem and the consequences of no action. The Taskforce considers that self-regulation is most effective where there are clearly defined problems and a low risk of serious or widespread harm to consumers.

_

⁷⁹ Submission number 28, p. 18.

⁸⁰ Submission number 26, p. 18.

⁸¹ Submission number 13, p. 4.

Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective, p. 38.

⁸³ Ibid, p. 34.

Market structure

Active industry association

A major contributing factor to effective self-regulation is the strength of industry support. Evidence of industry support can be gauged by the existence of an industry association. The ability and willingness of industry to organise itself collectively demonstrates a capacity to undertake self-regulation. An active industry association is most likely to lead to industry participants meeting the schemes' objectives.

The Australian Food and Grocery Council argued that active, well resourced industry associations are critical to providing the organisational structures and processes necessary for effective management of voluntary codes including their development, monitoring and enforcement, and to ensure they provide net benefit and are not unduly restrictive of competition.⁸⁴

The Victorian and Murray Valley Wine Grape Growers Council also agreed that a well-resourced industry association is more likely to provide effective self-regulation.⁸⁵

Similarly, the consultant's report stated that the development of a strong industry association covering the majority of firms in a market can form a solid foundation for effective self-regulation.⁸⁶

The Department of Industry Science and Resources also commented that a self-regulatory regime stands a greater chance of success if it is backed by a large and well structured industry association in a market with few industry participants. Primarily, this assists in regards to the costs associated with establishing and maintaining regimes; ensuring broad participation; and issues of enforcement, including penalties and sanctions. It noted that where no large industry association exists, costs would appear to be a prohibiting factor in developing and administering a regime. ⁸⁷

Australian Business Limited commented that where self-regulation operates within the context of general law such as the Trade Practices Act and the Fair Trading Acts, then it can add detail and industry-specific guidance to help market participants comply with the law and achieve competitive conduct. It commented that for this outcome to occur, strong general law is required accompanied by a group of market participants, usually organised around an industry body, who share a desire to set

85 Taskforce consultation in Melbourne, 13 June 2000.

46

⁸⁴ Submission number 30, p. 3.

Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective, p. 36.

⁸⁷ Submission number 31, p. 22.

some standards of conduct to guide participants and help adjudicate or resolve disputes.88

Insurance Enquiries and Complaints Limited submitted that its scheme and the *General Insurance Code* are effective because of the Australia wide commitment from the industry and the Insurance Council of Australia. It submitted that all industry members selling personal lines insurance are involved, as are consumers, policyholders, consumer groups, Federal Government and the Insurance Council. There is a high level of industry 'ownership' of the code and the scheme, in other words a high level of participation in the setting up, funding and ongoing development of them. ⁸⁹

On the other hand, the Department of Health and Aged Care commented that a challenge for self-regulatory schemes is where there is a lack of effective industry associations or where industry associations misrepresent industry members, or where there is limited commitment to a Code within the industry.⁹⁰

Similarly, in its regional consultations, the Taskforce heard that self-regulation is not conducive to small industries because the industry association does not have the money to promote the scheme.⁹¹ However, the Taskforce notes that the type of self-regulatory scheme will depend on what is trying to be achieved and it should be the one that effectively solves the identified problem and minimises costs for industry.

Micro Business Network commented that those businesses in the micro business sector (including home-based businesses) that are not part of an industry association would find it difficult to regulate because they have few resources and work long hours with often very little capital. It commented that microbusinesses operate in every industry but are difficult to target. 92

ASIC also noted that the need for a strong industry association may be reduced when the self-regulatory scheme has a 'functional' focus rather than institutional coverage. For example, a code of conduct that covers similar products or services regardless of the institutions that offer such products may provide a more commercial basis for membership than that which derives from a common industry perspective. That is, the common interest and commitment may be driven by a desire to meet appropriate market standards rather than the desire to meet the industry association's requirements. 93

The ability and willingness of industry to organise itself collectively demonstrates a capacity to undertake self-regulation. An active industry association and/or industry

90 Submission number 44, pp. 12-13.

47

⁸⁸ Australian Business Limited, 14 June 2000.

⁸⁹ Submission number 8, p. 3.

⁹¹ Taskforce consultation in Tamworth, 28 June 2000.

⁹² Submission number 2, p. 1.

⁹³ Submission number 37, p. 8.

commitment is most likely to lead to industry participants meeting the schemes' objectives.

Cohesive industry

An industry environment where there is a cohesive industry with industry participants committed to achieving their goals is most likely to administer effective self-regulation.

The Department of Health and Aged Care commented that a challenge for self-regulation is where there is diversity within the industry (such as the private health sector) and industry members have diverging rather than converging interests. The Department was conscious, however, that this may be addressed through structured and cooperative education strategies. ⁹⁴

The Association of Superannuation Funds Australia commented that in order for effective self-regulation it is important to have relatively homogeneous objectives and cultures within the industry to reach consensus. ⁹⁵ Similarly, the Institution of Engineers commented that self-regulation requires extensive community and business education, and requires a commitment from all industry players to work effectively. ⁹⁶

The Investment and Financial Services Association commented that industry based complaints schemes rely heavily on the commitment of industry for their success in resolving consumer complaints and building consumer confidence. A sense of 'ownership' on the part of industry participants is essential to maintenance of this commitment and to retention by schemes of their self-regulatory character and effectiveness.⁹⁷

Competitive market

Competitive markets may be more conducive to more effective self-regulation. In a more competitive market, participants are most likely to be committed to it to differentiate their products, or in fear of losing market share.

For example, the NRMA commented that the combination of self-regulation and competitive market forces creates a strong incentive for companies to comply with, and in many cases exceed, the levels of customer service and other conditions that are specified in self-regulatory codes of practice. Non-compliance with the codes by a particular company could see it lose market share to competitors. ⁹⁸

⁹⁴ Submission number 44, p. 12.

⁹⁵ Submission number 22, p. 3.

⁹⁶ Submission number 26, p. 23.

⁹⁷ Submission number 41, p. 4.

⁹⁸ Submission number 7, p. 3.

NRMA commented that the main markets that it operates in, namely general insurance and financial services, are characterised by intense competition between a significant number of industry participants, and this creates a strong incentive for companies to use adherence to codes as a marketing tool. It submitted that competition is also becoming more intense as new distribution channels such as e-commerce emerge.⁹⁹

Similarly, ASIC commented that self-regulation is more likely to be effective in a competitive market, as it will lessen the risk of such regulation becoming an anti-competitive structure.¹⁰⁰ However, as noted above, this may also mean that the achievement of industry common interest can be more difficult.

The consultant's report also commented that self-regulation tends to be more effective in those markets where there are relatively large numbers of competitive firms producing relatively homogeneous products. In such markets, firms can reap significant economies of scale by grouping together to self-regulate the activities of those firms within that group that impose costs on consumers, other firms within that group and the wider community. In addition, because of the homogeneous nature of the product there is a much greater probability that the external costs generated by one firm will adversely affect the sales of other firms producing those goods. ¹⁰¹

On the other hand, the Australian Consumers' Association considered that self-regulation works well when there is a small number of large players (such as banking or insurance) as opposed to a large number of small players (financial planners). ¹⁰²

Similarly, during Taskforce consultations some industry associations commented that fewer people or strong leadership makes it easier to self-regulate as industries can get the level of detail they desire rather than having generic codes.¹⁰³

The consultant's report noted that self-regulation is less likely to be effective in those markets that are dominated by a very small number of firms due to the existence of large economies of scale in production. In these markets, the firms are more likely to share a common interest in using self-regulation as a means of reducing, rather than increasing, the amount of competition between firms. In fact, in such cases, the ACCC is likely to consider such self-regulation to be anti-competitive. ¹⁰⁴

⁹⁹ Ibid.

¹⁰⁰ Submission number 37, p. 9.

¹⁰¹ Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective, p. 31.

¹⁰² Submission number 16, p. 1.

¹⁰³ Taskforce consultation with the Fund Raising Institute, Sydney, 30 November 1999.

¹⁰⁴ Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective, p. 31.

The Taskforce considers that self-regulation in a competitive market with homogeneous products is most likely to produce effective self-regulation.

Mature industry

Maturity in the market may be another factor underpinning the effectiveness of self-regulation. The Taskforce considers that a more mature industry with established players may be more willing and able to participate in self-regulation.

For example, the Australian Consumers' Association commented that voluntary self-regulation can have little effect where there are 'cowboys' who are not prepared to participate. It argued that in the telecommunications industry, there are some companies who participate in self-regulation and an increasing number of smaller players who do not. Again, it is a comparatively recently deregulated market. It argued that consumers using these companies have limited access to redress and do not enjoy adequate consumer protection. ¹⁰⁵

However, the Taskforce recognises that it depends on what self-regulation is trying to achieve. For example, a new industry may develop a self-regulatory scheme to develop consumer confidence.

Generally, a more mature industry with established players may be more willing and able to participate in self-regulation.

Industry and consumer interests

Wider community awareness

Self-regulation is likely to be most effective where firms recognise that their future viability depends not only on their relationship with their current customers and shareholders, but also on the wider community.

The consultant's report submitted that many medium to larger firms now recognise that their longer term profitability and viability, and their potential to attract new customers and investors, does not depend solely on how they are viewed by their current customers and shareholders. It also depends on how their activities are viewed by the wider community and the government, who may have a significant influence on their future sales, sources of funds, profitability and the regulatory environment. As a result, those firms are investing considerable amounts of time and money in developing their reputations as socially responsible corporations.

-

¹⁰⁵ Submission number 16, p. 1.

Introducing self-regulation can be an important factor in improving their corporate image. 106

The report observed that some self-regulatory codes not only try to improve market efficiency, but also seek to achieve a number of social welfare objectives. For example, the accountants' Code of Professional Conduct not only requires that members must safeguard the interests of their clients and employers, but also that they must not be in conflict with duties owed to the community and its laws. The Taskforce recognises that the failure of firms to act in a manner consistent with society's broad social objectives can have a damaging effect on their overall reputation and profitability, and that this provides a real incentive to implement effective self-regulation.

Incentives to make self-regulation effective

For industry self-regulation to be effective, there needs to be some vested interests or incentives to make it so. In other words, generally self-regulation needs to be in the self-interest of industry to not only occur, but also to be effective. The more incentives for businesses to make self-regulation work, then the more chance that self-regulation will be effective in achieving improved market outcomes for consumers.

The Australian Food and Grocery Council commented that the strongest incentive for industry to ensure that self-regulation is effective is the imperative of the industry as a whole and individual companies, to protect their reputations in the marketplace. It submitted that once a self-regulatory measure is established, and promoted by the industry as a commitment to a set of values and a desire to meet the needs of consumers, individual companies and the industry as a whole will strive to meet the benchmarks it has set.¹⁰⁸

The Consumers' Telecommunications Network commented that self-regulation will work when there is a substantial identity of interests with common benefit between carriers and an equal bargaining power of parties. It gave the example of 'end to end' network performance (quality of phone call of both ends will be the same) working well because everyone has a common benefit. 109

During the Taskforce consultations, the Financial Services Consumer Policy Centre commented that there were three reasons why industries self-regulate, namely:

1. threat of government regulation;

_

¹⁰⁶ Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective, p. 24.

¹⁰⁷ Ibid, p. 24.

¹⁰⁸ Australian Food and Grocery Council, 21 July 2000.

¹⁰⁹ Taskforce consultation with Consumers' Telecommunications Network, Sydney, 30 November 1999.

- 2. promotional opportunity; and/or
- 3. a means for product differentiation. 110

The Australian Consumers' Association was of a similar mind, submitting that there may be a 'carrot', such as the opportunity to differentiate a company by adhering to a code (especially if there are 'cowboys' in the market), or a 'stick' (such as industry, media or consumer pressure). ¹¹¹

ASIC was of the view that common interest will usually involve a mix of 'positive' and 'negative' incentives. An example of positive common interest is a desire to improve professionalism in the industry. ASIC argued that improving consumer confidence can have both positive and negative elements. Most self-regulatory schemes also seek to improve consumer confidence where it is currently inadequate such as in areas of new technology or where there have been problematic industry practices. ¹¹²

ASIC also submitted that negative self-interest can include the desire to avoid government, which may be an explicit 'threat' via direct government pressure or an implicit 'threat' arising out of the general direction of government policy reforms. At its best, this involves a genuine commitment to self-regulation that will deliver market improvements in a cost-effective manner. ASIC noted that, in the past, comprehensive self-regulatory schemes have generally been developed only where there has been a real threat of government or regulator intervention. ¹¹³

The consultant's report noted that 'external' costs and benefits may not be taken into account by firms and consumers when determining how much they should produce and consume. The effectiveness of self-regulation depends on the extent to which firms have the incentive and the ability to 'avoid' external costs or 'internalise' external benefits and costs. 114

Where there is no or little common interest, then it is harder to make self-regulation work. For example, PowerTel commented that one element that must be kept outside self-regulation is where significant conflicts of interest are likely to result from the self-regulatory process.¹¹⁵

During Taskforce consultations, it was evident that telecommunication codes that deal with commercial interest are being developed a lot faster than consumer codes, because of the self-interest by carriers.

¹¹⁰ Taskforce consultation with the Financial Services Consumer Policy Centre, as part of a consumer group roundtable discussion, Sydney, 30 November 1999.

¹¹¹ Submission number 16, p. 1.

¹¹² Submission number 37, p. 8.

¹¹³ Ibid.

¹¹⁴ Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective, p. 18.

¹¹⁵ Submission number 9, p. 1.

The Service Providers Industry Association (SPAN) commented that instances abound where the rate of progress on important regulatory/self-regulatory initiatives has fallen well below industry aspirations and expectations. Examples demonstrate that a self-regulatory model is ill suited to any situation where there are conflicting commercial interests to be reconciled and particularly where the bargaining power and information available to the parties are unbalanced. ¹¹⁶

SPAN commented that reasons for delay include the natural tendency of engineering and legal/regulatory people, who make up the self-regulatory workforce, to want to tease out all elements of technical complexity and risk in the process of code formulation. This attitude is understandable, but is tempered in normal business conduct by the imperative of targets and deadlines set by executive management to achieve goals such as time-to-market advantage over competition. That discipline is either absent or given insufficient weight within the self-regulatory framework. SPAN noted that a further reason for delay often quoted is the attitude of incumbent operators whose commercial interests are to perpetuate the status quo as long as possible.¹¹⁷

The Consumer's Telecommunications Network was concerned that, often, quality of service and profitability do not go hand-in-hand. During regional consultations, the Taskforce also heard that business-to-business industry self-regulation is difficult to introduce when there is a power imbalance between producers and buyers. 119

The consultant's report also noted that self-regulation is likely to be less effective in markets where firms, consumers and the wider community do not share a common interest in reducing the market failure. ¹²⁰

For industry self-regulation to be effective, then there needs to be incentives to make it work. The more cohesive an industry is with incentives to make self-regulation work, then the more chance that self-regulation will be effective and meet its objectives.

Market incentives

The existence of market incentives to comply with self-regulatory schemes are most likely to increase the effectiveness of self-regulation. An industry environment where self-regulation may be most effective is where there is voluntary participation with effective sanctions and incentives to ensure that there is little scope for non-participants in the scheme to enjoy the benefits.

_

¹¹⁶ Submission number 25, p. 3.

¹¹⁷ Ibid

¹¹⁸ Taskforce consultation with CTN, Sydney, 30 November 1999.

¹¹⁹ Taskforce consultations in Bunbury, 20 July 2000, and with the Victorian and Murray Valley Wine Grape Growers Council, 13 June 2000.

¹²⁰ Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective, p. iv.

For example, the Department of Industry, Science and Resources commented that voluntary participation—backed by strong incentives to participate—appears to provide a stronger framework and higher degree of success, independent of the size of the industry association. Whereas, it commented that mandatory participation and subsequent issues of compliance, enforcement, penalties and/or sanctions appear to depend primarily on the size and strength of the industry association. ¹²¹

The Institution of Engineers also commented that effective self-regulation requires not only standards or codes of practice, but also effective mechanisms for dealing with complaints with these codes.¹²²

The Association of Superannuation Funds Australia commented that for self-regulation to be effective industry requires effective enforcement and sanctions, for example the standards need to be well known and/or branded. If an industry member fails to meet these standards then some sanction is required (e.g. fine, 'shaming' or corrective advertising restrictions on licence to operate). It argued that such enforcement also requires/assumes that effective complaint procedures are easily accessible for consumers. ¹²³

During the consultations, one organisation commented that public shaming is like being 'dumped into custard — it is a soft landing, but it sticks'. 124

Similarly, NRMA commented that self-regulation also functions effectively in industries where brand name image and customer loyalty are important determinants of market share and profitability. Any damage to brand reputation through non-compliance with a code of practice could be very costly to restore. ¹²⁵

On the other hand, the Motor Trades Association of Australia commented that industry self-regulation has not been effective in relation to Franchising and Oilcode because sanctions are ineffective because offending parties can simply 'drop out' of the scheme and continue the offending behaviour. It argued that if there is no penalty or detriment for non-participation, then many will question why they should join. ¹²⁶

The Australian Consumer's Association argued that there is need for government underpinning of self-regulation. It considered that the involvement of ASIC in approving codes and dispute schemes, as part of the new regulatory framework for the financial sector being introduced as part of the Corporate Law Economic Reform Program, will ensure that self-regulation 'best practice' principles become legislative requirements. ¹²⁷

¹²¹ Submission number 31, p. 22.

¹²² Submission number 26, p. 15.

¹²³ Submission number 22, p. 3.

¹²⁴ Taskforce consultation with the Australian Press Council, Sydney, 1 December 1999.

¹²⁵ Submission number 7, p. 3.

¹²⁶ Submission number 1, p. 1.

¹²⁷ Submission number 16, p. 1.

The existence of market incentives and effective sanctions is most likely to increase the effectiveness of self-regulation, where participants will comply more with schemes, which it turn, can improve market outcomes for consumers.

Adequate industry coverage

An important element of self-regulation is coverage. The extent to which industry participants are prepared to sign up to a self-regulatory scheme will affect their ability to provide effective self-regulation. Where a scheme has a high level of consumer recognition, to the point where consumers will favour that scheme, then the scheme is most likely to be effective. There then may be market pressures for other industry participants to join the scheme.

A significant number of organisations commented that wide coverage was an important element of good practice. For example, the Financial Industry Complaints Service considered that self-regulation only works where the whole segment of a particular industry is covered by one scheme and rules are uniform. Similarly, the Association of Superannuation Funds Australia commented that for self-regulation to be effective, then it is necessary to have close to 100 per cent coverage of industry participants.

The Telecommunications Industry Ombudsman stated that its success has been due to its ability to maintain an appropriate level of consumer protection in a rapidly changing competitive environment, with more than substantial coverage of the telecommunications industry. A measure of the extent of this coverage is evident by the increase in membership from three members at its inception to over 900 members at the date of its submission. ¹³⁰

Further, ASIC commented that wide industry coverage may be easier to achieve in an industry with fewer and larger organisations as the problem of 'free riders' is less apparent. It stated that free rider problems can be of two broad types:

- Industry members may choose not to join the self-regulatory scheme at all. This can be a particular problem in those industries characterised by a large number of firms including many smaller firms or individual practitioners; and
- > Industry members may join the self-regulatory scheme, but choose not to properly adhere to the agreed rules.

ASIC argued that in both cases the free riding firm may gain competitive advantages by enjoying the public benefits of self-regulation while not bearing its costs. ¹³¹

129 Submission number 22, p. 3.

¹²⁸ Submission number 15, p. 3.

¹³⁰ Submission number 21, p. 8.

¹³¹ Submission number 37, p. 9.

The consultant's report also submitted that self-regulation is more likely to be effective where there is limited scope for adversely affected individuals and firms to 'free ride' on the benefits of self-regulation.¹³²

The Office of Small Business commented that a contributing factor to the success of self-regulation is the extent to which all levels of an industry participate in its development and maintenance. It asserted that it is generally accepted that smaller businesses may face proportionally higher costs, both in terms of time and resources, when seeking to provide input into regulatory regimes. It argued that this can lead to a lack of smaller business participation in and ownership of self-regulation, and a consequent lack of commitment by such businesses to the success of a scheme. Hence, the Office of Small Business commented that is important to ensure that smaller businesses are adequately represented in the development and administration of self-regulation. ¹³³

The Office of Small Business submitted that the manner in which smaller businesses may be accommodated will vary with each particular circumstance. It commented that an industry association may actively seek the input of all industry levels, alternatively it may be necessary for bodies set up to administer a self-regulatory regime to consult with smaller businesses. The Office of Small Business also commented that it is possible the development and administration of self-regulation will by itself adequately involve all levels of industry.¹³⁴

The effectiveness of any self-regulatory scheme will only be as good as the extent of its coverage. The extent to which industry participants are prepared to sign up to a self-regulatory scheme will affect the ability of them to provide effective self-regulation.

Cost advantages and/or increased flexibility

An industry environment where there are cost advantages and/or increased flexibility in developing and maintaining self-regulation compared with government regulation or the court system can underpin more effective self-regulation. Cost advantages, for both business and consumers, could include less formal mechanisms such as quick complaints handling and redress mechanisms.

The Federation of Australian Radio Broadcasters stated that self-regulation is usually faster and less expensive as well as more flexible and up-to-date than government regulation because industry has a better understanding of the problems and what their realistic solutions are.¹³⁵

¹³² Tasman Asia Pacific 2000, Analysis of market circumstances where industry self-regulation is likely to be most and least effective, p. 38.

¹³³ Office of Small Business, 5 July 2000.

¹³⁴ Ibid.

¹³⁵ Submission number 35, p. 1.

Similarly with court costs, a number of organisations commented that self-regulation is quicker and cheaper. For example, the Financial Industry Complaints Service Limited stated that it is a fact that dispute schemes provide a cheaper service than the courts and hopefully greater consumer retention for the industry involved. ¹³⁶

Echoing these comments, the Investment and Financial Services Association stated that in the Australian financial services sector, dispute resolution schemes have evolved as cost-effective alternatives to litigation. They have reduced pressure for government intervention in the industry, especially where such schemes have entailed complete market coverage. ¹³⁷

The Insurance Council of Australia commented that its self-regulatory initiatives have worked effectively because of the absence of any overlay of legalism or formality (such as an appeals process) which makes the Scheme accessible and attractive to consumers.¹³⁸

NRMA commented that an environment where an industry's products and distribution channels are undergoing continual change is also better suited to self-regulation rather than formal government regulation. This reflects the fact that self-regulation will generally be more flexible and adaptable to changing circumstances such as new technology and new products. The NRMA submitted that financial services is an example of an industry that is undergoing rapid and continuous change. ¹³⁹

Where there are cost advantages and/or increased flexibility in developing and maintaining self-regulation more effective self-regulation may be achieved.

Conclusion

In the course of the inquiry, it was suggested to the Taskforce that it should draw conclusions about where self-regulation is likely to succeed and where self-regulation has a poor prognosis and 'black letter law' should be the preferred option. ¹⁴⁰

However, the Taskforce ultimately concluded that the analysis of whether self-regulation is appropriate must be on a case by case basis. In some circumstances, the way in which a self-regulatory scheme is designed and implemented may overcome inherent handicaps in industry structure or market circumstances.

137 Submission number 41, p. 2.

¹³⁶ Submission number 15, p. 3.

¹³⁸ Submission number 18, p. 6.

¹³⁹ Submission number 7, p. 3.

¹⁴⁰ For example, Mr Randal Dennings, Clayton Utz, proposed that the Taskforce produce 'a matrix of factors for use as a guide by parliaments and industry bodies so as to isolate the desirable regulatory response geared to the governmental outcomes required' (Submission of 18 July 2000).

Nonetheless, the Taskforce believes that the findings of its report should be incorporated into existing guidelines for industry and policymakers.

The Taskforce encourages the government, in addition to existing guidelines and benchmarks, to provide industries with practical guidelines based on the principles flagged in this report to help inform/assist the development and review of self-regulatory schemes.

The Taskforce also encourages the government to consider up-dating its guidelines for policy makers involved in assessing options on the industry environment and market circumstances that are most likely to lead to effective self-regulation, based on the findings in this report.

Chapter 6

Good practice and cost effective self-regulation methods

The Taskforce is to inquire and report on best practice and cost effective self-regulation methods and approaches.

The Taskforce believes there is no single 'best practice' model for self-regulation because a successful model depends on particular market characteristics and needs to be designed accordingly. However, it is possible to identify critical elements of schemes which individually or collectively have underpinned successfully operating schemes. A number of these elements are identified through the report.

Good practice in self-regulation can be understood as significantly improving market outcomes for consumers at the lowest cost to businesses. A particular self-regulatory scheme may not be appropriate in circumstances where other forms of regulation are able to provide better outcomes at a lower cost. For example, the costs involved with a complex customer dispute resolution mechanism may not be justified if the scheme only receives a few complaints per year. Further, the costs involved in administering such a scheme may be translated into higher prices for consumers so, in this case, would not constitute a better market outcome for either business or consumers.

Principles

- Good practice in self-regulation involves addressing industry specific problems and objectives.
- ➤ The type of self-regulatory scheme should be the effective minimum solution.

Conclusions

Consultation

17. Consultation between industry, consumers and government can help ensure that specific problems and social policy objectives can be identified and addressed.

Coverage and publicity

- 18. Increased industry coverage of schemes ensures that the benefits from standards of practice in schemes flow to consumers. Wide coverage also ensures that consumers can identify self-regulatory schemes.
- 19. Clarity in the schemes' documentation can help industry understand their obligations and assist dispute schemes interpret legal rights. Clarity can also help consumers understand their rights.
- 20. Consumer awareness of schemes ensures that consumers know where to lodge complaints. Schemes are encouraged to make use of new technologies such as the Internet, make complaints cost free to the consumer, write sample letters of complaint, take oral complaints, provide personal contact and transfer complainants between schemes.
- 21. Industry awareness campaigns and education about schemes is needed to make sure industry participants understand their obligations and, where appropriate, understand the consequences of failing to abide by these obligations.

Administration

- 22. A good administrative body can identify issues, collect data, monitor the scheme, enhance credibility and ensure compliance costs are at an effective minimum level.
- 23. Data collection by an industry scheme is a valuable tool in identifying systemic issues and allows industry to address these problems, which in turn, can improve market outcomes for both businesses and consumers.
- 24. As consumers cannot guard against specific industry problems that they do not know exist, transparency in schemes is an important mechanism to ensure credibility and accountability.

Dispute procedures and sanctions

- 25. Industry adherence to self-regulatory schemes is essential to ensure that the benefits flowing from the standards of practice set by schemes are passed onto the consumer.
- 26. Where the standard of conduct has been breached, self-regulatory schemes should incorporate complaint handling and dispute resolution mechanisms to provide appropriate redress to consumers. The appropriate redress mechanism will depend on the nature of the specific problem and the consequences of non-compliance.
- 27. A range of sanctions can be used by industry in order to achieve compliance depending on the nature of the specific problem and consequences of non-compliance. The severity of the sanction should depend on the seriousness of the breach.
- 28. Industry needs to manage the risk of any anti-competitive practices in schemes, particularly where sanctions are involved.

Monitoring and reviewing

29. Monitoring of self-regulation is essential to ensure that it is still relevant to the industry addressing specific problems and improving market outcomes. In this context, reviews and annual reporting are important tools for monitoring schemes and can also assist in the transparency and accountability of schemes. Preferably, reviews should be periodic, independent and the results made publicly available.

Cost-effectiveness

- 30. Self-regulation comes at a cost, in administration, promotion and compliance. However, self-regulation can be cheaper (in terms of compliance costs) and more flexible than government regulation and the court system. Ultimately, the consumer bears the cost of regulation in most cases, as it is part of a firm's cost structure.
- 31. Any funding arrangement for self-regulation should be transparent and designed so as not to put businesses at a competitive disadvantage through excessive compliance costs.

Good practice in self-regulation

Self-regulation is very broad and covers guidelines, quality management systems, standards, codes, dispute resolution schemes etc. Although there is no one model for good self-regulation, the Taskforce considers that there are elements of good practice that are consistent amongst schemes.

Consultation

Addressing specific problems and objectives

The form of self-regulation adopted by industry should be the effective minimum solution to the specific problem to minimise compliance costs for business.

The Australian Chamber of Commerce and Industry argued that self-regulation allows industry to respond to concerns raised by consumers and identify solutions to problems by utilising the resources and expertise unavailable to government. It commented that under self-regulation, industry (often through associations) could assume responsibility for concerns raised by the community and is able to interact directly with stakeholders to resolve the problem. 141

The Australian Chamber of Commerce and Industry also submitted that self-regulation enables commerce and industry to respond more efficiently and effectively to the changing concerns of consumers. It will also empower users, whether business or householders, through the market-mechanisms. 142 In the telecommunications industry, the Telecommunications Industry Ombudsman commented that good practice in self-regulation involves the ability to address specific problems which affect consumers. Reduced to the most basic issue, the problems facing consumers are those that involve the transition from a previously monopolistic environment in telecommunications to one of open, but still regulated, competition. 143

The Telecommunications Industry Ombudsman commented that consumers have been faced with difficulties of choice not only amongst many more providers, but also amongst many new services and products with differing prices, as well as with relatively new technologies such as mobile communications and the Internet. These difficulties have contributed to disputes about bills — the single highest area of complaint to their scheme. It argued that it has resolved these types of complaints

¹⁴¹ Submission number 27, p. 5.

¹⁴² Ibid.

¹⁴³ Submission number 21, p. 9.

and has also highlighted systemic problems within industry forums and individual members.¹⁴⁴

Similarly, the Federation of Australian Commercial Television Stations submitted that the industry considers that the code approach is more effective and efficient than regulation in achieving public interest objectives, in terms of flexibility, responsiveness to community views, transparency and ease of use by stations and viewers alike.¹⁴⁵

Consultation

As touched on above, although self-regulation is the responsibility of industry, both consumers and the government are stakeholders. The Taskforce considers that consultation is not only important to ensure credibility of a scheme, but consumers can help identify specific problems within an industry and government can identify social or public policy objectives.

The Australian Chamber of Commerce and Industry submitted that a key element of managing self-regulation is the establishment of structures involving stakeholders to facilitate resolution.¹⁴⁶

The ACCC argued that if codes of conduct/self-regulation are going to be accepted by governments and the public at large, then credibility with stakeholders is absolutely vital, because only with such credibility will there be public acceptance of the code or an industry-based scheme and commitment to it by the appropriate regulators. It argued that to have any credibility at all there needs to be consultation with the appropriate consumer/community/user groups and appropriate regulatory/government agencies, as well as industry members. 147

During the Taskforce consultations, a number of schemes commented that they consult with consumer groups. The Financial Industry Complaints Scheme commented that some industry associations offer forums for consumers but most meet with recognised consumer groups. The Financial Industry Complaints Scheme commented that some industry associations offer forums for consumers but most meet with recognised consumer groups.

The Australian Pharmaceutical Manufacturers Association commented that it has close relationships with the Therapeutic Goods Administration and the ACCC. In addition, it maintains close relationships with prescribers and allied organisations to monitor the code and to seek comment and suggestions for its improvement.

-

¹⁴⁴ Ibid.

¹⁴⁵ Submission number 34, p. 10.

¹⁴⁶ Submission number 27, p. 12.

¹⁴⁷ Submission number 42, p. 25.

¹⁴⁸ For example, Taskforce consultations with the Australian Banking Industry Ombudsman and Insurance Brokers' Dispute Facility, Melbourne, 22 November 1999.

¹⁴⁹ Submission number 15, p. 4.

Consumers are also viewed by the Association as a key stakeholder with the Consumer Health Forum being represented on the Code of Conduct Committees. ¹⁵⁰

Standards Australia stated that it uses the internationally accepted principle of preparing standards that involve transparency and consensus. This includes:

- ➤ the use of committees to represent all relevant stakeholders;
- the issue of drafts for public comment (usually for 60 days); and
- > approval for publication by a consensus of all relevant stakeholders. 151

Similarly, the Australian Communications Authority commented that when assessing a code for registration it must be satisfied that consultation has been undertaken with the public, the industry, the ACCC, Telecommunications Industry Ombudsman, a consumer representative organisation, and for privacy codes, the Privacy Commissioner. ¹⁵²

However, the Taskforce recognises that consumer interests are typically diverse and highly dispersed in comparison to industry interests. ASIC commented that because of this it would usually be more difficult for consumers to independently generate sufficient resources and expertise to provide effective input to the full range of self-regulatory mechanisms without additional assistance. Importantly, ASIC argued that such input is required not only during the development of self-regulatory mechanisms, but also during the ongoing life of such mechanisms to ensure adequate accountability, appropriate independence and continuing relevance. ¹⁵³

Similarly, the Australian Consumers' Association commented that codes and self-regulation are drafted by the supply side of a market. Industry has the resources to create complex analysis and reports. Whereas, it argued the demand side, as represented by the consumer movement, have fewer resources to provide the same level of input. So while the opportunity to consult is there, the means are limited.¹⁵⁴ NSW Legal Aid also commented that these activities are labour intensive.¹⁵⁵

The Office of Small Business also argued that any code of conduct or self-regulatory mechanism needs to have genuine small business representation on code councils (or similar style bodies). It argued that there is an equal need to ensure the same representation for small business as for other affected groups such as consumers. ¹⁵⁶

The Taskforce supports the proposition that self-regulation should be developed and maintained in partnership between industry, the regulator and consumer

¹⁵⁰ Submission number 10, p. 1.

¹⁵¹ Submission number 13, p. 2.

¹⁵² Submission number 17, p. 4.

¹⁵³ Submission number 37, p. 13.

¹⁵⁴ Submission number 16, p. 2.

¹⁵⁵ Submission number 24, p. 6.

¹⁵⁶ Submission number 38, p. 4.

organisations. This partnership is essential to identify specific problems and to arrive at effective minimum solutions. The Taskforce also recognises the important role that consumer groups can play in self-regulation development and growth. ¹⁵⁷

Coverage and publicity

Adequate coverage

An important element of self-regulation is coverage.¹⁵⁸ The extent to which an industry association represents the majority of industry participants affects the ability of the association to deliver effective self-regulation. If schemes do not have wide industry coverage, then fewer consumers will enjoy improved market outcomes. The Taskforce considers that the effectiveness of any self-regulatory scheme will only be as good as the extent of its coverage.

Clarity

An industry seeking to self-regulate must be able to establish certain standards of conduct which members will support. This will ensure that members understand their obligations and consumers will be aware of their rights. As noted above, the standards will differ according to the specific problem and the industry concerned.

During Taskforce consultations, some organisations commented that self-regulatory schemes need to be written in a plain language that both consumers and industry can understand.¹⁵⁹

In relation to codes, the ACCC argued that in order to be accepted by all stakeholders it is important for the code to be drafted in a technically legal sense but it is even more important for the language to be plain and understandable to all of its readers. Clarity in the document will instil more confidence and certainty whilst any ambiguity or vagueness will militate against acceptance, support and compliance. This is particularly important in the area of understanding obligations and allowing for enforcement. ¹⁶⁰

As suggested by the ACCC, the code should set out clearly stated reasons why the code was established and what are the intended outcomes. To be effective in addressing consumer concerns a code needs to have rules which address common complaints and concerns about industry practices and which set performance

_

¹⁵⁷ Chapter 7 — approaches to promoting and coordinating self-regulation, looks at the role of consumer groups and government in more detail.

¹⁵⁸ Chapter 5 indicates that if a scheme does not have adequate coverage, then it has little chance of being successful.

¹⁵⁹ For example, Taskforce consultation with the Society of Consumer Affairs Professionals in Business, Canberra, 7 December 1999.

¹⁶⁰ Submission number 42, p. 25.

standards for participants. Such rules should address specific stated problems and not be written as broad general principles.¹⁶¹

As a general rule, the Taskforce considers that standards in a self-regulatory scheme ought to be specific and written in a plain and easy to understand manner. This will ensure that consumers understand their rights and industries understand their obligations. Clarity in the documentation will also assist dispute schemes interpret legal rights.

Consumer awareness

Self-regulation needs to be promoted and consumers need to be aware of various schemes. Consumer awareness of schemes ensures that consumers know where to lodge complaints. The Taskforce believes that access to self-regulatory schemes is crucial.

The Commonwealth Consumer Affairs Advisory Council commented that codes are invisible to certain consumers. In particular:

- urban elderly women and men;
- > rural and remote families:
- working parents who have no time;
- **people** isolated in their own homes because of poverty or ill health;
- people with low literacy and/or verbal skills;
- people of working age dependent on government support;
- > young people who have never had a full time job, permanent job; and
- non-English speaking people.¹⁶²

It argued that an effective code would place the onus on the provider to make consumers aware that a code exists and an easy access point to provide all necessary information about the code. The onus to advise consumers should arise during the course of a relevant transaction. 163

The Australian Consumers' Association stated that empowering consumers to interact with industry in a self-regulatory environment is crucial. Consumers need easy access to dispute resolution schemes and education about the requirements of industry codes, regardless of their literacy or financial background. It argued that

¹⁶¹ Ibid

¹⁶² Submission number 14, p. 4.

¹⁶³ Ibid.

one of the obvious areas that require improvement is in the provision of written evidence. For people of a low literacy level or non-English speaking background this can be difficult. The Association recommended the provision of oral evidence and the use of new technologies to facilitate greater consumer education and access to dispute schemes. ¹⁶⁴

The Law Council of Australia argued that the suggestion that self-regulation is for the benefit of consumers, as it keeps the prices of goods and services lower than would otherwise be possible, is only part of the equation. It submitted that the consumer must also have redress from an impartial umpire. Disadvantaged low income, poorly educated and foreign born consumers, who comprise a significant number of consumers, are not in a position to analyse the information or access legal and other representative systems to assess the information for them. ¹⁶⁵

Similarly, during the consultations held by the Taskforce, other organisations raised the importance of consumer awareness of schemes and suggested the schemes should be user-friendly. However, the Taskforce recognises that equality in access is not just isolated to self-regulatory schemes — similar issues of access apply to the court system for example.

The Telecommunications Industry Ombudsman scheme conducts regular two-year public awareness surveys, as well as biannual complaint satisfaction surveys. It commented that their scheme has also paid attention to consumers from non-English speaking backgrounds. There is also no cost to consumers accessing the scheme. The scheme has widely advertised 1800 Freecall and Freefax number and also accepts complaints on-line through its Website, as well as by mail and in person. ¹⁶⁶

The Federation of Australian Commercial Television Stations commented that consumers with a disability may complain via the telephone or on an audio cassette in the first instance. It stated that there has been a considerable degree of public awareness of the Code over the six years of its operation. This has been facilitated by on-air publication by stations. This has now been formalised in the revised Code. It commented that all stations must broadcast 360 on-air spots per year across all viewing zones about the code and the complaint process. In addition, they have established a national phone hotline (a 1800 number) which provides information about the code, how to make complaints and how to contact local stations. ¹⁶⁷

The Australian Subscription Television and Radio Association has also pursued avenues of promoting consumer awareness of its Code, through its website and on request from the 1300 call centre numbers of FOXTEL, OPTUS Television and AUSTAR. ¹⁶⁸

¹⁶⁴ Submission number 16, p. 2.

¹⁶⁵ Submission number 19, p. 8.

¹⁶⁶ Submission number 21, p. 10.

¹⁶⁷ Submission number 34, p. 8.

¹⁶⁸ Submission of 19 July 2000.

In regional areas, the Taskforce heard that consumer awareness of dispute resolution schemes is an issue. Although access is not so problematic due to modern technology, problems arise in knowing how to lodge complaints with schemes, having the time to write letters of complaint, the lack of personal contact and the time associated with having complaints investigated. During the Bunbury consultations, there was general agreement that codes of practice for example need to be advertised to ensure that consumers are aware of their existence. Similarly, the Mid West Development Commission in Geraldton, Western Australian, were concerned about how self-regulation is being carried out by some industry sectors in the less populated rural areas.

Throughout this report, the Taskforce has stressed that consumer awareness is an important element of good practice in self-regulation. Consumer awareness of schemes ensures that consumers know where to lodge a complaint. The Taskforce recognises that schemes can encourage access by utilising technology such as web-sites, by making any complaints cost free to the consumer, through writing sample letters of complaint, through taking oral complaints where possible, and through transferring complainants between schemes where possible.

Industry awareness

As well as consumer awareness, industry members need to be aware of what they are supposed to be doing in terms of compliance. Raising industry awareness of schemes ensures that industry participants understand their obligations to consumers.

The ACCC commented that in many cases a code fails to operate effectively, not because its principles and procedures are inadequate, but because employees or industry members are either unaware of the code or fail to follow it in day-to-day dealings. For example, to raise industry awareness, the Australian Subscription Television and Radio Association commented that they undertake regular codes presentations to relevant staff of platforms, channels and call centres. It argued that it is of particular importance that, with the level of staff turnover in the call centres, that these presentations are conducted regularly. ¹⁷³

Similarly, the Australian Direct Marketing Association commented that it has undertaken a major industry education initiative to assist members in ensuring that their organisations comply with their code. It commented that this has taken the form

68

¹⁶⁹ Taskforce consultations in Mildura (14 June 2000), Rockhampton (27 June 2000) and Tamworth (28 June 2000).

¹⁷⁰ Taskforce consultations in Bunbury (20 July 2000).

¹⁷¹ Mid West Development Commission, 30 June 2000.

¹⁷² Submission number 42, p. 27.

¹⁷³ Submission number 39, p. 5.

of full-day Code Compliance Workshops across Australia that have, at the time of their submission, attracted over 200 of their members. ¹⁷⁴

Administration

Administration

Actually administering a self-regulatory scheme can be a task in itself. A good administrative body can identify specific problems in an industry, collect data, monitor the scheme and enhance its credibility.

The Department of Industry, Science and Resources commented that an autonomous body with diverse stakeholder representation should be tasked with monitoring, maintaining and enforcing the regime. It is particularly important that such a body have responsibility for establishing and undertaking a continuous review program to ensure the regime is a 'living organism'.¹⁷⁵

The Commonwealth Consumer Affairs Advisory Council argued that codes must have a 'home' and an administration. It noted that it has been said that one of the key reasons for the success of the General Insurance Code is that not only does it have a proper administration but also a Board of Directors comprised of insurance company Chief Executive Officers together with other major stakeholders. ¹⁷⁶

The ACCC commented that industry-based code schemes aimed at delivering fair trading outcomes need to contain appropriate consumer/user representation on the administration committee. In some instances, representation by the appropriate regulatory authority on the code administration body can serve as a means of the regulatory body putting forward a public interest view. It argued that such representation provides transparency to the scheme by providing a 'public window' on its operations that ensures the industry group will be acting in the broader public interest.¹⁷⁷

The Australian Direct Marketing Association stated that the creation of its independent Code Authority brings greater transparency and accountability to complaints handling.¹⁷⁸ Similarly, Financial Industry Complaints Service commented that the scheme should be an entirely separate entity from the industry so there is no perception of bias.¹⁷⁹

The Taskforce considers that good administration of a scheme underpins good practice. It can identify issues, collect data, monitor the scheme, ensure compliance

¹⁷⁴ Submission number 36, p. 4.

¹⁷⁵ Submission number 31, p. 22.

¹⁷⁶ Submission number 14, p. 3.

¹⁷⁷ Submission number 42, p. 26.

¹⁷⁸ Submission number 36, p. 11.

¹⁷⁹ Submission number 15, p. 4.

costs are at an effective minimum level and enhance credibility and accountability. The Taskforce considers the type of administration (i.e. whether a scheme can be administered by individual firms, industry associations, or some form of independent body) will depend on the nature of the specific problem and the nature of the industry. ¹⁸⁰

The Taskforce also considers that industry self-regulatory bodies should endorse their code compliant members on the basis of continued compliance.

Data collection

Data collection by an industry scheme is important as a valuable source of market information about the origins and causes of complaints. It also enables identification of systemic problems which need to be addressed by industry members. This, in turn, can improve market outcomes for consumers.

ASIC stated that a vital role that self-regulation can play in the broader regulatory environment is to identify emerging industry risk areas. In doing so self-regulation can serve to alert industry to potential problems before they actually materialise in market misconduct. ¹⁸¹

The Consumer Law Centre of Victoria, Consumer Credit Legal Service (Vic) and the Financial and Consumer Rights Council (Vic) comment that systemic issues arise out of common practices by industry and/or experiences of multiple consumers. The identification of such issues provides valuable information to industry as to the effects of its processes on consumers as well as allowing solutions to be found. ¹⁸²

Generally, systemic problems can be identified two ways. First, an individual complaint may be such as to identify a system or process problem which has the potential to affect many consumers. Second, the accumulation of complaints and further statistical analysis will either identify or suggest the existence of a systemic problem.

For example, the Telecommunications Industry Ombudsman commented that its primary role in this context is notifying individual members, referring the issue to relevant regulators (e.g. the Australian Communications Authority or ACCC), or highlighting the issues in public forums and in the media.¹⁸³

¹⁸⁰ For further information on administering a scheme see the Ministerial Council of Consumer Affairs Guideline 1996, Fair Trading Codes of Conduct, Why have them, how to prepare them available from the Consumer Affairs Division of Treasury and State and Territory Fair Trading Offices, and also see the Commonwealth's Codes of Conduct Policy Framework released by the then Minister for Customs and Consumer Affairs in March 1998; this document is available on the Internet at http://www.treasury.gov.au

¹⁸¹ Submission number 37, p. 18.

¹⁸² Submission number 29, p. 13.

¹⁸³ Submission number 21, p. 12.

As discussed by the Consumer Law Centre of Victoria, Consumer Credit Legal Service (Vic) and the Financial and Consumer Rights Council (Vic), it is inherently difficult for industries to step back from their business to identify systemic problems, particularly where the process or practice is accepted within the industry. Further, the cost of implementing strategies to address such issues may be perceived to be contrary to a profit-making imperative. ¹⁸⁴

The Taskforce considers that data collection is a valuable tool in identifying systemic issues which need to be addressed by industry members which, in turn, can improve market outcomes for consumers. Also, depending on the nature of the specific problem being addressed, dispute resolution schemes can offer industry an effective means to identify systemic issues.

Transparency

Transparency is another essential feature of schemes. As consumers cannot guard against specific industry problems that they do not know exist, transparency in schemes is an important mechanism to ensure credibility and accountability.

A number of organisations consulted discussed the importance of transparency in enhancing the credibility of schemes. For example, the Australian Toy Association suggested that any self-regulatory arrangements should be transparent and its operation open to scrutiny, and subject to a process of review after a set period (approximately 3-5 years). 185

The Commonwealth Consumer Affairs Advisory Council commented that one thing that seems to be consistent amongst all codes is that consumers are not informed of who's complying and who's not. This information is clearly potentially important to consumers when choosing which company they may do business with. 186

The Taskforce considers that schemes ought to be transparent and open to scrutiny to improve market outcomes for consumers.¹⁸⁷

Dispute procedures and sanctions

Industry adherence to schemes

For self-regulatory schemes to achieve their objectives, compliance by industry members is a key feature. Compliance by industry ensures that specific industry

¹⁸⁴ Submission number 29, p. 13.

¹⁸⁵ Submission number 23, p. 3.

¹⁸⁶ Submission number 14, p. 3.

¹⁸⁷ For further information on transparency see *Benchmarks for Industry-based Customer Dispute Resolution Schemes 1997*, released by the then Minister for Customs and Consumer Affairs, Senator the Hon Chris Ellison. This publication can be accessed through the Treasury web-site: http://www.treasury.gov.au/publications.

problems are being addressed. As noted previously, some self-regulatory schemes are voluntary in their nature whereas others are mandatory. The degree of compliance can depend on a number of factors, such as whether firms obtain marketing benefits from it, whether they are in a competitive market, or whether they risk sanctions for non-compliance.

In house compliance

One means of compliance with self-regulatory schemes is through internal mechanisms.

In relation to codes, the ACCC commented that the code's administration body needs to ensure that each participant has some form of in house compliance system to ensure compliance with the code. It can also assist compliance at this level with advice and training. It noted that, in Australia, code compliance manuals are being developed for code schemes. These manuals are based on the recently released standard on compliance programs (AS3806). ¹⁸⁸

Similarly, ASIC commented that adequate training of staff is good practice, in other words compliance officers in individual firms, staff in scheme administrators and, where appropriate, staff in external complaints resolution schemes. ¹⁸⁹

NRMA also considered that self-regulation is enhanced by establishing a specialised unit within the company to deal with compliance. This helps to ensure that all customer concerns and issues are given a high priority, are managed by the appropriate business unit and are addressed in a consistent and timely manner. ¹⁹⁰

Similarly, the AAMI customer charter sets out some clearly defined rights for its customers.

A competitive market can be another driver of compliance. For example, the Australian Food and Grocery Council commented that industry self-regulation is most effective addressing issues where individual companies can play an active role in promoting voluntary codes, monitoring compliance in the market place, and contributing to code enforcement through a complaint resolution process.¹⁹¹

Similarly, ADMA considered that its direct marketing code brings peer pressure to bear on members who breach the code, a powerful coercive force in the marketplace. As such, member compliance is driven by enlightened self-interest, not fear of state intervention in business affairs. ¹⁹²

189 Submission number 37, p. 15.

¹⁸⁸ Submission number 42, p. 26.

¹⁹⁰ Submission number 7, p. 3.

¹⁹¹ Submission number 30, p. 13.

¹⁹² Submission number 36, p. 11.

The Taskforce considers that compliance with standards across the industry is necessary for good practice in self-regulation. It ensures that specific industry problems are being addressed and the benefits from standards of practice in schemes are flowing to consumers.

Complaint handling

Depending on the nature of the specific problem, self-regulation should incorporate complaint handling and dispute resolution mechanisms to provide appropriate redress to customers where the standard of conduct was breached. Redress encourages industry members to react promptly and fairly to complaints by having internal complaint resolution mechanisms and, where appropriate, subscribing to some form of fair and independent dispute resolution scheme.

These processes are essential to ensure that dissatisfied consumers have access to cost-effective mechanisms for resolving their complaints about the conduct of members of the code. The formal legal system involving court litigation is not designed to provide quick and cheap complaints resolution.

For example, the Institution of Engineers commented that the most secure protection for the community lies in the fundamental requirement of its code that registered persons must practice within the limits of their personal and professional competence, and in the assurance that they will be subject to effective disciplinary action if they fail to observe that constraint. As such, the Institution has procedures for dealing with complaints about members including investigation of the complaint and applying sanctions where appropriate. ¹⁹³

ASIC stated that accessible and effective complaint resolution mechanisms serve to buttress consumer confidence. They can also provide benefits to business, for example, by enabling industry to identify and address systemic consumer problems, thereby maintaining consumer confidence and avoiding the need for government intervention. ¹⁹⁴

The Taskforce considers that businesses should establish fair and effective internal procedures to address and respond to consumer complaints and difficulties:

- (a) within a reasonable time;
- (b) in a reasonable manner:
- (c) free of charge to the customer; and
- (d) without prejudicing the rights of the consumer to seek legal redress.

¹⁹³ Submission number 26, p. 15.

¹⁹⁴ Submission number 37, p. 16.

If a consumer is unsatisfied with the resolution process provided by the internal complaint handling mechanism, then it is good practice for the business to provide the consumer with information regarding any external dispute resolution body to which it subscribes or any relevant government body, such as a Fair Trading Agency.

NSW Legal Aid argued that any self-regulatory scheme must allow for the establishment of internal dispute resolution procedures and for the monitoring and improvement of such processes.¹⁹⁵

The NRMA stated that an element of good practice has been to standardise the process of dealing with customer concerns. For example, NRMA adopts a three step process of, first, referring the issue to the relevant business area, second, allowing the Customer Relations area to mediate and, third, taking the matter to the independent dispute resolution body. ¹⁹⁶

At the more interventionist end of the self-regulatory spectrum where businesses may be dealing with a large amount of complaints and/or dealing with complaints of a more serious nature, an external dispute resolution scheme may be appropriate. An independent body capable of adjudicating and exercising sanctions can further strengthen an external dispute resolution scheme.

The Taskforce considers that a business should provide clear and accessible information to consumers on any independent customer dispute resolution mechanism to which the business subscribes.

Such independent customer dispute resolution mechanisms should be:197

- (a) accessible;
- (b) independent;
- (c) fair;
- (d) accountable:
- (e) efficient; and
- (f) effective.

¹⁹⁵ Submission number 24, p. 4.

¹⁹⁶ Submission number 7, pp. 3, 4.

¹⁹⁷ For more information regarding external dispute resolution schemes, businesses are encouraged to consult the Commonwealth's *Benchmarks for Industry-based Customer Dispute Resolution Schemes*, 1997. This publication can be accessed through the Treasury web-site: http://www.treasury.gov.au/publications. There are also a number of resources available in relation to internal complaint handling, including AS4269, the Australian Standard on complaint handling. This standards is produced by the private organisation, Standards Australia.

NRMA Limited submitted that dispute resolution schemes should also be independently audited and/or reviewed. 198 The Taskforce takes up this issue below.

As discussed in chapter 8, a number of industries have an external dispute resolution scheme.

Consumer representatives on dispute resolution schemes can also ensure credibility and independence. A number of organisations including industry, consumer groups and government all commented on the usefulness of consumer representatives on the ACCC commented that, example. where appropriate, industry-based code schemes aimed at delivering fair trading outcomes need to contain appropriate consumer/user representation in complaints handling. It argued that such representation provides transparency to the scheme by providing a 'public window' on its operations that ensures the industry group will be acting in the broader public interest. 199

Similarly, the Law Council of Australia argued that as effective enforcement is the only way to protect consumers' rights, a minimum condition for successful self-regulation is the provision of industry funded independent consumer representatives, so that the various uneven elements of the consumers/producer relationship can be remedied.²⁰⁰ The Taskforce considers that self-regulatory schemes should aim to provide appropriate redress to consumers where the standard of conduct has been breached. Consumer redress is essential to ensure that dissatisfied consumers have access to cost-effective mechanisms for resolving their complaints about the conduct of members of schemes. The appropriate redress mechanism will depend on the nature of the specific problem trying to be addressed.

Sanctions for non-compliance

It will generally be desirable for the self-regulatory scheme to provide for a range of enforcement options, depending on the nature of the breach. For example, immediate expulsion may not be a suitable sanction for a minor breach of the scheme. However, effective sanctions can raise the level of credibility and consumer confidence in schemes. A comment often heard during the Taskforce consultations was that schemes need to have 'teeth'.

Industry associations use a range of different sanctions. For example, the Institution of Engineers commented that all of their members are bound by their Code of Ethics. They then have procedures for dealing with complaints about members and are able to apply a range of sanctions including expulsion and suspension of membership. 201 Similarly, the Australian Pharmaceutical Manufacturers Association Code of Practice

¹⁹⁸ Submission of 14 July 2000.

¹⁹⁹ Submission number 42, p. 26.

²⁰⁰ Submission number 19, p. 11.

²⁰¹ Submission number 26, p. 15.

contains a hierarchy of sanctions ranging from corrective advertising, fines of up to \$30 000, or expulsion.²⁰²

The Code of Practice adopted by the Fruit Juice Industry is supervised by an Industry Compliance Committee. Ultimate sanctions are law enforcement by the appropriate government regulatory bodies should the self-regulatory scheme be ignored or flouted by participants.²⁰³

The Department of Industry, Science and Resources argued that with voluntary participation, effective sanctions and incentives can be applied, with low scope for the benefits being shared with non-participants. It submitted that voluntary participation — backed by strong incentives to participate — appears to provide a stronger framework and higher degree of success, independent of the size of the industry association. ²⁰⁴

On the other hand, the Motor Trades Association of Australia commented that sanctions, if they exist, are usually not effective because of the voluntary nature of the regulatory scheme. For example, if a sanction is to be imposed for non-compliance, the offending party can simply 'opt out' of the regulatory scheme and continue with the behaviour. ²⁰⁵

Similarly, a number of other organisations during Taskforce consultations echoed these comments. For example, the Mallee Tenancy and Consumer Advice Service expressed concern that ombudsman schemes in particular could be 'toothless tigers' if schemes involved a drawn out process and decisions were unenforceable. It suggested that schemes need to have 'teeth' in order to be effective.²⁰⁶

ASIC argued that the consequences of inadequate enforcement in an industry such as the financial services industry can be serious. The industry is heavily dependent on consumer confidence — if consumers suffer financial losses due, for example to intermediary misconduct in contravention of the requirements of a self-regulatory scheme, the reputation of the Australian financial markets may suffer. Accordingly, ASIC considered that it is essential that self-regulatory schemes be underpinned by effective sanctions. ²⁰⁷

The Australian Consumers' Association recommended a 'toolkit' of actions underpinned with government involvement. These include:

²⁰² Submission number 10, p. 3.

²⁰³ Australian Business Limited, 14 June 2000.

²⁰⁴ Submission number 31, p. 22.

²⁰⁵ Submission number 1, p. 1.

²⁰⁶ Taskforce consultation in Mildura, 14 June 2000.

²⁰⁷ Submission number 37, p. 15.

- rewriting of misleading, incorrect or false consumer information;
- corrective advertising; and
- ► retraining staff. ²⁰⁸

The Department of Industry, Science and Resources also noted that the use of penalties and/or sanctions within self-regulatory regimes require careful consideration as they can lead to retaliatory action by participants. Similarly, the Motor Trades Association of Australia noted that if sanctions are pursued by the self-regulatory scheme administrators, perhaps a peer group or industry association, then there is a question about protecting the administrators from legal action by the party affected by the sanction. Since Industry association and Industry affected by the sanction.

The Taskforce considers that there should be a range of sanctions that can be used by industry in order to achieve compliance depending on the nature of the problem and the consequences of non-compliance. Sanctions can raise the level of consumer confidence in schemes. The Taskforce considers that the severity of the sanction should also depend on the seriousness of the breach.

Competitive implications and the authorisation process

Where industry has the commitment to collectively sanction breaches of a self-regulatory scheme there is the possibility that such action may amount to anti-competitive behaviour. In most cases such action may not amount to anti-competitive behaviour or the benefit to the public may outweigh such behaviour. However, to avoid any threat of legal action for breach of the competition provisions of the *Trade Practices Act 1974*, a procedure exists whereby industry can have the arrangement authorised.

Prior authorisation for such a collective arrangement can be sought from the ACCC which assesses whether there is sufficient public benefit flowing from the arrangement to outweigh any anti-competitive effects. An authorisation from the ACCC gives parties involved in the anti-competitive arrangement immunity from court action taken under the competition provisions of the *Trade Practices Act 1974*.

The authorisation process can be time consuming and expensive, but it does provide industry with protection against legal action. Hence, if industry believes that there is a significant risk that their scheme has anti-competitive elements then authorisation

²⁰⁸ Submission number 16, p. 2.

²⁰⁹ Submission number 31, p. 22.

²¹⁰ Submission number 1, p. 1.

may be a prudent course of action to pursue.²¹¹ It is generally up to industry to manage this risk and ensure any self-regulatory scheme encourages competition.

For example, the Australian Direct Marketing Association stated that punitive actions against members, which may otherwise be judged anti-competitive have been authorised by the ACCC.²¹²

The Taskforce considers that industry needs to manage the risk of any anti-competitive practices in schemes, particularly where sanctions are involved. Although the Taskforce recognises that there needs to be a public benefit justification process, the Taskforce notes that there has been some criticisms of the authorisation process.²¹³

Monitoring and reviewing

Monitoring

As discussed elsewhere, monitoring is an important aspect of self-regulation to ensure that agreed standards are being met. Establishing a self-regulatory scheme is only part of the equation. Industry also needs to be aware that it has a continual responsibility to ensure that self-regulation is addressing its objectives and ethical members are not being disadvantaged.

ASIC commented that the compliance monitoring mechanisms should be tailored to the particular scheme's circumstances. The appropriate compliance monitoring mechanisms will depend on the identified regulatory outcomes and the nature of the particular industry sector. There are various methods for monitoring compliance, including:

- > the internal controls of the individual firm;
- annual reporting on compliance;
- independent monitoring;

²¹¹ See Part VII — *Trade Practices Act 1974.* Aspects of a code which might require the code to be authorised include restrictions on membership, prohibitions on code members dealing with other industry participants, and imposing sanctions for non-compliance where such sanctions effect a party's ability to compete. Most applications attract a fee of \$15000. Details of the ACCC authorisation process are available from the ACCC website at: http://www.accc.gov.au/adjudication/fs-adjudicate.htm.

²¹² Submission number 36, p. 11.

²¹³ Chapter 7 — approaches to promoting and coordinating industry self-regulation, discusses the authorisation process further.

- > an external compliance audit; or
- ► the regulator. 214

The Australian Chamber of Commerce and Industry stated that encouraging managers and employees to take responsibility for their own actions is its preferred approach.²¹⁵

However, some organisations have argued that the government needs to monitor self-regulation more closely. For example, the Royal Aeronautical Society argued that self-regulation must always be accompanied by a rigorous system of dialogue with, and policing by, the government agency responsible for the safety of the public.²¹⁶

The Taskforce considers monitoring is crucial to good practice in self-regulation. Monitoring ensures that the scheme is addressing specific problems within an industry. The Taskforce recognises that the role of government in monitoring will depend on the circumstances. As a general principle, if there is a public policy objective to do so (e.g. health and safety reasons), then the government may choose to be directly involved in the monitoring of schemes.²¹⁷

Accountability

The self-regulatory scheme should publicly report whether its standards are being met. This can improve credibility and consumer confidence in schemes.

The Australian Toy Association commented that any self-regulatory arrangement needs to be accountable in terms of the body administering the scheme.²¹⁸

The ACCC commented that annual reports on the operation of the code should be produced by the code administration committee, allowing for periodic assessment of the scheme's effectiveness.²¹⁹

For example, the Australian Pharmaceutical Manufacturers Association stated that breaches of the code are reported in their annual reports. ²²⁰ Similarly, the Australian Supermarket Institute stated that public reporting occurs annually on the Scanning Code's operation. ²²¹

²¹⁴ Submission number 37, pp. 15, 16.

²¹⁵ Submission number 27, p. 5.

²¹⁶ Submission number 40, p. 4.

²¹⁷ Chapter 7 — approaches to promoting and coordinating industry self-regulation, discusses the role of government in this context further.

²¹⁸ Submission number 23, p. 3.

²¹⁹ Submission number 42, p. 27.

²²⁰ Submission number 10, p. 4.

²²¹ Submission number 11, p. 2.

The Taskforce considers that accountability is an element of good practice in self-regulation. The Taskforce recognises that annual reports are a useful tool to allow a periodic assessment of the scheme's effectiveness.

Review

The Taskforce considers that periodic reviews of standards should also be undertaken to ensure that they are being met and are relevant and up-to-date. This ensures that the scheme is still appropriate to the specific problems it is seeking to address and allows for other stakeholders, such as consumers and government, to be involved. Preferably, reviews should be periodic, independent and be made publicly available.

The Taskforce also acknowledges that self-regulatory schemes need the time and opportunity to evolve and rectify problems as they arise.

ASIC stated that self-regulatory schemes should be regularly reviewed for efficiency and effectiveness. Such reviews are essential to deal with market changes due to innovation and other forces which can rapidly lead to out-of-date regulation. It argued that reviews of individual schemes are usually best conducted by an independent consultant in consultation with the stakeholders involved in the development of the scheme (i.e. industry members, consumer organisations and the regulator). 222

ASIC suggested that scheme reviews should be undertaken at least once every three years. This should encompass the content of the code and the operation of the external complaint resolution scheme. It argued that it is also desirable that reviews are publicly conducted wherever possible.²²³

Insurance Enquiries and Complaints Limited commented that their code and scheme are reviewed periodically so that a flexible approach towards a changing market place is maintained. In particular, it commented that the scheme has evolved over time with the jurisdiction being progressively widened. The company issues an annual review which reports on all aspects of the scheme, the code and now the privacy principles.²²⁴

Cost-effective practice in self-regulation

Throughout this report, reference has been made to the benefits of self-regulation. However, self-regulation does come at a cost to both the industry and consumer. The costs involved in administering an inefficient self-regulatory scheme may be translated into higher prices for consumers resulting in a poor market outcome for

²²² Submission number 37, p. 19.

²²³ Ibid.

²²⁴ Submission number 8, p. 4.

both business and consumers. Compliance costs can also be high for business, which in turn, can be passed onto consumers. However, the Taskforce is not touting that schemes should be cost-effective at the risk of sacrificing consumer rights for example. It is necessary to ensure that the scheme is the effective minimum solution for the specific problem in hand.

Benefits of self-regulation over explicit regulation and courts system

During the inquiry, the Taskforce received a lot of anecdotal evidence to suggest that self-regulation is more cost effective than government regulation and the court system. For example, *A Guide to Regulation* (1998) commented that there are cost advantages from tailor-made solutions and less formal mechanisms such as access to quick complaints handling and redress mechanisms. However, the Taskforce recognises that self-regulation is not necessarily cheaper than government regulation. Furthermore, it is ultimately the consumer that bears the cost of any form of regulation in most cases.

NRMA stated that compared with more direct government regulation, in many cases industry self-regulation results in lower regulatory costs. For example, for general insurance claims, handling costs can be reduced by producing policy documentation in a standardised format. It also argued that self-regulation will normally have faster and simpler dispute resolution procedures, again resulting in lower regulatory costs. In competitive markets, these lower regulatory costs are generally passed onto consumers. ²²⁶

The Telecommunications Industry Ombudsman also commented that its dispute scheme has provided a free and timely forum for the redress of consumer complaints in contrast to costly and time consuming action in courts or consumer tribunals.²²⁷

In addition, the Office of Small Business stated that although small business often have legal recourse in disputes, their access to justice can be constrained by the cost of going to court, the long time and delays before their case is heard, the disparity in the quality of representation and their need to preserve business relationships. In many cases, neither party achieves a satisfactory result from a Court judgement. ²²⁸

In most cases, the Office of Small Business argued that self-regulatory approaches can offer small business a low-cost, quick and flexible system for resolving disputes. It submitted that this provides a viable alternative to litigation, typically achieving a success rate of around 80 per cent, without costly and time-consuming legal action. For example, the Office of Small Business quoted that some studies show that using

²²⁵ Chapter 5 indicates that if there are cost advantages and/or increased flexibility in using self-regulation, then a scheme has a better chance of success.

²²⁶ Submission number 7, p. 2.

²²⁷ Submission number 21, p. 9.

²²⁸ Submission number 38, p. 2.

dispute resolution schemes can cost as little as five percent of the cost of going to court.²²⁹

Cost of schemes

While a lot of the benefits and elements of good practice in self-regulation have been flagged, the schemes come at a cost. And depending on the specific problem being addressed, this cost can be significant.

As stated by the Institution of Engineers Australia any form of regulation, whether self-regulation or otherwise, requires an allocation of resources and will necessarily involve compliance costs that may or may not become onerous. It argued that effective self-regulation requires not only standards or codes of practice, but also effective mechanisms for dealing with complaints of non-compliance with those codes. Self-regulation requires extensive community and business education, and requires a commitment from all industry players to work effectively. ²³⁰

Similarly, the Federation of Australian Radio Broadcasters noted that while some may see the overlay of an implementation and monitoring system as a level of bureaucracy, it is clear that there is an expectation that industries and industry bodies take responsibility for the effective operation of their systems of self or co-regulation. If there is a cost in either time or money then so be it — 'self-regulation is not meant to be cost free regulation', according to one observer.²³¹

The cost of self-regulatory schemes can vary greatly. At one end of the spectrum, industry initiatives that improve the amount of information available to consumers to make informed choices can be relatively inexpensive. For example, a guideline may only involve printing and staff time costs.

Similarly, more sophisticated schemes can be fairly inexpensive depending on what is trying to be achieved. For example, the Australian Supermarket Institute stated that the cost of the Scanning Code administration includes materials, printing and distribution, and staff time handling issues. It is estimated this cost would not exceed \$40 000 per annum. ²³²

Whereas, at the more interventionist end of the self-regulatory spectrum, schemes can cost more. For example, the Federation of Australian Commercial Television Stations estimated that the cost of its scheme to the industry is at least \$3 million annually. It commented that the industry's Code of Practice places considerable responsibility on individual stations, and is relatively resource-intensive and costly

²²⁹ Ibid.

²³⁰ Submission number 26, p. 23.

²³¹ Submission number 35, p. 10.

²³² Submission number 11, p. 4.

to operate. However, it believed the industry supports the process because it is more efficient, simple and direct than any regulatory alternative. ²³³

Further, the bigger dispute resolution schemes that resolve customer disputes are expensive to run. For example, the Australian Banking Industry Ombudsman, Telecommunications Industry Ombudsman and General Insurance Enquiries and Complaints schemes all cost in excess of \$3 million per year.²³⁴

Compliance costs

However, the administrative costs of self-regulatory schemes is only part of the story. Compliance costs are also associated with self-regulatory schemes. It is necessary to ensure that self-regulation does not itself become a burden to industry with onerous compliance costs, particularly for small businesses.

Again, there is a spectrum of compliance costs for business. At the lower end of the spectrum, schemes that raise the level of information may simply involve preparing a disclosure document for example. Similarly, in complying with standards, there are low compliance costs involved in following the standard on work safety boots. Whereas, the quality management system standard (ISO9000) produced by Standards Australia is more expensive to implement depending on the size of the company. ²³⁵

At the more interventionist end of the spectrum, codes and dispute resolution schemes can be expensive for businesses to comply with. The Financial Industry Complaints Service stated that various levels of funding are required depending on the number of disputes and the quality and expertise of staff. Generally, the costs are allocated to members by a capacity to pay , in other words the wealthier members pay the most. 236

For example, Insurance Enquiries and Complaints Limited stated that its scheme is funded by a combination of a levy upon insurers and fee per case charges to insurers. Approximately 60 per cent of its budget is met by a levy upon the personal lines premium income of member companies. Companies either pay the minimum levy of \$1 600 or a levy based on their proportion of personal lines premium income. The other 40 per cent of the budget is met by a fee per case payment. ²³⁷

In comparison, the Telecommunications Industry Ombudsman scheme commented that its funding mechanism is based on two principles. First, that each member pays for the scheme's complaint handling services based on the number, and relative percentage of total complaints, raised by the scheme against that member. Second,

²³³ Submission number 34, p. 9.

²³⁴ Taskforce consultations with industry dispute schemes, November 1999.

²³⁵ Taskforce consultation with Standards Australia, Sydney, 29 November 1999.

²³⁶ Submission number 15, p. 4.

²³⁷ Submission number 8, p. 4.

the complaint handling fees are structured in a way as to provide a financial incentive for members to resolve complaints in a timely manner. The handling fees are \$15 for an initial enquiry, up to \$1 130 for dispute resolution. As a corollary, a member has no financial obligation to the Telecommunications Industry Ombudsman if no complaints are made against that member. There are no membership or joining fees associated with the scheme.

The Taskforce recognises that as businesses become more aware and familiar with a self-regulatory scheme then compliance costs can be reduced. Indeed, often during the self-regulatory scheme's first year of operation compliance costs are particularly high as businesses are training staff etc, but in the following years costs are usually reduced. However, compliance costs can also increase later down the track.

For example, the Service Providers Industry Association argued that the self-regulatory participation burden is likely to increase, rather than decrease in the telecommunications. It considered the increasing complexity of inter-working as the wholesale market and new applications develop will drive this trend, leading to calls for more and more codes and standards, as well as the need to revisit and overhaul existing codes. The accelerating take-up of e-commerce and e-business will also lead to more demands for codes and standards to meet consumer expectations and industry inter-working requirements. ²⁴⁰

The Service Providers Industry Association argued that as more regulatory and industry bodies engage in the process of writing standards, codes and guidelines, compliance becomes an increasing burden on organisations. It commented that not only is it difficult or impossible for most organisations to participate in the creation of these instruments, it becomes very difficult to be fully aware of the range of regulatory instruments that apply to an individual organisation and its business operations. When this situation is reached, it introduces major challenges for staff trainers and compliance managers.²⁴¹

Cost to smaller industry associations and businesses

A self-regulatory scheme stands a greater chance of success if it is backed by a large and well structured industry association. Primarily, this assists in regards to the costs associated with establishing and maintaining regimes, ensuring broad participation, and issues of enforcement, including sanctions.

Where no large industry association exists, costs would appear to be a prohibiting factor in developing and administering a regime. As discussed by the Department of Industry, Science and Resources, in such instances, sharing of the costs between participants and beneficiaries should be considered on a cost recovery basis. It

84

²³⁸ Telecommunications Industry Ombudsman, Annual Report 1998-99, p.10.

²³⁹ Submission number 21, p. 10.

²⁴⁰ Submission number 25, p. 4.

²⁴¹ Ibid.

argued that such cost recovery principles would be regime specific but should encompass a clear process in determining who should pay for developing and administering the regime and how charges should be structured — preferably based on tangible outcomes. ²⁴²

Similarly, the Office of Small Business argued that it is generally accepted that small business is less able than big business to cope with the costs of participating in a scheme such as a code of conduct, particularly when such a scheme is funded by industry levies. It argued that while big business is usually able to absorb such costs, small businesses in many cases have no option but to pass on these costs to the consumer in the form of higher prices for goods and services. This can place small business at a competitive disadvantage to their larger counterparts.²⁴³

The Office of Small Business also argued that the methodology for the administration of fees should be clearly established before any self-regulatory scheme is initiated. It suggested that the methodology should be as transparent as possible once in place. Small business is not likely to feel any ownership of the process of self-regulation if the funding process is not accessible to them. Transparency also ensures that charges for administering a scheme have a direct relationship to actual administration costs, and that charges are regularly reviewed to maintain this relationship. ²⁴⁴

The Taskforce considers that any funding arrangement for self-regulation should be transparent and designed so as not to put businesses at a competitive disadvantage.

Minimum standards

As discussed previously, the notion of the effective minimum solution should apply to self-regulatory schemes.

For example, the National Insurance Brokers Association does not support the imposition of best practice standards as they are not always appropriate. In many cases the standards set are in fact best practice. However, this is only where it is appropriate. It argued that where this is not the case, minimum standards apply that provide proper protection for consumers. ²⁴⁵

The National Insurance Brokers Association also noted that depending on the resources or corporate culture of the relevant entity, they choose whether to exceed these minimum standards. Certain entities may not be able to meet the compliance costs of best practice even though minimum standards will provide consumers with appropriate service and protection.²⁴⁶

85

²⁴² Submission number 31, p. 22.

²⁴³ Submission number 38, p. 3.

²⁴⁴ Submission number 38, p. 4.

²⁴⁵ Submission number 3, p. 6.

²⁴⁶ Ibid.

Similarly, Clayton Utz argued that codes of conduct should be designed and drafted with an appropriate minimum standard in mind. Such a standard must be realistically set to ensure that it can realistically be complied with by industry participants. ²⁴⁷

The Taskforce recognises that standards in self-regulatory schemes should be the effective minimum solution to the specific problem.

Summary

The appropriate form of self-regulation will depend on what is trying to be achieved which will vary depending on the industry. Good practice in self-regulation involves improving market outcomes for consumers at the lowest cost to businesses.

²⁴⁷ Submission number 43, p. 4.

Chapter 7

Approaches to promoting and coordinating industry self-regulation

The Taskforce is to inquire and report into approaches to promoting and coordinating industry self-regulation, including the appropriate role of government and the development of industry codes as well as other approaches to self-regulation.

The key players in the promotion of industry self-regulation have always been industry, government and consumer advocates. The purpose of this chapter is to examine the respective roles of each group in the establishment of particular schemes.

The roles of industry, government and consumer groups are dynamic, adapting to the changing face of the Australian economy and, in particular, responding to competitive pressures, regulatory reform, new technologies and the increasing globalisation of consumer markets.

Industry has shown a growing enthusiasm for initiating self-regulation to engender consumer confidence in new products and new technologies. Government has increasingly promoted self-regulatory options as part of its broader commitment to regulatory reform. Consumer groups have embraced self-regulatory schemes to address consumer problems that cross jurisdictional boundaries. It is also worth noting the growth of third party certification schemes, particularly in the online environment, hinting at the commercial imperatives driving businesses to subscribe to some form of voluntary industry self-regulation to win consumer trust.

There is a variety of options for designing and promoting self-regulatory schemes and what works for one industry may not work for another. It follows that the 'mix' of industry/government and consumer involvement that works well for one self-regulatory scheme may be inappropriate for another.

This chapter first examines the role that industry has played in promoting and coordinating self-regulatory schemes. This chapter then discusses the role of government as a stakeholder, developer, promoter, monitor and enforcer of schemes, as well as the crucial role that consumer groups have played and will continue to play in the development of industry self-regulation. Finally, the chapter analyses a number of options to better co-ordinate and promote self-regulation including discussion of whether a centralised government agency, an oversight committee, or model codes would be appropriate.

Conclusions

Industry approaches to promoting self-regulation

- 32. Experience has shown that industry will initiate a self-regulatory scheme in response to a clear commercial imperative to win consumer confidence and boost sales.
- 33. Industry may promote self-regulation as an alternative to government regulation where there is perceived to be a serious market failure or important social policy objective.

Role of government in promoting and coordinating self-regulation

- 34. Government involvement in self-regulation is justified when there is a public policy objective that would otherwise call for a regulatory response.
- 35. Government can assist in analysing systemic problems in an industry and in facilitating the design of a self-regulatory response to address those systemic problems.
- 36. Government can assist in integrating schemes into the regulatory framework.
- 37. Government is uniquely placed to promote international cooperation and harmonisation of self-regulatory initiatives.
- 38. The degree of government involvement will depend on the significance of the market failure or social policy objective being addressed and the consequences of self-regulation proving ineffective.

Role of consumer advocates in promoting self-regulation

- 39. Consumer input is important in the development and in maintaining the relevance of self-regulation. Consumer advocates can promote consumer confidence in self-regulatory schemes.
- 40. Consumer participation will be limited by human and financial resource constraints if there is no external financial assistance forthcoming.

Other conclusions

41. Code administration authorities established by industry should take responsibility for the monitoring and review of self-regulation, in consultation with government and consumer groups.

Role of industry

Industry has initiated a host of self-regulatory schemes for a variety of reasons. ²⁴⁸ For commercial reasons, industry may develop a scheme to win consumer confidence and boost sales. Industry may also promote self-regulation as an alternative to government regulation.

The promotion of self-regulatory schemes is beneficial as a marketing tool to differentiate participants from competitors.

Industry self-regulation can also build consumer confidence when introducing new technology to the market. The *Australian Code of Practice for Computerised Checkout Systems in Supermarkets* was introduced to ease the transition away from individual item pricing to barcodes, including the provision of free items when the scanned price is higher than the shelf price.²⁴⁹

The Australian Chamber of Commerce and Industry commented that under self-regulation, industry (often through associations) can assume responsibility for concerns raised by the community and is able to interact directly with stakeholders to resolve the problem. It suggested that industry associations can play a key role in delivering a coordinated approach to issues thus ensuring national consistency. ²⁵⁰

The Australian Supermarket Institute commented that the Code of Practice for the Fruit Juice Industry was developed by the Australian Citrus Industry Council to promote truth in labelling and fair trade of orange juice and other products. It has been designed to ensure that fruit juice is not adulterated and that the public is not otherwise misled about fruit juice products. The Australian Supermarket Institute commented that in taking this action itself, the fruit juice industry feels that it has enhanced both the image of the industry and the marketability of its fruit juice products. ²⁵¹

Similarly, the Australian Society of Certified Practising Accountants and the Institute of Chartered Accountants in Australia submitted that their commitment to self-regulation is vitally important when demonstrating to clients, the business community and governments the high degree of integrity and professionalism which members of the Accounting Bodies possess. To this end, the Accounting Bodies have sought to ensure that these groups can have confidence in the ethical standing and technical competence of our members and, in turn, in the quality of various

²⁴⁸ Chapter 3 outlines reasons for self-regulation and the types of schemes adopted. The consultant's report, Tasman Asia Pacific 2000, *Analysis of market circumstances where industry self-regulation is likely to be most and least effective*, also discusses particular industry's initiating schemes.

²⁴⁹ Submission number 11, p. 2.

²⁵⁰ Submission number 27, p. 5.

²⁵¹ Submission number 11, p. 4.

accounting, auditing and other assurance and advisory services which the Bodies provide. ²⁵²

This pattern is also evident internationally. Indeed, international experience teaches that the actions of one national industry association may act as a catalyst for similar industries in other countries to follow suit. For example, the Canadian Chemical Producers' Association introduced its Responsible Care initiative in 1985 and it has since become a global alliance with over 40 countries participating in the scheme. Each company which subscribes to the Responsible Care ethic is required to abide by six codes of practice which contain more than 150 requirements. Verification checks are undertaken every three years on each member-company of the CCPA to ensure the codes are being followed.²⁵³

Another example of a global voluntary self-regulation is the work conducted by the International Organisation for Standardisation (ISO). The ISO is a non-governmental, worldwide federation of national standards bodies from 130 countries that formulates and implements internationally agreed voluntary standards across a variety of goods and services. ²⁵⁴

The Taskforce considers that industry will promote self-regulation when there are reasons to do so, such as to enhance consumer confidence to boost sales or 'head off' government regulation.

Role of government

Industry self-regulation by its definition, is regulation by industry. However, government involvement in self-regulation can range from general policy guidance to more interventionist models. Government is in a position to identify particular problems or social policy objectives and can assist in designing a self-regulatory response to address them.

Throughout the inquiry process, the role of government in self-regulation has been a contentious issue. Hence, the Taskforce has focused on the role of government in this chapter.

Spectrum of government involvement

As with different types of self-regulation, not surprisingly there is a spectrum of government involvement ranging from little or no involvement to a more interventionist approach.

-

²⁵² Submission number 33, p. 1.

²⁵³ For more information on the CCPA and Responsible Care program please see http://www.ccpa.ca/.

²⁵⁴ For more information please see www.iso.ch.

At one end of the spectrum, industry initiatives that improve the amount of information available to consumers to make informed choices might have little or no government involvement at all. For example, industry guidelines or customer service charters fall into this category.

Moving along the spectrum, standards developed by Standards Australia will often see government as a stakeholder providing comments.

'Light touch' approaches include the work of the Consumer Affairs Division in Treasury and the ACCC in advising schemes on voluntary codes. For example the ACCC has assisted industry with the *Australian Code of Practice for Computerised Checkout Systems in Supermarkets*.

Whereas, the government has incorporated codes into a regulatory framework in other cases. For example, over the last decade, the Commonwealth has established regulatory regimes for broadcasting and telecommunications that incorporate industry codes of conduct. Further, the Commonwealth is presently in the process of developing and implementing regulatory regimes in the financial services sector and privacy standards for personal information handling in the private sector. Both regimes allow for the development of industry codes and complaint handling schemes.²⁵⁵

Similarly, at the international level there is a range of government involvement in self-regulation. Self-regulatory policy has been promoted by the OECD as an alternative to formal regulation and is being embraced by member countries at different rates. Countries such as Australia and Canada are more actively embracing self-regulation, with each government researching and reviewing self-regulatory practice. ²⁵⁶

The Taskforce notes that there is currently a range of government involvement in self-regulation depending on the industry and the particular scheme.

Public policy objectives of government

The degree of government involvement in self-regulation will depend on the public policy objective. ²⁵⁷ Government involvement in self-regulation is justified when there is a public policy objective that would otherwise call for a regulatory response. This can include, for example, to encourage industry to develop codes that will assist in compliance with the law.

²⁵⁵ During the closing stages of the Taskforce inquiry, the Australian Broadcasting Authority released its final report on the *Commercial Radio Inquiry*. The Taskforce notes that the report highlights the importance of raising awareness of schemes.

²⁵⁶ For further information on self-regulatory policy in Canada and the UK see Office of Consumer Affairs, 1998, *Voluntary Codes: A Guide for their Development and Use*, Industry Canada.

²⁵⁷ As discussed in chapter 5, self-regulation is more appropriate where there is no strong public interest concern. Examples of public policy objectives are health and safety issues.

The Institution of Engineers Australia commented that self-regulation is appropriate as applied to the provision of some, but not all, engineering services. The Institution commented that those areas of engineering practice that represent a risk to public health and safety or where there is a significant asymmetry of knowledge between the professional engineer and the consumer require a co-regulatory approach that statutorily limits the provision of certain types of services to competent practitioners.²⁵⁸

On the other hand, the Australian Food and Grocery Council commented that the existence of market failure does not in itself justify government intervention as the market may correct itself over time. Government intervention may not be a remedy to the failure. It commented that a fundamental principle upon which government bases its regulatory and legislative policy is that the onus of proof in determining whether regulation will correct market failure is on the proponents of the regulation. It also argued that intervention must provide the highest net benefit to consumers as individuals, and the community as a whole. ²⁵⁹

The Taskforce also recognises that government will be involved in aspects of particular industries. For example, the scope of service delivery in telecommunications, and the proposed privacy legislation that will impact across all industries.²⁶⁰

As a general principle, the Taskforce recognises that government has a role in self-regulation to ensure that self-regulation targets and achieves the relevant public policy objective.

Government involvement

As discussed elsewhere, the Taskforce considers that self-regulation is a three-way partnership between industry, consumers and government.

From an industry perspective, the Australian Pharmaceutical Manufacturers Association commented that an advantage of their current self-regulatory system is that it is an outward demonstration of successful partnership and trust between government and industry. Similarly, the Australian Supermarket Institute commented that the ACCC has been of great assistance in developing their Code's provisions and ensuring that they have remained relevant and effective. ²⁶²

The Financial Industry Complaints Service commented that the whole of the self-regulatory process was satisfactory. Government was quite willing to allow industry to develop the schemes. The schemes that the Financial Industry

259 Submission number 30, p. 2.

²⁵⁸ Submission number 26, p. 23.

²⁶⁰ Privacy Amendment (Private Sector) Bill 2000.

²⁶¹ Submission number 10, p. 5.

²⁶² Submission number 11, p. 7.

Complaints Service has been involved with have approval from ASIC and consult with Treasury. ²⁶³

Cable & Wireless Optus commented that the involvement of regulators as 'observers' throughout the process of code and standard development has several benefits. First, it has enabled the industry to address regulator concerns at the outset rather than address issues at the end of the process. Secondly, it has resulted in regulators being fully aware of current industry discussions and hence familiar with the context for industry decisions. ²⁶⁴

Cable & Wireless Optus surmised that in planning for self-regulatory schemes, it is considered beneficial for the roles and responsibilities of regulators to be clearly defined at the outset of the process and actively managed to encourage a cooperative approach.²⁶⁵

Participants at a Taskforce meeting held in Rockhampton agreed that government consultations are an important channel for communication between rural Australians and decision-makers. Participants commented that there is a general recognition that governments cannot solve all the problems facing regional Australia, however they considered that the rural community would appreciate receiving direct feedback from government representatives about their concerns.²⁶⁶

The Taskforce considers that government, among other roles, is a stakeholder in self-regulation and can offer assistance in the development and in maintaining the relevance of self-regulation.

Government can promote and assist in the development of self-regulation

The Government has had experience with many industries wishing to introduce self-regulation. It is therefore in a good position to advise and guide industry in different ways of approaching self-regulation.

By publicising the benefits of self-regulation to consumers, the government may also be able to help create an incentive for industry to effectively self-regulate. Maintaining the higher standards required by a self-regulatory scheme may impose costs on businesses which will be reflected in the prices charged to consumers. If consumers are not well informed about the benefits of dealing with a member of the self-regulatory scheme, there will be little incentive for businesses to incur the extra costs required to establish or join the scheme, and those businesses that do so may be at a competitive disadvantage. Once the scheme wins acceptance within the market, there should be less need for the government to maintain its involvement.

-

²⁶³ Submission number 15, p.4.

²⁶⁴ Submission number 6, p. 3.

²⁶⁵ Ibid

²⁶⁶ Taskforce consultation in Rockhampton, 27 June 2000.

The Insurance Council of Australia commented that it is important for government to be supportive of self-regulatory codes and schemes, and to promote and encourage consumer confidence in their use and effectiveness. The Council commented that government could assist industry in the development of self-regulation by promoting a more informed understanding of the role and scope of industry codes among consumer groups. It is particularly important to promote an understanding of the value of self-regulation when it provides a more effective alternative to legislation for both consumers and industry. The Council argued that having formally approved a code, by a sufficiently transparent process, government should become a supporter of its aims and objectives. ²⁶⁷

The Australian Chamber of Commerce and Industry also urged government to address consumer scepticism of industry self-regulation through education and promotion of successful self-regulatory schemes.²⁶⁸

Further, the Australian Direct Marketing Association commented that consumer education should be pursued jointly with industry groups using the resources of government and industry.²⁶⁹

In other Taskforce discussions with stakeholders, it heard that the government can also help assist industries in adjusting to a self-regulatory scheme. In discussions with the Secretariat to the Queensland Red Tape Reduction Taskforce, it commented that in a recently deregulated environment, governments can assist industries by facilitating the development of a self-regulatory scheme.²⁷⁰

Similarly, participants at a Taskforce meeting held in Rockhampton commented that the government could assist industries more when self-regulatory schemes are being introduced.²⁷¹

The Government has also promoted self-regulation through guidelines, directories and the Internet. For example, the Business Entry Point will contain a database of codes across Australia.²⁷² Similarly, the Commonwealth's *Directory of Consumer Dispute Resolution Schemes and Complaint Handling Organisations* sets out consumer

²⁶⁷ Submission number 18, pp. 7-8.

²⁶⁸ Submission number 27, p. 12.

²⁶⁹ Submission number 36, p. 8.

²⁷⁰ Taskforce consultation in Brisbane, 26 June 2000. Established by the Queensland Government, the Queensland Red Tape Reduction Taskforce provides advice to the Deputy Premier on how to reduce the burden of regulation on Queensland business. It recently prepared Guidelines on Alternatives to Prescriptive Regulation which are available from the Queensland Department of State Development.

²⁷¹ Taskforce consultation in Rockhampton, 27 June 2000.

²⁷² The Business Entry Point site can be accessed via http://www.business.gov.au. The Business Entry Point provides businesses with access to resources from approximately 50 Commonwealth Government agencies, 100 State and Territory agencies and 125 industry associations.

dispute resolution schemes, complaint handling organisations and other useful contacts such as fair-trading agencies.²⁷³

As discussed in chapter 6, the government has also released a publication titled *Benchmarks for Industry-based Customer Dispute Resolution Schemes* which is a guide to industry in designing and improving such schemes.²⁷⁴ The Benchmarks were developed to apply primarily to nationally based customer dispute schemes set up under the auspices of an industry. However, the underlying policy principles have also proven useful to smaller and non-industry schemes.

The Taskforce notes that the Minister for Financial Services and Regulation is committed to reviewing the Commonwealth's *Benchmarks for Industry-based Customer Dispute Resolution Schemes* this year.

The Australian Communications Authority has also produced its own guide, titled *Developing telecommunications codes for registration*, which addresses its approach to assessing codes for registration. The Authority commented that it has received positive feedback from ACIF and industry players about the clarity and comprehensiveness of the guide. ²⁷⁵

Similarly, the Office of Small Business produced *Resolving Small Business Disputes*, which it commends as a resource for examining issues around self-regulatory approaches to dispute resolution.²⁷⁶

The ACCC has also produced guidelines to assist industry with compliance and the authorisation process.²⁷⁷ Similarly, ASIC commented that it has also engaged in a range of activities aimed at promoting alternative dispute resolution processes in the financial services industry. This includes chairing the Complaints Scheme Roundtable which provides a forum for the promotion and support of complaints schemes in the sector.²⁷⁸

The Taskforce considers that government can have a role to play in the development, and promotion, of self-regulation.

²⁷³ Consumer Affairs Division, Commonwealth Department of Treasury 2000, *Directory of Consumer Dispute Resolution Schemes and Complaint Handling Organisations*, 4th edition, released by the Minister for Financial Services & Regulation, the Hon Joe Hockey MP. This document is available on the Internet at http://www.treasury.gov.au/publications/pubs.asp.

²⁷⁴ Benchmarks for Industry-based Customer Dispute Resolution Schemes 1997, released by the then Minister for Customs and Consumer Affairs, Senator the Hon Chris Ellison. This publication can be accessed through the Treasury web-site: http://www.treasury.gov.au/publications.

²⁷⁵ This guide is available from the Australian Communications Authority or through the Internet at: http://www.aca.gov.au/codes/.

²⁷⁶ Available from the Office of Small Business or http://www.dewrsb.gov.au/group_osb/smallbus/resolvedisp/index.htm.

²⁷⁷ For more information visit the ACCC website at: http://www.accc.gov.au.

²⁷⁸ Submission number 37, p. 5.

Government can assist in the analysis of systemic issues

Government can also play a role in the analysis of systemic issues. At the more interventionist end of the self-regulatory spectrum, some schemes may be handling complex and/or a large number of consumer complaints. Where appropriate, the government can assist in the analysis of systemic issues arising from these complaints. Government can also assist in facilitating the design of a self-regulatory response to address those systemic issues.

The Consumer Law Centre of Victoria, Consumer Credit Legal Service (Vic) and the Financial and Consumer Rights Council (Vic) commented that it is inherently difficult for industries to step back from their business to see systemic issues. They suggested that one of the primary strengths of a co-regulatory rather than self-regulatory approach is the ability to build in processes to identify systemic issues within an industry. Within co-regulatory models this presently best occurs through industry based external alternative dispute resolution schemes or direct regulatory oversight.²⁷⁹

The Taskforce recognises that, where appropriate, government can assist in the analysis of systemic issues.

Integration into the regulatory regime and international harmonisation

Government also has a role to assist in ensuring that self-regulation is integrated into the regulatory framework within and outside Australia. Government is in a good position to promote international cooperation and harmonisation of self-regulatory schemes.

For example, the Australian Communications Authority stated that government consultation protects against inconsistency with legislation. For example, the *Telecommunications Act 1997, Trade Practices Act 1974, Privacy Act 1988, Disability Discrimination Act 1992* and State fair trading legislation may all have implications for codes, as may various instruments setting out provisions or standards that must be adhered to. The Authority argued that representation by government agencies ensures that legislative advice is available at the code development stage. ²⁸⁰

ASIC also commented that government can assist in the process of developing schemes — this will include providing information and advice designed to ensure that the schemes are properly integrated into the regulatory framework. ASIC commented that it has extensive experience in dealing with a range of self-regulatory mechanisms that can be brought to bear during the consideration of any self-regulatory scheme. ²⁸¹

²⁷⁹ Submission number 29, p. 13.

²⁸⁰ Submission number 17, p. 5.

²⁸¹ Submission number 37, p. 32.

NRMA stated that an important role for government is to work with State Governments and industry to try to ensure that Federal and State based codes that apply to the same or similar products are as uniform as possible. This will reduce compliance costs, which is a particularly important issue as many companies start to market their products and services nationally. Consumers would also benefit through simpler and more standardised guidelines and codes. ²⁸²

The Association of Superannuation Funds of Australia Limited commented that ad hoc or differing codes and standards or poorly designed regulations or laws can significantly increase the overheads to industry (and to final consumers). It commented that increased overheads can be the outcome where state laws apply.²⁸³

The Taskforce also recognises that Australia is part of a bigger game, and self-regulatory schemes should be aware of overseas standards. For example, Standards Australia commented that consideration of regulation versus self-regulation must be written within a framework of encouraging and facilitating global trade.²⁸⁴

Similarly, the Australian Toy Association commented that any imposts or obligations on companies under any self-regulatory arrangement should not disadvantage Australian companies vis-à-vis international competition.²⁸⁵

NRMA also stated that with the increasingly globalised nature of many markets and the growth of e-commerce, the government also has an important role to play in working to achieve international harmonisation. Consumers would then receive the same level of protection in each market. Conversely, it will help to prevent Australian businesses from losing customers to overseas countries that may offer cheaper products but provide less consumer protection through industry self-regulation. ²⁸⁶

The Taskforce considers that government can assist industry in developing and maintaining the relevance of self-regulatory schemes by raising awareness of the regulatory framework operating within and outside Australia.

Is government just cost-shifting?

Some organisations have argued that government is simply shifting the cost of regulating onto industry. For example, the Australian Food and Grocery Council commented that it does not support self-regulatory measures that are simply seeking

283 Submission number 22, p. 4.

²⁸² Submission number 7, p. 5.

²⁸⁴ Submission number 13, p. 1.

²⁸⁵ Submission number 23, p. 3.

²⁸⁶ Submission number 7, p. 6.

to shift the regulatory resource costs forcing industry to accept the contingent liability of developing and maintaining a regulatory measure.²⁸⁷

The Taskforce considers that self-regulatory schemes should be developed and assessed using the fundamental principles flagged in this report, in other words identification of the specific problem and objective, consultation, assessment of the benefits and costs and the application of the effective minimum solution to the specific problem.

Regulatory creep

During the consultations, some organisations raised the issue of regulatory creep. That is, where self-regulation develops into more quasi-regulation or co-regulation.

This issue was also flagged in the *Grey-letter Law* report, which stated that 'those consulted raised concerns that sometimes what starts out as self-regulation can become widely accepted practice, gain an imprimatur from a government agency, and then become embodied in a quasi-regulatory arrangement, and may become black letter law'.²⁸⁸

Similarly, the Australian Chamber of Commerce and Industry commented that it is concerned about the potential for self-regulatory schemes to become quasi-regulation. Often if a scheme appears to be working well, there has been a tendency for governments to want to formalise it thereby changing the nature of the self-regulatory industry approach. Many incidents occur and are managed under self-regulatory schemes without the need to involve the regulator. The Australian Chamber of Commerce and Industry commented that this is a difficult issue to address especially when there is a prevalent view among sections of the community that only regulation is a sufficient safeguard. ²⁸⁹

ASIC noted that perceptions of regulatory creep may simply reflect a move towards a more coherent and less ad hoc relationship between government and the scheme, rather than representing the growth of an 'overly formalised' approach.²⁹⁰

As a general principle, the Taskforce considers that government should be involved in schemes when there is a public policy objective to do so. If self-regulation is working well, then there is no need for any, or more, government involvement. As noted above, government can assist in promoting the benefits of self-regulatory schemes to consumers.

²⁸⁷ Submission number 30, p. 6.

²⁸⁸ Grey-letter Law: Report of the Commonwealth Interdepartmental Committee on Quasi-regulation, 1997, p. XIV. The report is available at: http://www.pc.gov.au/orr/.

²⁸⁹ Submission number 27, p. 6.

²⁹⁰ ASIC, 14 August 2000.

Degree of government involvement in monitoring self-regulation

There is a range of options to monitor self-regulation, including some form of government involvement. To a large extent, the degree of government involvement in monitoring will depend on the industry concerned, the nature of the industry specific problem and the importance of self-regulation meeting its aims.

There are various models for government supporting and enforcing self-regulation. For example, government can underpin schemes in legislation to improve their effectiveness.²⁹¹

As a broad principle, the Investment and Financial Services Association supported the view expressed in the *Grey-letter Law* report (1997, p. 81) that 'Government should not, however, be directly involved in the monitoring and review of schemes which are self-regulatory. Otherwise, the essential character of self-regulation may be lost. Government involvement may change the character of the self-regulatory scheme to one of quasi-regulation.'²⁹²

Similarly, the Australian Chamber of Commerce and Industry noted that many incidents occur and are managed under self-regulatory schemes without the need to involve the regulator.²⁹³

However, ASIC commented that there is a role for government, in some cases, to monitor compliance with the code, and assist in the process of reviewing the operation of self-regulatory schemes, which may lead to alterations to the scheme or to other regulatory responses (e.g. regulator-issued standards to law reform recommendations).²⁹⁴

ASIC argued that the capacity for timely regulator intervention is especially important in the case of financial services because the industry is heavily dependent on consumer confidence. If the regulator does not possess the power to intervene in a timely fashion in the case of market failure, consumer confidence in the integrity of the Australian financial markets may be compromised.²⁹⁵

Similarly, PowerTel commented that any effective self-regulatory regime needs to recognise that 'blackspots' will emerge and government regulators must be empowered to take remedial action quickly.²⁹⁶

²⁹¹ For more information about underpinning codes in legislation at the Commonwealth level see Department of the Treasury 1999, *Prescribed codes of conduct — policy guidelines on making industry codes of conduct enforceable under the Trade Practices Act 1974.* See also Appendix C of the Taskforce's draft report for further discussion about various models of government enforcement of self-regulation.

²⁹² Submission number 41, p. 3.

²⁹³ Submission number 27, p. 6.

²⁹⁴ Submission number 37, pp. 32-33.

²⁹⁵ Ibid.

²⁹⁶ Submission number 9, p. 5.

The Royal Aeronautical Society also argued that self-regulation must always be accompanied by a rigorous system of dialogue with, and policing by, the government agency responsible for the safety of the public.²⁹⁷

The Australian Direct Marketing Association commented that there is a role for government to channel complaints to appropriate industry bodies and monitor the effectiveness of industry-based complaints handling. This will encourage a continuous dialogue between government and the private sector in the interest of consumer satisfaction. ²⁹⁸

During its regional consultations, the Taskforce heard that rural primary industries like to have some level of government involvement as a safety net. In particular, concern was expressed over rogue traders being able to keep operating under a self-regulatory scheme and potentially undermine it. Some participants suggested that this problem needed to be addressed by government to ensure the safety of consumers and so reputable businesses were not at a competitive disadvantage.²⁹⁹

The Taskforce emphasises that rogue players will not necessarily disappear with more formal government regulation. There can be no guarantee that rogue traders will comply with the law.

As a general principle, the Taskforce considers that the degree of monitoring by government will depend on the degree of market failure and the consequences of self-regulation failing to achieve its objectives. The Taskforce recognises that government has an interest in the review of schemes, and can 'step up' if self-regulation is failing and can help assess whether self-regulation is the most appropriate mechanism.

Authorisation process

Industry should be aware of the authorisation process.³⁰⁰ As noted by the Australian Food and Grocery Council, government also has a role in ensuring that markets are sufficiently competitive and that firms do not enter into arrangements that lessen the degree of competition.³⁰¹

The Department of Industry Science and Resources commented that self-regulatory initiatives attempt to create an informal set of industry standards of behaviour. In most circumstances this has a positive outcome, reducing circumstances of undesirable conduct and providing a better interface between an industry and its market. There is a risk, however, that self-regulatory initiatives can give rise to anti-competitive behaviour, by either suppressing competition between firms, broad

298 Submission number 36, p. 8.

²⁹⁷ Submission number 40, p. 4.

²⁹⁹ Taskforce consultation in Tamworth, 28 June 2000 and Bunbury, 20 July 2000.

³⁰⁰ Chapter 6 discusses the authorisation process.

³⁰¹ Submission number 30, p. 2.

agreements on prices or by forming defacto industry cartels which consolidate market power. The Department commented that during the drafting of the Oilcode, for example, considerable efforts had to be expended to avoid the code entering the area of price setting or market structure, both areas where potentially anti-competitive conduct could arise. 302

Similarly, the Department of Industry Science and Resources argued that self-regulatory initiatives in the form of design standards can be used to establish effective barriers to entry in a marketplace, entrenching the position of existing market participants against the interests of new market entrants. Initiatives on product quality can, while couched in the interests of consumers, contribute to the establishment of a protected market which may not be in consumers' ultimate interest. 303

The Department of Industry, Science and Resources submitted that, while the ACCC monitors anti-competitive arrangements, and acts where it can be shown that such arrangements are not in the public interest, consideration needs to be taken in the preparation of self-regulatory initiatives to minimise anti-competitive effects. 304

During the course of the inquiry, some industry associations raised some concerns over the authorisation process. One industry association commented that authorisation fees are potentially a deterrent for smaller associations.³⁰⁵ At present, there is no discretion under the Trade Practices Act 1974 for the ACCC to waive fees. However, the Taskforce notes that under the national competition policy review process, the government is committed to reviewing fees (including the authorisation fee) charged under the Trade Practices Act 1974 this year.

Another association that has been through the authorisation process, the Australian Direct Marketing Association, commented that the process needs to be more clearly defined. The association submitted that time limits should be placed on public consultation, that it was not the role of the process to redesign or redraft industry documents, and that the applicant should be consulted on the content and context of any public announcement regarding the application.³⁰⁶

The Taskforce has made the ACCC aware of these concerns.

However, overall, organisations consulted found the authorisation process useful and a necessary process for protecting competition. Indeed, the Consumer Law Centre of Victoria submitted that it was a legitimate part of the ACCC's statutory

³⁰² Submission number 31, pp. 19-20.

³⁰³ Ibid. 304 Ibid.

³⁰⁵ Taskforce consultation with the Internet Industry Association, Canberra, 6 March 2000.

³⁰⁶ Submission number 36, p. 9.

role to assist in the redesign of self-regulatory schemes to ensure that such schemes met the 'public benefit' test in the Trade Practices Act.³⁰⁷

The Taskforce considers that the authorisation process is essential in providing a public benefit justification process. However, the Taskforce considers that industry should manage the risk of having to seek authorisation in the first place.

Centralising government responsibility

During the course of consultations, a number of organisations raised the issue of centralising responsibility for self-regulation into one government agency. Some organisations considered that it was confusing and bureaucratic for industry to know which regulator(s) to deal with. For example, the Service Providers Industry Association (SPAN) commented that regulatory oversight of information industries is fragmented across many agencies and organisations. It submitted that industry-specific regulation applicable to telecommunications service providers involves:

- ➤ The Department of Communications, Information Technology and the Arts;
- ➤ The Australian Broadcasting Authority;
- ➤ The Australian Communications Authority;
- ➤ The ACCC:
- The Attorney-General's Department (privacy, interception etc.);
- ➤ The Australian Communications Industry Forum (the principal self-regulatory resource for the industry);
- The Telecommunications Industry Ombudsman; and
- ► The Australian Communications Access Forum. 308

SPAN commented that coupled with a variety of other agencies that administer generic industry and consumer codes and legislation, as well as industry association codes of conduct, the totality represents a complex web of requirements to be understood and followed by industry participants.³⁰⁹

SPAN argued for clear allocation of areas of responsibility to lead agencies, expressed in simple terms that are widely understood and respected by government, industry and consumer interests. Those lead agencies should set the basic regulatory principles that guide regulatory and self-regulatory activities (an example is the

³⁰⁷ Submission of 24 July 2000.

³⁰⁸ Submission number 25, pp. 2-3.

³⁰⁹ Ibid.

establishment of Privacy Principles by the Attorney-General and Privacy Commissioner, consumer protection would be presumably handled by the Treasury).³¹⁰

Similarly, NRMA commented that another action by government that could help to promote more effective self-regulation would be to establish a specialised Unit within a department such as Treasury to deal with self-regulatory issues. This would help to ensure a consistent approach to self-regulation by government as well as streamline the administrative process for business and consumers. ³¹¹

However, government agencies argued that each agency has its own specialised area and to try to centralise this expertise would be futile. For example, as noted above, ASIC has responsibility for the financial sector whereas the Department of Health and Aged Care oversees the health industry. The agencies argued that a centralised agency dealing with self-regulation would lead to one big, bureaucratic organisation which would be detrimental to industry.

ASIC submitted that centralising responsibility into one agency to cover self-regulatory arrangements across all industries would not be successful. It argued that it is more appropriate for government agencies and regulators with responsibilities for particular industry sectors to deal with self-regulation in that sector. ASIC considered that departmental and agency roles should be made clear and noted that the Consumer Affairs Division of Treasury provides general advice on self-regulatory issues.³¹³

To some extent, the Consumer Affairs Division in Treasury does offer industry advice on developing codes of practice in the first instance. This advice revolves around the Codes Kit containing previously mentioned publications such as *the Codes of Conduct Policy Framework* and *Benchmarks for Industry-based Customer Dispute Resolution Schemes.*³¹⁴ In addition, the ACCC also assists self-regulatory schemes, particularly with compliance and enforcement.

The Taskforce considers that centralising government responsibility for self-regulation would result in a loss of expertise. However, the Taskforce considers that government needs to ensure Departmental and agency roles in self-regulation are clear.

³¹⁰ Ibid.

³¹¹ Submission number 7, p. 6.

³¹² Taskforce consultation with Government agencies held in Canberra, 6 December 1999.

³¹³ ASIC, 14 August 2000.

³¹⁴ Codes of Conduct Policy Framework released by the then Minister for Customs and Consumer Affairs in March 1998; this document is available on the Internet at http://www.treasury.gov.au (choose Consumer Affairs/Publications/Industry Self-Regulation Publications).

Role of consumers

The Taskforce recognises that consumer groups play an important role in developing and maintaining the relevance of self-regulation. A number of organisations (including industry, consumer groups and government agencies) have expressed the importance of consumer participation in adding credibility to self-regulatory schemes.

Consumer input

The Australian Communications Authority commented that consultation with consumer groups and the public is one of the most important requirements of developing a telecommunications code for registration. Consultation with these groups is assured by Part 6 of the *Telecommunications Act 1992* which requires the Australian Communications Industry Forum (ACIF) to consult with at least one consumer representative organisation, and obliges ACIF to also provide a minimum 30 day consultation period for comments from the public. In practice, ACIF usually provides a 45 day consultation on consumer codes.³¹⁵

The Australian Communications Authority suggested that consumer organisations in particular have made a very significant contribution to the development of ACIF codes. Consumer representatives participate on all six Consumer Codes Reference Panel Working Committees and advise on all consumer issues which arise in code development.³¹⁶

It also commented that consumer groups and representatives have been instrumental in determining the priorities and work programme of the ACIF Consumer Codes Reference Panel. The Authority commented that they have made significant research and policy contributions to the work of the Reference Panel, both in terms of sharing results of research undertaken under the auspices of other organisations and institutions, and in undertaking contracted research and policy work on ACIF's behalf. During the public comment phase consumer representatives provide advice on targeting interested groups and utilising networks to distribute information about the draft code.³¹⁷

Similarly, the Department of Health and Aged Care commented that consumers are becoming sceptical when they perceive self-regulation imposed as a cheap option, and for the benefit of business. The Department argued that, overall, if self-regulation is to work effectively, there needs to be consideration of how to encourage strong community involvement to ensure it remains open to rigorous, public scrutiny. In this context it is worth noting the level of public debate involving the passage of sensitive primary legislation through Parliament. Overall, there is a need for a clearer picture, or at least guidelines, on the role of self-regulation in

³¹⁵ Submission number 17, p. 7.

³¹⁶ Ibid.

³¹⁷ Ibid.

supporting genuine tripartite solutions. If such developments do not occur, we risk seeing declining consumer faith in self-regulation, and other useful alternative forms of regulation (such as public health promotional activities). 318

Similarly, industry bodies have acknowledged the importance of consumer input. For example, the Insurance Council of Australia recognised the necessary role to be played by consumer groups in the development and maintenance of effective industry self-regulation. Likewise, the Financial Industry Complaints Service commented that consumer groups have been part of the working party for the development of their scheme. It suggested that it is important that the general public has confidence in the schemes and consumer groups assist in providing this. 320

Insurance Enquiries & Complaints Limited also commented that both industries and consumer groups can learn and benefit from each other as schemes can improve the level of information on problems that consumers are facing.³²¹

One observer, from the Safe Food Consumers (& Growers) Association, stressed that there is a strong need for consumer input. He submitted that underpinning industry self-regulation there must be an independent body responsive to the rights of consumers to regulate for safety, quality and fair pricing of all products and services.³²²

The Consumer Law Centre of Victoria, Consumer Credit Legal Service (Vic) and the Financial and Consumer Rights Council (Vic) commented that in order to deliver outcomes which serve the interests of consumers as well as industry it is necessary to include expert consumer advocacy and participation in the development and oversight of co-regulatory schemes or models.³²³

The Consumer Law Centre of Victoria, Consumer Credit Legal Service (Vic) and the Financial and Consumer Rights Council (Vic) endorse consumer participation in the following processes:

- Government and regulatory consultative processes;
- public hearing and enquiries;
- the development, amendment and review of industry standards and conditions of service and supply;
- the development of consumer friendly products and services;

³¹⁸ Submission number 44, p. 13.

³¹⁹ Submission number 18, p. 7.

³²⁰ Submission number 15, p. 4.

³²¹ Insurance Enquiries & Complaints Limited, Melbourne, 20 June 2000.

³²² Mr Frank Ash, Safe Food Consumers (& Growers) Association, 12 July 2000.

³²³ Submission number 29, p. 14.

- the development of codes, charters and dispute resolution mechanisms; and
- ▶ the oversight and management of external dispute resolution schemes. 324

At the international level, other countries such as the United Kingdom and Canada also acknowledge the valuable contribution that consumers can make to successful self-regulatory schemes. Canada has focussed on consumer contribution to policy development particularly in the area of voluntary codes, in conjunction with non-governmental organisations and the private sector, to ensure consumer protection.³²⁵

Human and financial resource constraints

The Taskforce also recognises that consumer participation is constrained by limited resources. For example, the Consumer Law Centre of Victoria, Consumer Credit Legal Service (Vic) and the Financial and Consumer Rights Council (Vic) commented that consumer groups and consumers generally are largely under-resourced in terms of time and funding, both of which are necessary to meaningfully participate. This is particularly the case when compared to the resources generally available to industry participants. The Consumer Law Centre of Victoria, Consumer Credit Legal Service (Vic) and the Financial and Consumer Rights Council (Vic) argued that as part of the examination of specific industries proposing to move to a less prescriptive regulatory model, industries should be required to propose models for the inclusion of consumer representatives. That representation should be proportional with industry participation. ³²⁶

Similarly, the Consumers' Telecommunications Network commented that there are resource implications in meeting the demand for consumer input arising from many different sources. Organisations such as themselves which rely substantially on volunteer contributions find it increasingly difficult to attract and retain sufficient consumer representatives to meet the needs of the various regulatory and self-regulatory bodies.³²⁷

Further, NSW Legal Aid argued that, as noted in the *Grey-letter Law* report, there needs to be effective resourcing provided to consumer groups to participate in the drafting and subsequent reviews of codes, and in the bodies responsible for overseeing the code.³²⁸

NSW Legal Aid commented that if self-regulatory or co-regulatory schemes are to work properly and deliver the benefits envisaged, then there needs to be effective

³²⁴ Ibid.

³²⁵ For further information please see Office of Consumer Affairs, Industry Canada Fall 1999 — Winter 2000, *Consumer Quarterly*, Volume 5, Number 1.

³²⁶ Ibid.

³²⁷ Submission number 32, p. 2.

³²⁸ Submission number 24, p. 6.

participation of consumer representatives and, therefore, a commitment by government to properly resourcing those representatives.³²⁹

The Australian Consumers' Association recommended that a levy be imposed on the supply side of the market of other industries to ensure appropriate demand side input.³³⁰

The Small Enterprise Telecommunications Centre Limited (SETEL) noted that government grant funding has enabled consumer groups to adopt a broader focus on telecommunications issues and to participate in self-regulatory forums.³³¹

SETEL commented that the 'voice' of consumers has been strengthened through this process and the involvement in the self-regulatory process has enabled the presentation of the various consumer perspectives to a wider audience in the telecommunications industry. The major discrepancy existing today is the imbalance of input into the self-regulatory process caused by restrictions in funding of consumer representation so as to be able to 'compete' with the input (and influence) of the industry representatives.³³²

SETEL also contended that the self-regulatory model used in the telecommunications industry in particular, pits some of Australia's most powerful corporations against consumer representatives in a frequently unequal battle. It believed that the funding of consumer representation in any self-regulating industry should be commensurate with the size and resources of the industry and the extent to which consumer representation is required in self-regulatory forums.³³³

In other words, SETEL argued that 'David must at least have a sling and a stone when going in to battle with Goliath'. 334

ASIC also argued that consumer organisations would require access to sufficient resources if they were to contribute effectively to the process of developing self-regulatory schemes.³³⁵

The Taskforce recognises that consumer groups, and consumer participation more generally, assume an important role in the development and in maintaining the relevance of self-regulation. The Taskforce also recognises that consumer participation will be limited by human and financial resource constraints if there is no external assistance forthcoming.

333 Ibid.

³²⁹ Submission number 24, p. 6.

³³⁰ Submission number 16, p. 2.

³³¹ Submission number 20, pp. 2-4.

³³² Ibid.

³³⁴ Ibid.

³³⁵ Submission number 37, p. 13.

Oversight committee

During the course of consultations, the Taskforce has heard a number of organisations advocating for an oversight committee for self-regulation. For example, the Commonwealth Consumer Affairs Advisory Council commented that in a maintenance and monitoring role, the government could create a small 'Code' authority independently reporting annually on the performance of endorsed codes (award schemes etc), innovations and shortcomings. It would also be responsible for endorsement and public education. The Advisory Council discussed that this may be able to be achieved at effectively no cost by redirecting existing ASIC and/or ACCC funds being applied to self-regulation issues. ³³⁶

Similarly, the Australian Consumers' Association cited that while a lot of effort goes into the formation of self-regulatory standards, this diligence is rarely followed through in practice. For example, it submitted that a series of recommendations from a review have not been implemented. The Association would support the recommendation to create a tri-partisan (government, consumers and industry) oversight committee that would seek to ensure self-regulation is not only implemented but also maintained. 337

Further, Clayton Utz commented that a body could be responsible for administering the codes and liaising with industry bodies to ensure that the various codes are reviewed and amended where appropriate at regular intervals. This body could also be responsible for maintaining a code register website and interacting with bodies that are proposed to perform similar functions under legislation such as the Privacy Commissioner. It commented that to ensure the regulatory burden imposed by such administration is not too heavy on code subscribers it is recommended that such an administrator's role be 'light handed' and facilitative rather than compelling. 338

Similarly, one participant in the Brisbane consultations with a background in regulatory policy, Mr Euan Morton, commented that there is scope for a private organisation or government body to compare and rate various codes to improve the level of information on them. He acknowledged that an oversight body would add an extra layer of bureaucracy, but he considered that the benefits in assisting consumers would outweigh the costs.³³⁹

However, the Taskforce recognises from an industry perspective, some form of oversight committee would add an extra level of bureaucracy that industry has to deal with. Hence, industries may shy away from developing self-regulatory schemes or not commit as many resources than it would have otherwise. Further, to a large extent, the government already plays a role in monitoring self-regulation and ensuring that it is still relevant.

337 Submission number 16, p. 2.

³³⁶ Submission number 14, p. 6.

³³⁸ Submission number 43, p. 3.

³³⁹ Taskforce consultation in Brisbane, 26 June 2000.

ASIC commented that while there is a need for effective ongoing monitoring of self-regulatory schemes, it does not believe that an oversight committee is the best way to deal with issues and would unnecessarily add to existing agency structures.³⁴⁰

Developing a model code of practice

As discussed elsewhere, there is no single industry model of self-regulation. However, there are elements of good practice in self-regulation that are generally consistent across schemes.³⁴¹ Some organisations have argued that a model code could be developed along these elements of good practice.

Clayton Utz commented that there should be uniform requirements for codes. It discussed the establishment of a register for codes of conduct, where a quality assessment procedure could be implemented to ensure that a particular code was appropriate for inclusion in the register. To this end, a set of core requirements should be developed and finalised as being necessarily evident in each code appearing in the register as well as ensuring a degree of uniformity between different codes. ³⁴²

Clayton Utz suggested that such core requirements should include:

- complaint handling and dispute resolution procedures which comply with AS4269 (the Australian Standard on complaints handling);
- > sanctions and penalties appropriate to the industry for deliberate and continuous breaches of the code;
- the information privacy principles (where appropriate);
- provisions dealing with transactions conducted via electronic commerce (where applicable); and
- procedures for the regular review and amendment of the code.

Clayton Utz argued that by making code provisions more uniform, industry participants which are regulated by multiple codes of conduct can achieve compliance more effectively without needing to develop multiple code compliance systems.³⁴³

Similarly, the Australian Direct Marketing Association commented that the role of government in respect of codes should be to both promote national consistency by developing model codes under the auspices of (for example) the Ministerial Council

109

³⁴⁰ ASIC, 14 August 2000.

³⁴¹ See chapter 6.

³⁴² Submission number 43, p. 2.

³⁴³ Ibid.

on Consumer Affairs, where appropriate, and to encourage self-regulatory industry bodies to develop codes based on the model.³⁴⁴

Further, the Commonwealth Consumer Affairs Advisory Council suggested that there is a role for government in establishing a government endorsed 'Code' framework (developed through Australian Standards). It argued that only participants in codes who meet the requirements can advertise that fact, and participants in codes must advertise that their code does not meet the requirements. Participants in an industry with a complying code who do not subscribe should advertise the fact. The Advisory Council suggested that part of the endorsement process would involve the examination of competition issues, for example cartels of big players constructing codes which may have the effect of forcing out smaller players.³⁴⁵

As discussed throughout the report, the Taskforce considers there is no single industry model for self-regulation or codes. However, there are some general principles of good practice as discussed in chapter 6.

-

³⁴⁴ Submission number 36, p. 8.

³⁴⁵ Submission number 14, p. 6.

³⁴⁶ The Taskforce also recognises that the Ministerial Council of Consumer Affairs guideline 1996, Fair Trading Codes of Conduct, Why have them, how to prepare them is a useful reference tool for industry developing codes.

Chapter 8

Options that facilitate the improvement and harmonisation of dispute resolution schemes

The Taskforce is to inquire into and report on options that facilitate the improvement and harmonisation of dispute resolution schemes while reducing costs to industry and improving outcomes for consumers.

Effective dispute resolution is a crucial element of industry self-regulation offering redress to consumers and it can also identify systemic problems in the industry. Dispute resolution schemes are an excellent monitoring tool increasing performance and industry standards.

However, dispute resolution schemes come at a cost. In particular, they can be costly for small industry groups. One option is for dispute resolution to operate across different sectors with similar products/services which may reduce costs to business and reduce confusion for consumers. Also, there may be scope to consolidate dispute resolution schemes.

The following chapter explores some options that facilitate the improvement and harmonisation of dispute resolution schemes.

Principles

- The type of dispute resolution scheme, if required, should depend on the nature of the complaints and type of self-regulatory model.
- A scheme is only as effective as its broader coverage of industry participants, so it should aim for comprehensive membership.

Conclusions

- 42. In the future dispute resolution schemes may operate across different sectors with similar products/services, driven by changes in technology and market circumstances. Harmonisation of schemes would be less costly and less confusing to consumers and the use of umbrella-type arrangements with a single co-ordinated access point would likewise be of assistance to consumers.
- 43. Promotion of dispute resolution schemes to consumers raises their awareness of the availability of quick and inexpensive redress.

Range of dispute resolution schemes

In recent years, the growth in self-regulation has seen an expansion in the range of redress mechanisms available to consumers well beyond the traditional sphere of Small and Consumer Claims Courts and their equivalents. A number of industry-based dispute resolution schemes provide consumers with a cheap and accessible means of resolving disputes.

The form of dispute resolution schemes also varies depending on the nature and quantity of complaints. Two of the biggest dispute resolution schemes are the Australian Banking Industry Ombudsman and the General Insurance Enquiries and Complaints Scheme. These schemes have a large number of staff dealing with up to 50 000 inquiries per year. In contrast, the Telephone Information Services Standards Council, which handles consumer complaints about primarily '0055/1900' telephone numbers, has a small number of staff and generally deals with complaints of a low monetary value. Also consumer complaints of a low monetary value.

As discussed in chapter 6, dispute resolution schemes can be expensive. A 'Rolls Royce' dispute resolution scheme is not necessary if the complaints are minor and are of small monetary value or significance. Again, the type of dispute resolution

³⁴⁷ Submission number 37, p. 30.

³⁴⁸ Taskforce consultation with the Telephone Information Services Standards Council, Sydney, 1 March 2000.

scheme should be that which imposes the least cost of compliance consistent with achieving the identified objectives.

As a general principle, the Taskforce considers that the type of dispute resolution scheme that best meets consumer needs will depend on the nature of the complaints. However, an overarching consideration that will apply to all schemes is that a scheme is only as effective as its coverage of market participants, or the extent to which consumers can factor dispute resolution participation into their purchasing decisions.

Convergence

A challenge for self-regulatory schemes is that Australian markets are dynamic markets with increasing convergence and globalisation. Clayton Utz suggested that future codes will need to be more inter-industry rather than intra-industry and adaptable to take into account changing consumer relationships resulting from advances in technologies and adapting delivery systems.³⁴⁹

Hence, market forces may lead self-regulatory schemes being developed along more functional lines. For example, convergence of industry sectors and the rapid redrawing of industry boundaries are occurring in telecommunications. The Service Providers Industry Association commented that the telecommunications industry brings a tradition of detailed regulation and standardisation and an engineering culture that insists on industrial strength, reliability and simplicity of user interfaces. The information technology industry sector has relied on a relative absence of regulation to support decades of rapid growth and technological change. Convergence poses continuing challenges for regulators and the self-regulatory process. The Association argued that there needs to be recognition that these basic regulatory principles will need to be directed primarily at cross-industry activities rather than industry-specific transactions and processes. 350

Similarly, convergence continues to occur in the finance sector, generating pressure for the merger of codes and dispute resolution schemes. However, the financial services industry is different to other industries in that there is a proliferation of dispute resolution schemes and there are other factors driving rationalisation in the industry. ASIC commented that membership of an approved scheme has become mandatory for an increasing number of industry participants, and existing complaints resolution schemes are subject to consideration under ASIC's Policy Statement 139. Further, ASIC commented that the Corporate Law Economic Reform Program (CLERP) requirements for dispute resolution scheme membership are also likely to be a key driver of rationalisation. Convergence and the establishment of corporate composite service providers will drive fewer schemes.

350 Submission number 25, p. 2.

³⁴⁹ Submission number 43, p. 1.

³⁵¹ Submission number 37, p. 26.

Potential for dispute resolution schemes to operate across different sectors with similar products/services

The Taskforce considers that it is possible for dispute resolution schemes to cover different sectors of industries with similar products/services on a shared cost basis as a means of capitalising on economies of scale. For example, the Australian Direct Marketing Association (ADMA) has a dispute mechanism that operates across a range of different sectors and industries. ADMA commented that their code binds all ADMA members and all employees, agents or subcontractors of ADMA members. It requires members to ensure their suppliers comply, by requiring that this is a condition of contracts between members and their suppliers. The Code seeks to curb behaviour by members that may be inconsistent with widely accepted best practices in direct marketing. ADMA commented that this approach has been applied consistently across all direct marketing media and direct marketing activity on both the user and supplier sides of the ADMA membership. 352

In theory, there are a number of advantages for dispute resolution schemes to operate across different industries and sectors with similar products/services. The main advantage is that more rationalised schemes will benefit from lower costs overall by virtue of the economies of scale. These lower costs can be passed onto industry members and consumers. Consumers would also benefit through a less confusing and better promoted dispute resolution system. From a consumer's perspective, a 'one stop shop' with a single co-ordinated access point decreases the need for consumers to be referred from one scheme to another.

The Financial Industry Complaints Service suggested that schemes should be willing to cover different sectors of industry if it is of no cost to them. Cross industry schemes would be more effective so long as the industries have some common thread. ³⁵³

However, as discussed elsewhere, industry commitment is very important in establishing and maintaining self-regulatory schemes.³⁵⁴ Sharing schemes may lead to a loss of 'industry ownership' and therefore commitment in some schemes. During Taskforce consultations some industry bodies also commented that larger schemes would need to avoid becoming bureaucratic.³⁵⁵ Clearly, harmonisation of schemes would require some greater formality in administration.

The Investment and Financial Services Association commented that industry schemes consistently benefit from expertise and industry knowledge on the part of those who establish them. The need for schemes to recruit directors, adjudicators,

_

³⁵² Submission number 36, p. 4.

³⁵³ Submission number 15, p. 4.

³⁵⁴ Chapter 5 indicated that industry commitment leads to more effective self-regulation.

³⁵⁵ For example, Taskforce consultation with the Insurance Brokers' Dispute Facility, Melbourne, 22 November 1999. It commented that it would be hard to point the consumer in the right direction if there was just one broad scheme covering financial services.

panel members and other personnel who have specialised knowledge will, in the case of very large schemes, translate to a need for bureaucratic processes and a consequent remoteness from both industry members and consumers. ³⁵⁶

Further, although the Law Council of Australia recognised that there is no reason why dispute resolution schemes cannot operate across different sectors of industries, the Council commented that consumer rights must be maintained. In particular, the Law Council of Australia argued that consumers (unlike industry) are not concerned about the cost of these schemes. Any suggestions of making cross industry schemes more cost effective conjures up staff cuts and procedure cuts. To the consumer, the costs of goods and services that are perhaps slightly lower in the short run is inadequate compensation for the erosion of the rights they currently have and the rights they should have. 357

The Taskforce recognises that in the future cross-sector dispute resolution schemes may be used more, driven by changes in technology and market circumstances. Harmonisation of schemes would be less costly and less confusing to consumers and the use of umbrella-type arrangements with a single co-ordinated access point would likewise be of assistance to consumers. However, the issue is whether specialised functional services can be provided by fewer schemes.

Shared case management/dispute resolution mechanisms

Apart from the possibility of dispute resolution schemes covering different sectors of industries with similar products/services, it may also be possible for industries to share dispute resolution schemes. In particular, there are opportunities for schemes to share backend services to the benefit of the respective industries. For example, Insurance, Enquiries and Complaints Limited commented that it currently performs accounting, payroll, and Company Secretary functions for the Financial Industry Complaints Service scheme to the financial benefit of both organisations and their associated industries.³⁵⁸

Similarly, the Australian Banking Industry Ombudsman considered that sharing services between schemes is a good concept.³⁵⁹

ASIC commented that rationalised schemes will benefit from lower costs overall by virtue of the economies of scale arising from a single budget (e.g. for administration, property and marketing) that would otherwise have to be generated for several schemes. These lower costs can be passed onto industry members. ³⁶⁰

115

³⁵⁶ Submission number 41, p. 4.

³⁵⁷ Submission number 19, p. 10.

³⁵⁸ Submission number 8, p. 4.

³⁵⁹ Taskforce consultation in Melbourne, 20 June 2000.

³⁶⁰ Submission number 37, p. 25.

ASIC appreciated that an appropriate degree of flexibility in the structure of self-regulation should not be pursued at the expense of appropriate minimum standards, structural efficiencies and better market outcomes for industry members and consumers.³⁶¹

However, Insurance, Enquiries and Complaints Limited stated that careful consideration must be given to the beneficial effect that a high degree of industry 'ownership' has had for its scheme and other schemes. Also, the Law Council of Australia stated that shared case management and dispute resolution mechanisms can be viable only if consumer safeguards are maintained.

In addition, one smaller industry association observed that there were no economies of scale to join with other schemes.³⁶⁴

On balance, the Taskforce recognises that sharing services is one means for smaller industry associations and smaller businesses to be involved in dispute resolution schemes.

Overall, the Taskforce considers it is important that the structure of self-regulation is flexible and does not become the site of institutional 'empire building' that will prove costly for both industry members and consumers.

Promoting dispute resolution schemes

The promotion of dispute resolution schemes is both beneficial to industry and to the consumer. For industry, a dispute resolution scheme can be used as a marketing tool to differentiate themselves from competitors. Whereas, for the consumer, the promotion of dispute resolution schemes informs them of their rights.

The Taskforce heard that an United Kingdom industry association was loath to promote its dispute resolution scheme in fear that people might find out about it. While this story is extreme, it illustrates that unless a scheme is promoted then a consumer may not even know to avail themselves of a quick and inexpensive redress mechanism. Access is very important for effective self-regulation. The Taskforce considers that the promotion of schemes is the responsibility of both industry and government, and to a lesser extent consumer groups and counselling services.

The Law Council of Australia commented that industry can improve access by providing effective, simple communication which allows the consumer to fairly present grievances and complaints. Industry should provide information packages to

_

³⁶¹ ASIC, 14 August 2000.

³⁶² Submission number 8, p. 4.

³⁶³ Submission number 19, p. 10.

³⁶⁴ Taskforce consultation with the National Furnishing Industry Association of Australia Ltd, Melbourne, 23 November 1999.

³⁶⁵ Taskforce consultations in Melbourne, 22 November 1999.

consumers when goods or services are provided. ³⁶⁶ For example, Insurance Enquiries and Complaints Limited stated that consumer access to the scheme is ensured by the Code requirement that a member company inform the consumer about the external dispute resolution scheme once an internal dispute resolution decision has been reached. ³⁶⁷

Similarly, the Telecommunications Industry Ombudsman (TIO) commented that consumers are aware of the Scheme's jurisdiction and of the limits on that jurisdiction. Apart from general media and other activities, the Scheme provides a booklet on the TIO's functions to every complainant who contacts the TIO. ³⁶⁸

Further, for promotion to be effective for consumers it also needs to focus on the organisations that consumers go to when there is a problem , for example Fair-trading agencies or financial counsellors. There will always be insufficient resources to educate consumers of a service which is only likely to be used in a minuscule number of cases.

The Government can also contribute to the harmonisation of dispute resolution schemes. A couple of Commonwealth initiatives are discussed below.

Directory of Consumer Dispute Resolution Schemes and Complaint Handling Organisations

One initiative that has been useful is the Commonwealth's *Directory of Consumer Dispute Resolution Schemes and Complaint Handling Organisations* which sets out consumer dispute resolution schemes, complaint handling organisations and other useful contacts such as fair-trading agencies.³⁶⁹ The Directory is a practical reference guide for individual consumers and organisations that advise consumers and small business.

A number of industry associations, consumer groups and government agencies submitted that the Directory has been useful. For example, the Telecommunications Industry Ombudsman commented that there is more than anecdotal evidence that the Directory is an important initiative. The Directory is commonly used by consumer groups, legal aid workers and financial counsellors to ensure that consumers are directed to the appropriate redress mechanism. ³⁷⁰ Similarly, the

³⁶⁶ Submission number 19, p. 10.

³⁶⁷ Submission number 8, p. 4.

³⁶⁸ Submission number 21, p. 11.

³⁶⁹ Consumer Affairs Division, Commonwealth Department of Treasury 2000, *Directory of Consumer Dispute Resolution Schemes and Complaint Handling Organisations*, 4th edition, released by the Minister for Financial Services and Regulation, the Hon Joe Hockey MP. This document is available on the Internet at http://www.treasury.gov.au/publications/pubs.asp.

³⁷⁰ Submission number 21, p. 11.

Financial Industry Complaints Service commented that the Directory is 'most helpful' and the scheme distributes it when presenting seminars.³⁷¹

Financial Complaints Referral Centre

As discussed elsewhere, access to schemes is very important for self-regulation to be effective and there is evidence to suggest that consumers get confused and frustrated if they have to go through multiple schemes. In February 1998, a Commonwealth initiative to overcome the problem of consumer confusion associated with a proliferation of schemes in the financial services industry was the introduction of the Financial Complaints Referral Centre, operated by ASIC.

It operated as a portal to the dispute resolution schemes in the financial services industry. The Centre's staff advised callers to refer their complaints to the relevant service provider in the first instance and provided contact details for the appropriate dispute resolution scheme(s) if a deadlock remained.

The operations of the Centre were reviewed by ASIC in late 1998 and February 1999. ASIC commented that the reviews found that around 5 to 6 calls per day were being referred to the industry-funded complaint schemes. For the purpose of comparison, the two largest industry ADR schemes — the Australian Banking Industry Ombudsman and the Insurance Enquiries and Complaints — each receive between 45,000 to 50,000 telephone contacts per year.

The views expressed by industry and consumer groups during the Taskforce consultations also indicated that the Financial Complaints Referral Centre has not been effective in channelling calls.

ASIC commented that during the Referral Centre's development, industry stakeholders expressed reservations not only about the actual demand for a central gateway, but also that the promotion of the Centre might detract from the efforts of the dispute resolution schemes to effectively promote themselves. Thus, detracting from industry's sense of ownership of the established dispute resolution processes. ³⁷³

ASIC submitted that there was also resistance from existing dispute resolution schemes to devoting significant resources to the establishment and operation of a central gateway without detailed evidence that there was a sufficient level of demand for the service.³⁷⁴

The Taskforce notes that the Referral Centre has now been closed. The referral of complainants to those dispute resolution schemes dealing with general and life

³⁷¹ Submission number 15, p. 5.

³⁷² Submission number 37, p. 30.

³⁷³ Ibid.

³⁷⁴ Ibid.

insurance, superannuation and banking is a role that ASIC's Infoline would have played in any event, and that it will continue to play in the future. 375

The Taskforce understands the schemes themselves already have established referral mechanisms to assist consumers to locate the most appropriate dispute resolution mechanism.

-

³⁷⁵ ASIC Infoline number is 1300 300 630.